AN ETHICO-LEGAL ANALYSIS OF BENEFIT SHARING FOR HEALTH RESEARCH IN SOUTH AFRICA

by

PRISCILLA CHUNDA-SINOMBE

STUDENT NUMBER: 61524727

submitted in accordance with the requirements

for the degree of

DOCTOR OF PHILOSOPHY IN LAW

at the

UNIVERSITY OF SOUTH AFRICA

SUPERVISOR: PROF M LABUSCHAIGNE

CO-SUPERVISOR: PROF S MAHOMED

JUNE 2023

DECLARATION

Name:	PRISCILLA CHUNDA-SINOMBE
Student number:	61524727
Degree:	Doctor of Philosophy in Law
Title of thesis:	AN ETHICO-LEGAL ANALYSIS OF BENEFIT SHARING FOR HEALTH RESEARCH IN SOUTH AFRICA
	INOMBE, declare that the above thesis is my own work and have used or quoted have been indicated and acknowledged by nces.
I further declare that I sub- within the accepted require	mitted the thesis to originality checking software and that it falls ements for originality.
	e not previously submitted this work, or part of it, for mother qualification or at any other higher education institution.
(R) Ja	
SIGNATURE	DATE
SIGNATURE	DATE

ACKNOWLEDGMENTS

I dedicate this work to my children, Donattie, Leandra and Maya Sinombe. May the world be kind to you. It is a world full of possibilities.

I am deeply indebted to my supervisors, Professor Melodie Labuschaigne and Professor Safia Mahomed for their constructive feedback and the knowledge and expertise they shared with me. Thank you both for starting this journey with me; because of you, it has been a pleasant and knowledgeable journey.

I am very grateful to Ms Lorinder Gerber for the editing of this thesis.

It would be amiss to not thank Professor Rose Hayeshi, my dearest friend, who encouraged me to start this journey and therefore had to listen to all my dilemmas.

Lastly, I would like to thank my family; especially my parents, my spouse and my children, for being my inspiration and motivation during this journey. Your constant belief in me and your cheerleading kept me going. Thank you.

ABSTRACT

Health care research conducted in LMICs, including South Africa, has exploded in the last few decades. Albeit a welcome development, there has emerged the fear of and actual exploitation of host communities. As health care research has become funded by private business and often in North-South collaborations at academic institutions by foreign government agencies, potential exploitation is a reality, as there is usually an unfair distribution of risks and benefits among the parties involved. While it stands true that the overall goal of health research is to attain global health and wellness for all, health research using HBMs cannot occur in an exploitative environment that takes unfair advantage of people's vulnerabilities. Benefit sharing should be a tool for guarding against exploitation and not the basis of a strategy to address urgent global health needs or resolve inherent issues of global distributive justice.

In an attempt to identify the best benefit sharing model for health research for South Africa from an ethico-legal view perspective, one that tempers (not diminishes) commercial interests, redresses economic imbalance and gives research participants fairer and more active roles in influencing the sharing of benefits, this thesis canvasses the current legal framework for benefit sharing in South Africa, as well as other jurisdictions.

The thesis concludes that a benefit sharing framework, based on the charitable trust model, could be adopted in South Africa. This framework recognises the various stakeholders that are part of a research project at different levels of society, whilst simultaneously acknowledging that it is possible to have different types of fair benefits at each stakeholder level, even in the absence of a final, tangible benefit. In terms of this model, academic medical centres and/or research institutions would cease to be brokers of the HBMs and instead become custodians of the samples. This proposed framework will promote compliance with data privacy and informed consent requirements without compromising its value as an information-rich HBM supplier. It would also make the first recipients of HBMs the trustees of the HBMs instead of brokers having legal fiduciary duties over the HBMs, whilst permitting the use of the donated HBMs in a way that benefits the donor as a beneficiary of the trust. Moreover, this trust model will be in the ideal position to create and facilitate continuous communication channels with the donor community, researchers, policymakers and teaching hospitals for fostering trust in health research and its benefits for all stakeholders involved.

CHAF	PTER 1	
	KGROUND TO THE STUDY, PROBLEM STATEMENT, OBJECTIVES A MEWORK	ND
1.1 B	ackground, literature review and problem statement	9
1.2	Rationale of study	22
1.3	Research question	22
1.4	Aims and objectives	23
1.5	Expected outcomes of the study	23
1.6	Methodology	24
1.7	Limitations of study	24
1.8	Chapter outline	24
CHAF	PTER 2	
BENE	FIT SHARING IN HEALTH RESEARCH	
2.1	Introduction	26
2.1.1	Defining benefit	28
2.2	A brief overview of the recognition of benefit sharing globally and in	
	South Africa	32
2.2.1	Background	32
2.2.2	Benefit sharing relating to non-human biological resources	33
2.2.3	Benefit sharing relating to human biological resources	36
2.2.3.	1 UNESCO Universal Declaration on Bioethics and Human Rights	38
2.2.3.	2 Human Genome Organisation Project: Ethics Committee Statement	40
2.3.3.	3 Declaration of Helsinki	42
2.2.3.	4 Council for International Organisations and Medical Sciences (CIOMS)	44
2.2.4	Benefit sharing in South Africa: non-human biological resources	46
2.2.4.	1 Biodiversity Act 10 of 2004	46
2.2.5	Benefit sharing in South Africa: human biological resources	49
2.2.5.	1 Department of Health: Ethics in Health Research-Principles, Processes	and
Struc	tures, 2015	50
2.2.5.	2 Health Professions Council of South Africa (HPCSA) Guidelines	51
2.2.5.	3 National Material Transfer Agreement for Human Biological Material	52
2.3 TI	he ethics of benefit sharing	54
2.3 The ethics of benefit sharing 2.3.1 Vulnerability		55
2.3.2	Exploitation	58
2.3.3	Undue inducement	62
2.4 A	Itruism and justice in health research	65
2.4.1	Altruism in health research	65
2.4.2	Justice in health research	66

2.4.2.	1 "Fair benefit: the case of the Majengo sex workers"	68
2.5 Be	enefit sharing models	70
2.5.1 The private benefit sharing model		70
2.5.2	The altruistic benefit sharing model	71
2.5.3	2.5.3 The charitable trust benefit sharing model	
2.6 Cc	onclusion	74
CHAP	PTER 3	
ETHIC	CO-LEGAL FRAMEWORK RELEVANT TO BENEFIT SHARING	
3.1	Introduction	76
3.2	Ownership of HBMs in the era of biotechnology	77
3.2.1	Ethico-legal issues relating to ownership of HBMs and human dignity	78
3.2.2	Proposed ethical models when considering the regulation of HBMs	81
3.3	South African legal framework regarding ownership of HBMs; intellectua	
prope	rty rights and benefit sharing	83
3.3.1	Constitution of the Republic of South Africa, 1996	83
3.3.2	National Health Act 61 of 2003 and relevant Regulations	86
3.3.2.	1 Regulations Relating to the Use of HBMs	87
3.3.2.2	2 Regulations Relating to the Artificial Fertilisation of Persons	89
3.3.2.3	3 Regulations Relating to Tissue Banks	90
3.3.2.4	4 Regulations Relating to Stem Cell Banks	91
3.3.2.5 Regulations Regarding the General Control of Human bodies, Tissue, Blood,		
Blood	Products and Gametes	92
3.3.2.0	6 National Material Transfer Agreement of Human Biological Material	93
3.3.3	Protection of Personal Information Act 4 of 2003	95
3.3.4 Intellectual Property Rights from Publicly Financed Research and Development Act (IPR Act) 51 of 2008		
3.3.5	Patent Act 57 of 1978	101
3.3.6	Department of Health Ethics in Health Research Guidelines 2015	103
3.4	Foreign case law	104
3.5	Conclusion	111
CHAP	PTER 4	
BENE UGAN	FIT SHARING MODULES IN THE UNITED KINGDOM, AUSTRALIA AN NDA	ID
4.1 Int	troduction	113
4.2 Et	hico-legal benefit sharing frameworks for health research using HBMs	116
4.2.1	United Kingdom	116

4.2.1.1 Human Tissue Act of 2004 (HT Act)	116
4.2.1.2 Human Tissue Authority (HTA)	117
4.2.1.3 Human Research Authority (HRA)	118
4.2.1.4 Medical Research Council (MRC)	119
4.2.2 Australia	121
4.2.2.1 The National Statement on Ethical Conduct in Human Research	
(The National Statement)	121
4.2.2.1.1 Research involving marginalised communities	123
4.2.2.1.2 Research involving international collaborations	123
4.2.2.2 Australian Code for the Responsible Conduct of Research (The Code)	123
4.2.3 Uganda	124
4.2.3.1 Constitution of Uganda	125
4.2.3.2 National Guidelines for Research Involving Humans as Research Particip (National Guidelines)	ants 125
4.2.3.2.1 Research involving international collaborations	126
4.2.3.2.2 Research HBMs and ownership of human biological samples	126
4.2.3.3 National Research Biobanking Guidelines	127
4.2.3.3.1 Intellectual Property	128
4.2.4 South Africa	129
4.2.4.1 Department of Health Ethics in Health Research Guidelines, 2015	129
4.2.4.2 Health Professions Council of South Africa (HPCSA)	130
4.2.4.3 National Material Transfer Agreement for Human Biological Material	130
4.3 Comparative analysis	131
4.3.1 Table 1: Positions on ownership of HBMs and benefit sharing in Australia, t	he
UK, Uganda and South Africa	137
4.4 Conclusion	138
CHAPTER 5	
BENEFIT SHARING AGREEMENTS AND A TEMPLATE FOR BENEFIT SHAR FOR HEALTH RESEARCH IN SOUTH AFRICA	ING
5.1 Introduction	140
5.2 Legal frameworks for benefit sharing	141
5.2.1 Nagoya Protocol on Access to Genetic Resources and Fair and	
Equitable sharing of benefits arising from their utilisation	142
5.2.2 HUGO Ethics Committee Statement on Benefit Sharing	142
5.2.3 CIOMS International Ethical Guidelines for Health-related	
Research involving Humans	143
5.2.4 San Code of Research Ethics	143

5.3 Challenges to benefit sharing	144
5.4 Proposed benefit sharing models	147
5.4.1 Table 2: Types of benefits that accrue to various stakeholders in a research project	148
5.4.2 Reasonable availability	148
5.4.3 Fair benefits model	149
5.4.4 Private benefit sharing model	149
5.4.5 Altruistic model of benefit sharing	150
5.4.6 Charitable trust model	151
5.4.7 H3Africa's capacity building model	152
5.4.8 Table 3: Benefit sharing models limited to LMICs	153
5.4.9 Ethical benefit sharing model that includes multiple stakeholder benefits	154
5.4.10 Table 4: Summary of the elements of the two-dimensional benefit sharing 158 framework	
5.5 Proposed benefit sharing model for health research in South Africa	156
5.5.1 Annexure A: Proposed benefit sharing agreement framework	158
5.6 Conclusion	161
CHAPTER 6	
CONCLUSION AND RECOMMENDATIONS	
6.1 Introduction	163
6.2 Recommendations	165
6.2.1 Benefit and benefit sharing in the context of health research: underlying ethico-legal norms	165
6.2.2 Historical concept of benefit sharing	167
6.2.3 Review and evaluation of existing ethical and legal frameworks relevant to Sou Africa	
6.2.3.1 Ownership of HBMs in South Africa	170
6.2.3.2 National Health Act 61 of 2003 and relevant Regulations	171
6.2.3.3 National Material Transfer Agreement for Human Biological Material	172
6.2.3.4 Protection of Personal Information Act 4 of 2013	173
6.2.3.5 Intellectual Property Rights from Publicly Funded Research and	
, , ,	174
6.2.3.6 Patent Act 57 of 1978	175
6.2.4 Department of Health Ethics in Health Research Guidelines, 2015	176

6.2.5 Existing regional and international benefit sharing frameworks	176
6.2.5.1 United Kingdom	176
6.2.5.2 Australia	178
6.2.5.3 Uganda	179
6.2.6 Key elements in the development of a benefit sharing agreement	
for health research in South Africa	180
6.2.6.1 Legal frameworks for benefit sharing	181
6.2.6.2 Challenges to benefit sharing	182
6.2.6.3 Proposed benefit sharing models	183
6.2.6.4 Proposed model for a benefit sharing agreement for human health research in	
South Africa	187
6.4 Conclusion	187
7. BIBLIOGRAPHY	
7.1 Books	190
7.2 Case law	192
7.3 Theses and dissertations	193
7.4 Journal articles	193
7.5 South African legislation	206
7.6 South African policies and guidelines	207
7.7 Foreign legislation	207
7.8 Foreign guidelines	207
7.9 International and regional instruments and guidelines	209
7.10 Internet sources	210
7.11 Newspaper articles	217
7.11 Newspaper articles 7.12 Miscellaneous sources	217 218
• •	

CHAPTER 1

BACKGROUND TO THE STUDY, PROBLEM STATEMENT, OBJECTIVES AND FRAMEWORK

1.1 Background, literature review and problem statement

Healthcare research conducted in developing countries, including South Africa has exploded in the last decade.¹ According to the Global Forum for Health Research:²

Every year only a 10th of global expenditure is spent on health research and development by the public and private sectors to address issues that affect the poorest 90% of the world's population. This is what is called the 10/90 gap.³

Although this 'research explosion' is a welcome occurrence for neglected diseases, it has brought a unique set of problems, mostly regarding the fear of and actual exploitation of host communities.⁴ It is assumed that this research is safeguarded by ethical guidelines and regulations that provide guidance to sponsors and researchers. However, it has become increasingly difficult to protect communities and individuals when healthcare research becomes a privately funded business.⁵ It also emerged from North–South healthcare research collaborations based at academic institutions that most of the donors are foreign government agencies.⁶ Thus, there are indeed genuine concerns about potential exploitation through the unfair distribution of risks and benefits among the parties involved.

It is against this background that the concept of benefit sharing arises. In reality, research in lower- and middle-income (LMIC) countries is frequently conducted in communities that have often been exploited by higher-income countries (HICs). This is particularly evident

¹Lairumbi *et al* "Forms of benefit sharing in global health research undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities' in medicine* 7(1):1–8.

²Global Forum for Health Research. The 10/90 report on health research 2000.Geneva: Switzerlandhttps://apps.who.int/iris/bitstream/handle/10665/66474/Global_Forum_for_Health_Research_e ng.pdf?sequence=1&isAllowed=y (Date of use: 20 September 2023).

³Global Forum for Health Research. The 10/90 report on health research 2000. Geneva. Switzerland ⁴Evans N.G., Hills K and Levine A.C. "How should the WHO Guide Access and Benefit Sharing during

Infectious Disease Outbreaks" 2020 AMA Journal of Ethics 22(1):28-35.

⁵Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: A new wave of bioexploitation under the guise of clinical care" 2016 *South African Medical Journal* 106(2):136-138.

⁶H3A Consortium, Rotimi C, Abayomi A *et al* "Research capacity. Enabling the genomic revolution in Africa" 2014 *Science* 344 (6190):1346–1348.

in South Africa, a country that has not yet recovered from the ravages of Apartheid, during which communities were marginalised into distinct geographical locales along racial lines.⁷ Adding to South Africas' socio-economic challenges, are the fact that inequality amongst its citizens remains the highest in the world; electricity supply shortages are prevalent; structural challenges have increased; and that the country has shown persistent weak growth, not to mention that the ravages of the COVID-19 pandemic are still evident.⁸ Inequitable and/or neocolonial practices and power imbalances are evident in many ongoing research collaborations between the global North and global South.⁹ There is growing consensus by researchers in the global North, as they become sensitised to global inequities in research practices generally, that research sponsors from industrialised countries are obliged to provide benefits to the research participants and host communities whenever they conduct research in LMICs.¹⁰

The mapping of the entire human genome brought about a staggering increase in human genomic and biobank research globally, including in South Africa. This includes extensive North—South collaborations in these genomic studies with a growing number of studies being conducted on the African continent.¹¹ Genomic research and biobanking depend on vast amounts of samples and data from diverse populations. African genetic diversity is unmatched,¹² making African samples and data highly valuable. Problems arise when there is a unidirectional flow of these samples and data out of Africa, particularly from South Africa, without proper legal protection to prevent or at least minimise the exploitation of vulnerable communities.¹³

-

⁷Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440. 8https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpo-

zaf.pdf (Date of use: 23 September 2023).

⁹Bedeker A, Nichols M, Allie T et al., "A framework for the promotion of ethical benefit sharing in health research" 2022 BMJ Global Health 7(2): e008096.

¹⁰Bedeker A, Nichols M, Allie T et al., "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2): e008096.

¹¹Guardasani D, Carstensen T, Tekola-Ayele F *et al* "The African Genome Variation Project shapes medical genetics in Africa" 2015 *Nature* 517:327–32.

¹²Tishkoff SA, Reed FA, Friedlaender FR *et al* "The genetic structure and history of Africans and African Americans" 2009 *Science* 324:1035–44.

¹³Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 BMC *Medical Ethics* 17(1):1–9.

Considering the historical injustices brought about by a lack of integrity, cultural insensitivities and unfair collaborations, ¹⁴ it is not surprising that the exploitation of local researchers and populations has ensued. ¹⁵ Hence, the concept of benefit sharing must be taken very seriously. An international collaborative genetic research project undertaken with villagers living in Anhui, an extremely impoverished province in China, ¹⁶ is an appropriate example of such exploitation. This study was conducted between 1995 and 2000. It involved a collaboration between researchers from Harvard University in the United States, local research institutes and the Chinese government. ¹⁷ At the time, the Chinese government was aggressively driving the agenda for international research collaborations in order to robustly advance the science and technology fields in the country. This eagerness by Chinese academics, institutions and government led to scientific research institutions and researchers from HICs taking advantage of the host communities and local researchers. ¹⁸

In 2002, a report surfaced, indicating that a Chinese-American scientist, with the assistance of a grant from the National Institutes of Health (NIH) and biopharmaceutical companies, had been collecting blood samples from the villagers in the Anhui province for five years. ¹⁹ The blood samples were transferred to a US biobank for research into several diseases. Pharmaceutical companies invested a considerable amount of money in this research as they envisaged a lucrative market in drugs as the end product. ²⁰ The

-

¹⁴Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 BMC *Medical Ethics* 17(1):1–9.

¹⁵O'Daniel J and Haga SB "Public perspectives on returning genetics and genomics research results" 2011 Public *Health Genomics* 14(6):346–55.

¹⁶Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in Schroeder D, Cook Lucas J et al (eds) Ethics dumping, case studies from North-South research collaborations (Springer International Publishing: Cham, 2018) 71–80.

¹⁷Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in Schroeder D, Cook Lucas J *et al* (eds) *Ethics dumping, case studies from North-South research collaborations* (Springer International Publishing: Cham, 2018) 71–80.

¹⁸Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in Schroeder D, Cook Lucas J *et al* (eds) *Ethics Dumping, Case Studies from North-South research collaborations* (Springer International Publishing: Cham, 2018) 71–80.

¹⁹Pomfret J and Nelson D "An isolated region's genetic mother lode" 2000-12-20 Washington Post https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-mother-lode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

²⁰Pomfret J and Nelson D "An isolated region's genetic mother lode" 2000-12-20 Washington Post https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-mother-lode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

participant communities did not know why their blood samples were being collected.²¹ These genetic harvest experiments were later found to have compromised research ethics, for example, the recruitment of 16 686 participant farmers instead of 2000 in the approved protocol of an asthma study, the drastic reduction of monetary compensation to the farmers and the use of bronchodilators that were different from the approved ones in the protocol.²² No ethics committee had sanctioned the collection of the participants' blood. There was also no prior informed consent by the participants and very little benefit to them. Harvard University benefitted significantly from monies received from the NIH and pharmaceutical companies. In March 2002, the US Department of Health and Human Services found that Harvard University's genetic project in China had violated multiple regulations in ethics, participant safety and supervision and management.²³ The Chinese government, via the ministries concerned, tried to limit and halt exportation of the genetic samples. However, it was too late, as Harvard University already had an enormous amount of Chinese DNA samples in its biobank.

Lower- and middle-income countries often do not realise their full potential in genetic research due to lack of specific expertise and technical and manufacturing capabilities. Nevertheless, it is critical that given the history of exploitation in LMICs, the law cannot be silent about genetic research and biobanks. South African law has not addressed issues around such research until very recently, when the National Material Transfer Agreement (MTA) was gazetted in July 2018.²⁴ The issue around access to genomic resources and benefit sharing are relevant in modern-day South Africa because of an increasing number of research collaborations between South African universities and universities in HICs countries, as well as their respective government departments.²⁵ These collaborations have given rise to complex ethical, legal and social issues.²⁶

²¹Pomfret J and Nelson D "An isolated region's genetic mother lode" 2000-12-20 Washington Post https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-mother-lode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

²²Xiong L and Wang Y "Harvard University's genetic research in China is illegal" 2002 *Outlook Weekly* 15:48–50

²³Xiong L, Wang Y "Harvard University's genetic research in China is illegal" 2002 *Outlook Weekly* 15:48–50.

²⁴South Africa National Health Act No. 61 of 2003 Material Transfer Agreement for Human Biological Materials, Government Gazette No. 41781 2018.

²⁵Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

²⁶Dhai A, Mahomed S and Sanne I "Biobank and human health research: balancing progress and protections" 2015 *S Afr J BL* 9:55–59.

Human biobanks can store and distribute human biological materials (HBMs) in perpetuity. These materials include DNA, RNA, blastomeres, polar bodies and human tissues²⁷ and their associated data for the purposes of health research, which has precipitated ethicolegal issues such as, but not limited to, benefit sharing that arises from the primary and secondary use of such samples.²⁸

It is not improbable to imagine that due to rapid developments in biotechnology, large amounts of tissue samples may be leaving South Africa and the region for health research conducted in HICs in a questionable manner.²⁹ The reason for this is that biobanks are guided by a very flexible legal framework. The Department of Health (DoH) Ethics in Health Research Guidelines,³⁰ and the recently gazetted Material Transfer Agreement (MTA) for Human Biological Materials,³¹ attempt to legally regulate biobanks in South Africa. However, because the tissue bank regulations promulgated in terms of Chapter 8 of the National Health Act (NHA) refer to tissue banks but do not mention biobanks, researchers and clinicians tend to rely on the regulations for tissue banks, which, in turn, have been deemed inadequate.³²

Considering South Africa's background of exploitation in research and indeed, that of many other African countries, it is imperative that these governments establish a template for benefit sharing agreements that outlines how benefits from health research should be shared, with whom and through what mechanisms. These benefit sharing agreements should be based on justice in an attempt to remedy past exploitation and uphold public interest. In addition, these agreements must also respect local traditions.³³

²⁷Republic of South Africa. Regulations relating to the use of Human Biological Materials No R177 Pretoria. Government Gazette 2012: Section 1.

²⁸Mahomed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8(Suppl 1):60-64.

²⁹Mahomed S, Behrens K, Slabbert M and Sanne I "Managing human tissue transfer across national boundaries-an approach from an institution in South Africa" 2016 *Dev World Bioeth* 16:39–35.

³⁰Department of Health, South Africa. Ethics in Health Research: Principles, Processes and Structures. 2nd ed Pretoria DoH.

³¹Republic of South Africa. Regulations relating to the use of Human Biological Materials No R177 Pretoria Government Gazette. 2012: Section 1.

³²Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediment to accessing healthcare" 2019 *S Afr Bioethics Law* 12(1):27–31.

³³Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

Simm's claims that a universal consensus exists between researchers, public health officials amd research participants. regarding the need to share benefits arising from collaborative research.³⁴ The problem arises when defining a 'fair benefit' since it is unclear what the nature of such benefits should be. Would it be to address justice at the micro level (relating to individuals participating in the research) or justice at the macro level (meaning for the common good of all)³⁵, or at both levels? It is also important to question when, how and with whom benefits should be shared in a study, as well as what such a benefit would entail.

Dauda and Dierickx³⁶ maintain that the concept of benefit sharing is almost always accompanied by controversies and contradictions associated with what the notion entails and what its definition is. Benefit sharing in health research is the process or act of sharing the benefits that derive from research fairly and equitably.³⁷ Benefit sharing is considered to be one of the important benchmarks for ethical research in LMICs.³⁸ Most people in these countries live in poverty and have no access to decent medical care. It seems fair that any research conducted should improve the quality of their healthcare.

Therefore, questions arise about exactly what a benefit is and how it should be justified, especially in a world where research is meant to be altruistic in nature and for the advantage of all humankind. In the field of human genetic research, for example, the Human Genome Organisation's Committee on Ethics, Law and Society defines a benefit as follows:

A benefit is a good that contributes to the well-being of an individual and/or a given community (e.g., by region, tribe, disease-group ...). Benefits transcend avoidance of harm (non-maleficence) insofar as they promote the welfare of an individual and/or of a community. Thus, a benefit is not identical with *(sic)* profit in the

³⁴Simm K "Benefit-sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

³⁵Simm K "Benefit-sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics*, *Society and Policy* 1(2):29–40.

³⁶Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

³⁷Mohamed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8:60–64.

³⁸Emmanuel EJ, Wendler D, Killen J and Grady C "What makes clinical research in developing countries ethical? The benchmarks of ethical research" 2004 *J Infect Dis* 189:930–937.

monetary or economic sense. Determining a benefit depends on needs, values, priorities and cultural expectations.³⁹

The exact nature and specification of a benefit is a very controversial topic. ⁴⁰ According to Dauda and Joffe, ⁴¹ two models of benefit sharing dominate the ethical debate. These models have their roots in the concept of the common heritage of humankind that was used to guide the idea of benefit sharing when it first surfaced. With the generalised view that the underlying purpose of all studies is the creation of generalizable knowledge, ⁴² Dauda and Dierickx explain that the idea of the common heritage of humankind evolved from the doctrine of *res communis*, which directs that resources obtained from common heritage territories are not meant to be monopolised, possessed or owned by individuals, communities or the state. The use of such resources has to be subjected to the rights and interests of all humankind. ⁴³ This idea promotes equal sharing of all resources, suggesting that benefit sharing should be used to address the differences between HICs and LMICs. ⁴⁴ The authors note that a benefit need not be tangible but could be in another form, such as technology transfer, which would result in capacity building.

Treaties that emphasise benefit sharing in the context of the common heritage of humankind include the United Nations Convention on Law of the Sea (UNCLOS)⁴⁵ and the International Undertaking on Plant Genetic Resources (IUPGR).⁴⁶ The UNCLOS treaty strives for sharing benefits from the seas, regardless of whether the country is landlocked or coastal. It also urges other countries to consider the vulnerability of developing countries in accessing benefits in Article 140, Paragraph 1, 13.⁴⁷ The IUPGR stipulates that the genetic resource of plants should be for the use of all humankind to benefit present and

³⁹HUGO Ethics Committee:

https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,the%20issue%20of%20benefit%2Dsharing (Date of use: 22 May 2023).

⁴⁰Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

⁴¹Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 Developing World Bioeth 18:165–170.

⁴²Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170.

⁴³Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170.

⁴⁴Basler K *The concept of the common heritage of mankind in international law* (Published by Nijhoff Publishers 1998).

⁴⁵UN Convention on the Law of the Sea publication

https://www.imo.org/en/ourwork/legal/pages/unitednationsconventiononthelawofthesea.aspx#:~:tex t=The%20United%20Nations%20Convention%20on,the%20oceans%20and%20their%20resource s (Date of use: 23 May 2023).

⁴⁶FAO: International undertaking on Plant and Genetic Resources Rome: Electronic Publishing 1983:10.

⁴⁷United Nations: *United Nations Convention on the Law of the Sea* Montego Bay: United Nations Publication 1982:71.

future generations.⁴⁸ The proposal stated in the IUPGR, coupled with increasing cases of bioprospecting and concern from developing countries regarding the exploitation of indigenous genetic resources without fair and just compensation, render this benefit sharing model problematic to implement and it also allows for the adoption of sovereign rights to biodiversity.⁴⁹

This leaves two benefit sharing models: The first model is rooted in reciprocity and is known as the reasonable availability model. This model argues that research benefits for host communities must emerge from the research findings⁵⁰. The model is endorsed by the Council for International Organisations of Medical Sciences (CIOMS).⁵¹ It is argued that in this model, communities involved in clinical research are susceptible to exploitation and need guarantees that they will benefit from the research. Critics of the model argue that this model is restricted to tangible products only, which, in itself, is a form of exploitation.⁵²

The second, the fair benefit model, is similar to the reasonable availability model but differs in that it argues that there are more ways than just tangible benefits for sponsors to meet their benefit obligations. Furthermore, the fair benefit model asserts that the host communities should be allowed to determine the fairness of the benefits that are to be provided. This fairness model is typically exemplified by the Majengo sex workers case study in Kenya. The research involved the Majengo Observational Cohort Study (MOCS) and started in the late 1980s in the Majengo slum in Nairobi. Funded by the Canadian government and the Public Health of Nairobi City council, it was thought the sex workers could aid in the development of a vaccine against HIV. A clinic was set up to provide basic outpatient medical care to female sex workers. This clinic also served as a research facility for collaborative research in HIV vaccine development as it was discovered that some of these sex workers had developed immunity to HIV, despite long-term exposure to

_

 ⁴⁸FAO: International undertaking on plant and genetic resources Rome: FAO Electronic Publishing 1983:10.
 ⁴⁹Dauda D and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 BMC Medical Ethics 14(1):1-8.

⁵⁰Participants in the 2001 Conference on Ethical Aspects of Research in Developing Countries "Moral standards for research in developing countries: from reasonable availability to fair benefits" 2004 *Hastings Cent rep* 34:17–27.

⁵¹CIOMS in collaboration with the World Health Organisation "International Ethical Guidelines for Biomedical Research involving Human Subjects" 2016 Geneva.

⁵²Wolitz R, Emanuel EJ and Shah S "Rethinking the responsiveness requirements for international research" 2009 *Lancet* 2374: 847–49.

⁵³Wolitz R, Emanuel EJ and Shah S "Rethinking the responsiveness requirements for international research" 2009 *Lancet* 2374: 847–49.

⁵⁴Andada P and Cook Lucas J HIV/AIDS Research Case. A Report for GenBenefit, 2007 https://www.uclan.ac.uk/genbenefit (Date of use: 23 May 2023).

HIV. Blood, cervical, vaginal and saliva samples were drawn from the women with their consent and used to study the epidemiology and immunology of HIV. In 2005, national guidelines for the research and development of HIV/AIDS vaccines were developed in specific response to this case in Kenya.⁵⁵ The guidelines provide for the "fair and equitable sharing of benefits" arising from research results attained from biological materials.⁵⁶ In this regard, the fair benefit was that the socioeconomically disadvantaged sex workers received access to healthcare and free antiretrovirals while the researchers obtained sound research results.

It is worth noting that the proponents of the fair benefit model also argue that the benefits to the community and/or participant should be directly proportional to the risks associated with the research.⁵⁷ In their opinion, those who contribute more and bear more risks should receive more of the benefits from the collaboration.⁵⁸

Benefit sharing has been established as a principle of international law in the context of non-human genetic resources via the Nagoya protocol,⁵⁹ a supplementary agreement to the Convention on Biological Diversity (CBD).⁶⁰ South Africa is a signatory to the Nagoya protocol, which is a binding agreement regarding the benefit sharing of non-human genetic resources.⁶¹ The guidelines relating to human genetic material represent a shift away from a concept that is protected and bound by law, towards non-binding regulations, as documented in the UNESCO Declaration on the Human Genome and Human Rights (1997),⁶² the HUGO Ethics Committee Statement on Benefit Sharing (2000),⁶³ the

⁻

⁵⁵Kenyan Ministry of Health. Kenyan Ministry of Health National Guidelines 2005 https://www.globalgiving.org/pfil/1108/projdoc.pdf (Date of use: 23 May 2023).

Ministry of Health. National Kenyan Guidelines for Research and Development of HIV/AIDS Vaccines Appendix 5: Biological Material Transfer Agreement. Kenya: Ministry of Health 2005 http://www.globalgiving.org/pfil/1108/projdoc.pdf (Date of use: 23 May 2023).

⁵⁷Lie RK "Fair benefit approach revisited" 2010 Hastings Cent rep. 40(4):3–3.

⁵⁸Lie RK "fair benefit approach revisited" 2010 Hastings Cent rep. 40(4): 3-3.

⁵⁹Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 23 May 2023).

⁶⁰The Convention on Biological Diversity 1992. http://www.cbd.int/abs/about (Date of use: 23 May 2023).

⁶¹Mohamed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8:60–64.

⁶²UNESCO: UNESCO Declaration on Human Genome and Human Rights. Paris, 1997:1.

⁶³HUGO Ethics Committee: HUGO urges genetic benefit sharing. Community Ganet 2000, 3:88–92.

UNESCO Declaration on Bioethics and Human Rights (2005)⁶⁴ and the Helsinki Declaration of the World Medical Association (WMA, 2013).⁶⁵

Dauda and Dierickx⁶⁶ suggest that benefit sharing in international health research should be formulated into a legal framework. The legal framework is necessary because benefit sharing is mostly ignored even though it is known to be an ethically sound concept. This recommendation is sound and resonates with the aims and objectives of this study, which seeks to recommend an ethico-legal framework for benefit sharing in South African health research. Health research in South Africa is governed by the National Health Act (NHA).⁶⁷ The National Health Research Ethics Council (NHREC) was formed under the mandate of section 69(1) of the NHA and is tasked with the responsibility of determining guidelines for the functioning of health research ethics committees.⁶⁸

It is worth noting that although the NHA was proclaimed in 2003, certain Regulations in Chapter 8 of the NHA which deal with human biological materials (for example, blood, blood products, tissue and gametes) were only enacted in 2012.⁶⁹ Furthermore, the government of South Africa only gazetted a national template of a Material Transfer Agreement of Human Biological Materials (HBMs) under the NHA in 2018.⁷⁰

South African law does not legally enforce policies, ethical guidelines and local documents, although these may point to instances of professional misconduct in the context of health research, which may have legal consequences. Only legislation, regulations promulgated in terms of legislation, and judicial precedent are legally binding and enforceable. However, under current legal literature, ethical guidelines are considered 'soft law' or customary international law in the case of international ethical guidelines.⁷¹ International instruments may be referred to when local laws are unforthcoming, but a judge is not obliged to

⁶⁴UNESCO Declaration on Bioethics and Human Rights https://en.unesco.org/about-us/legal-affairs/universal-declaration-bioethics-and-human-rights#:~:text=and%20human%20rights-,1.,interest%20of%20science%20or%20society (Date of use: 23 May 2023).

⁶⁵World Medical Association Declaration of Helsinki - Ethical Principles for Medical Research Involving Human Subjects. Accessed August 28 2019. http://www.wma.net/policies-post/wma-declaration of helsinki-ethical-principles-for-medical-research-involving-human-subjects/. (Date of use: 23 May 2023).

⁶⁶Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

⁶⁷⁶¹ of 2003, Chapter 9.

⁶⁸Section 72(6)(a) of the NHA.

⁶⁹Proclamation No 11 Government Gazette 35081 of 27 February 2012.

⁷⁰Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷¹Andanda P, Schroeder D, Chaturvedi S *et al* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Lucas JC (eds) *Benefit sharing: from biodiversity to human genetics* (New York Springer 2013) 333–364.

consider these guidelines, despite the fact that these guidelines may have persuasive force. The National Department of Health's Ethics in Health Research Guidelines of 2004 derive from section 72 of the NHA, mandating the NHREC to table ethical guidelines concerning health research involving human participants. These ethical guidelines have been endorsed in the 2014 Regulations relating to Research with Human Participants. The guidelines were revised in 2015, regulation 2(a) of the revised 2015 guidelines state that there are the minimum benchmark or standard to be followed in health research involving human participants, thereby affording the 2015 guidelines legal standing.

The South African Biodiversity Act 10 of 2004⁷⁶ is the only piece of legislation that regulates benefit sharing agreements in terms of illustrating and defining what the agreement should contain.⁷⁷ The Act undeniably *excludes genetic material of human origin* in section 80(2) (b)(i); rather, it applies to bioprospecting and research which involve indigenous biological resources. Chapter 6 of the Act prohibits the removal of any biological material and any bioprospecting from and in South Africa unless a permit has been granted.⁷⁸ For this permit to be granted, section 82 of the Act requires the applicant to conclude a material transfer agreement (MTA) with the stakeholder (which could be a person, organ of state, community or indigenous community). The MTA should regulate the provision of or access to the biological resources and a benefit sharing agreement that provides for sharing by the stakeholder in any future benefits that may be derived from the relevant bioprospecting.⁷⁹ The sitting minister responsible for national environmental management must approve the benefit sharing arrangement. Chapter 6 of the Act also allows the issuing authority to mediate negotiations between the applicant and stakeholder fairly and equitably.

Similarly, the South African national MTA relating to the use and transfer of human biological material (HBM) provides in section 7 that the sharing of benefits should be

⁷²The Constitution of South Africa 1996, Section 39 (1).

⁷³South Africa. National Health Act No 61.2003, Section 72.

⁷⁴South Africa. National Health Act No 61.

⁷⁵South Africa. Regulations relating to Research with Human Participants. GR No R719 in Government 3800 of 19 September 2014.

⁷⁶Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

⁷⁷Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

⁷⁸Section 81.

⁷⁹Section 82(2).

discussed and negotiated solely between the provider and recipient before the materials are transferred to the recipient.

The Biodiversity Act, although excluding human biological material, provides a template for a legally binding benefit sharing arrangement that could be adjusted or developed for research involving Human Biological Materials, be it tissue, genetic material or clinical research. A legally binding benefit arrangement would instil a culture of trust and provide for fair and equitable research collaborations.

The concept of benefit sharing has received considerable attention in South Africa with regard to *non-human* genetic material. This is best exemplified by the San Code of Ethics, which defines how researchers are to interact when dealing with the San peoples of Southern Africa,⁸⁰ following the San Hoodia case of 2003.⁸¹ In the absence of legal guidelines, the San people were able to benefit from the South African Research Council for Scientific and Industrial Research (CSIR) and the pharmaceutical company Phytopharm in the form of royalties, for their patenting of the *Hoodia gordonii* plant, a plant that is indigenous to the home of the San people, the Kalahari Desert, and which has been used by the San people to curb hunger for centuries.⁸²

There is a paucity of legal literature in South Africa relating to benefit sharing agreements involving human biological material in general. In 2010, following a publication by *Nature*⁸³ regarding the sequencing of the genomes of Archbishop Desmond Tutu and a few Khoisan individuals, Slabbert and Pepper⁸⁴ commented on the troubling nature of the lack of legally binding benefit sharing frameworks in South Africa and the concept of genomic sovereignty.

The regulation of access to and benefit sharing of human genetic resources in South Africa was also examined in a 2011 paper. This paper concluded that despite an enormous number of genomic studies occurring in South Africa, there is an absence, both

⁸⁰Chennelis R and Steenkamp A "International genomics research involving the San people" in Schroeder D, Cook Lucas J et al (eds) Ethics dumping, case studies from North-South research collaborations (Springer International Publishing 2018).

⁸¹Chennels R and Steenkamp A "International genomics research involving the San people" in Schroeder D, Cook Lucas J *et al* (eds) *Ethics dumping, case studies from North-South research collaborations* (Springer International Publishing 2018).

⁸²The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

⁸³ Nature 2010 463-857.

⁸⁴Slabbert MN and Pepper MS "A room of our own?' Legal lacunae regarding genomic sovereignty in South Africa" 2011 *73 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 432–50.

internationally and locally, of a clear regulatory framework that guides these studies in drawing up benefit sharing arrangements that are agreeable to all parties in research regarding human genetic material. ⁸⁵

In 2015, Mahomed and Sanne⁸⁶ debated the requirement of benefit sharing in the context of genetic research. Citing the Majengo and the San Hoodia cases, they concluded that these two cases could be cited as examples of successful benefit sharing arrangements that could be used to form a template for legally binding benefit sharing arrangements. As recently as 2013, Sathar,⁸⁷ Dhai *et al* conducted a retrospective study regarding compliance with national and international guidelines involving the use, collection, storage, transfer and benefit sharing of HBM in collaborative research between South Africa and HICs. When reviewing the records of one South African institution's research ethics committee specifically, the authors found many instances of HBMs leaving South African borders, without the legally required export permits and MTAs, to international destinations. This paper recommended that benefit sharing in collaborative research using HBMs is best addressed through MTAs, which have been a legal requirement for any collaborative research project in South Africa since July 2018.

Moodley and Singh⁸⁸ recently cautioned on the risks of research participants' lack of trust, in the context of biobanking in South Africa, and that the absence of a clear legal framework and national legislation governing simple processes from sample collection to sample export are seriously impinging on the biobanking revolution.

Earlier in 2014, Chennells⁸⁹ emphasised the need for equitable access to human biological resources in developing countries, which includes ensuring that benefit sharing in genomic research is implemented to prevent exploitation in resource-poor countries.

⁸⁵Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 74 *Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

⁸⁶Mahomed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8 (2)(1):60–64.

⁸⁷Sathar A, Dhai A and Van der Linde S "Collaborative international research: ethical and regulatory issues pertaining to human biological materials at a South African institutional research ethics committee" 2014 *Dev World Bioethics* Vol 14:150–157.

⁸⁸Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

⁸⁹Chennells R Equitable access to human biological resources in developing countries: benefit sharing without undue inducement (PhD Lancashire University 2014).

One year later, Moodley's study⁹⁰ provided recommendations on how the governance of biobanks in South Africa could be developed to engender trust in potential donors and communities in the context of international collaborative health research. She explained that in health research, good governance is rarely employed as it is in the corporate world.⁹¹

The backdrop of this study, as well as the review of South African scholarly literature on benefit sharing discussed above, point to a problem statement that requires critical legal analysis, namely that there is currently no prescribed benefit sharing model or template in South African health research. This dire shortcoming, as contextualised above, provides the context for the rationale of this study, which is discussed next.

1.2 Rationale of the study

Considering the legislative vacuum outlined above, namely the absence of a legally prescribed template or guidelines directing benefit sharing agreements relating to human biological material, as well as the related challenge of unmonitored movements in the region and internationally of HBMs, tissue samples and valuable research data during health research collaborations, the need for a legal response in this regard is clear and justified. In a world where inequitable North–South research collaborations have become commonplace, the risk exists that South Africa, as one of the most genetically diverse populations, may not be able to benefit from its genetic diversity—at the expense of its people.

The global explosion of genomic research and biobanking makes it imperative that the law and ethics are not silent on protecting South African researchers and research participants from the risks of exploitation and unfair research agreements. There is a definite need to provide the best possible legal and ethical framework and national legislation to govern simple processes, from sample collection to sample export.

1.3 Research question

In South Africa, there are currently no national guidelines governing benefit sharing in health research, specifically genetic health research, which is the focus of this study. The research question of this study explores which benefit sharing model would best regulate

⁹⁰Moodley K Legitimacy, trust and governance in biobanking in South Africa (MBA UCT 2015).

⁹⁰Moodley K Legitimacy, trust and governance in biobanking in South Africa (MBA UCT 2015).

⁹¹Moodley K Legitimacy, trust and governance in biobanking in South Africa (MBA UCT 2015).

benefit sharing in the context of South African health research from an ethico-legal perspective. In addition, by exploring this question, this study hopes to provide a robust foundation for further research questions relating to benefit sharing in the context of health research.

1.4 Aims and objectives

This study aims to critically analyse the current ethical and legal guidelines in South Africa that have a bearing on benefit sharing or benefit sharing agreements in health research, intending to recommend an ethico-legally justified benefit sharing model for South Africa. This model should serve to protect South African institutions, researchers, research participants or research subjects and communities while, at the same time, promoting collaborative research.

The above-mentioned aim of the study requires consideration of the following objectives related to the aim of the study:

- (1) To determine what is meant by a 'benefit' and 'sharing' in the context of health research generally, as well as ethical and legal norms and values that inform the concept of benefit sharing.
- (2) To explore the concept of benefit sharing from a historical perspective.
- (3) To address the controversial question of the ownership of human biological materials in South Africa.
- (4) To critically review and evaluate current ethical and legal frameworks that have a bearing on benefit sharing in health research in South Africa.
- (5) To critically analyse existing regional and international benefit sharing agreements to identify guiding principles for the best possible benefit sharing agreement.

1.5 Expected outcomes of the study

It is morally just that those who participate in research should benefit from the process. Benefit sharing is important in research programmes as it ensures equitable and just distribution of the benefits that arise from research. 92 The primary intended outcome of this study is the development of a model benefit sharing agreement for South African researchers that would protect South African research and researchers, as well as the South African population, without hindering research collaborations in health research.

⁹²Bedeker A, Nichols M and Allie T "A framework for the promotion of ethical benefit sharing in health research" 2022 BMJ Global Health 7:e008096.

This is morally imperative and it is a matter of justice if it is agreed that all who participate in research should benefit from the said research and correct the exploitation in colloborative health research between LMIC and HIC. The generic model benefit sharing agreement would be presented to the National Department of Health for use towards national policy guidelines and regulations. The benefit sharing model should highlight commercial interests, but at the same time, redress economic imbalances and accomplish the fair and equal distribution of profits (if any) and burdens of the collaborating researchers.

1.6 Methodology

This research involves legal and normative research analysis. The research was libraryand desktop-based and no human participants were involved. The study is informed by relevant laws and ethical literature relevant to the topic.

This study also adopts a qualitative research approach, drawing on national and international legislation and documents to guide the research. The study entails a critical analysis of relevant legislation, case law, journals, articles, books and internet sources.

1.7 Limitations of the study

Ethico-legal literature on benefit sharing is limited and may impose a limitation. However, as stated elsewhere in this proposal, the study aims to provide a theoretical foundation for further research on the topic of benefit sharing, which should ideally be followed by an empirical study in future.

1.8 Chapter outline

The ethico-legal analysis of benefit sharing in health research is a complex topic requiring a careful reading of relevant legal and ethical sources. The following section details the outlay of the chapters in the thesis:

Chapter 1 details the background to the research problem, the problem statement, a literature review of the topic, the research question(s), the methodology, as well as possible limitations.

Chapter 2 introduces the history and concept of benefit sharing. It attempts to best define benefits and the ethics around benefit sharing. The chapter also explains benefit sharing models in detail and the arguments in favour of and against these models.

Chapter 3 explores the ethico-legal framework relevant to benefit sharing in South Africa in different sectors of health research. The emphasis is on biobanks and genomic research benefit sharing arrangements, including an analysis of the concept of the ownership of genetic material.

Chapter 4 examines the legal framework of benefit sharing models in the United Kingdom, Australia and Uganda. These are analysed to discern the most pertinent guiding principles that may inform the development of a benefit sharing model for South Africa.

Chapter 5 analyses the key elements of benefit sharing arrangements and presents a template of a benefit sharing agreement for health research in South Africa.

Chapter 6 concludes the thesis and discusses the relevant findings and specific recommendations for local researchers entering into collaborative research with either private companies, foreign institutions and/or foreign government institutions.

CHAPTER 2

BENEFIT SHARING IN HEALTH RESEARCH

2.1 Introduction

The concept of benefit sharing around the involvement of human biological resources is a relatively new phrase that was developed and coined in the past three decades. ⁹³ Despite the many years of discussion and debate it is also a highly unresolved topic, ⁹⁴ because of intern alia there exists a belief that health research should be altruistic in nature and because the concept of benefit sharing defeats altruism, ⁹⁵ and what the concept entails and whats its definition is. ⁹⁶

The past few decades have witnessed an explosion of research in the context of healthcare research being conducted in developing countries including, but not limited to, South Africa. PNot surprisingly, there has also been a proliferation of biobanks. Reports reflect that since 2015, the global tissue engineering market accounted for an estimated USD 23.3 billion, with projections that it would exceed USD 94.7 billion in the succeeding years. Phe mapping of the entire human genome brought about a monumental increase in human genomic and biobank research globally, including in South Africa. African genetic diversity is unmatched, making African human biological samples and data highly valuable. This fact precipitated a rise in North—South collaborations in genomic studies, with a growing number of these studies conducted on the African continent. Molecular diagnostics have

⁹³Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

⁹⁴ Schroder D "Benefit sharing: it's time for a definition" 2007 *J Med Ethics* 33:205–209.

⁹⁵Berg K "The ethics of benefit sharing" 2001 *Clin Genet* 59:240–243.

⁹⁶Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

⁹⁷Lairumbi G, Parker M, Fitzpatrick R and English M "Forms of benefit sharing in global health research undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities in medicine* 7(1):1–8.

⁹⁸Schroeder D and Lucas J "Benefit sharing: from biodiversity to human genetics—an introduction" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 1–8.

⁹⁹Hexa Research. Tissue engineering market analysis, market size, application analysis, regional outlook, competitive strategies and forecasts, 2016–2024. Felton, CA: Hexa Research, 2015 http://www.hexaresearch.com/research-report/tissue-engineering-market (Date of use: 23 May 2023).

¹⁰⁰Tishkoff SÁ, Reed FA, Friedlaender FR *et al* "The genetic structure and history of Africans and African Americans" 2009 *Science* 324:1035–44.

¹⁰¹Guardasani D, Carstensen T, Tekola-Ayele F *et al* "The African genome variation project shapes medical genetics in Africa" 2015 *Nature* 517:327–32.

emerged as the leading industry within the molecular diagnostic industry.¹⁰² Market research¹⁰³ estimated the value for these products in 2020 at US\$ 9.2 billion, projected to reach US\$23.9 billion by 2030.

This health research explosion in lower- and middle-income countries (LMICs), combined with the global commercialisation of health research products, sometimes via privately funded business, ¹⁰⁴ point to the need to address the issue of benefit sharing in the context of human biological resources.

Research conducted in LMICs is often undertaken in communities that have historically been exposed to exploitation by high-income countries (HICs). As outlined in chapter one, because of the lack of a clear legislative framework governing benefit sharing in health research in South Africa, the question that arises is what should happen when donors provide their human biological resources for health research purposes. Unlike the clear and legally binding obligations of the Convention on Biological Diversity (CBD) that provide a legal framework governing the access and use of a natural resource of one sovereign country by another party, 106 no legally binding international instrument exists to guide the use of human biological materials in health research. 107

In truth, without legal obligations and protective measures, exploitation will occur. Historical injustices in such cases are often brought about by a lack of integrity, cultural insensitivity and unfair collaborations. The concept of benefit sharing in the context of research in LMICs must be taken seriously because not taking LMIC seriously is an injustice, which would perpetuate exploitation. 109

¹⁰²Chaturvedi S, Crager S *et al* "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 153–178.

¹⁰³Molecular Diagnostic market Allied Market Research https://www.alliedmarketresearch.com/molecular-diagnostics-market (Date of use:20 September 2023)

¹⁰⁴Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

¹⁰⁵Evans NG,Hills K and Levine AC "How should WHO Guide Access and Benefit Sharing During Infectious Disease Outbreaks" 2020 *AMA Journal of Ethics* 22(1): 28-35.

¹⁰⁶Mahomed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8(Suppl 1):60–64.

¹⁰⁷Schroder D "Benefit sharing: it's time for a definition" 2007 *J Med Ethics* 33:205–209.

¹⁰⁸Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

¹⁰⁹ Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

This chapter addresses the concept of benefit sharing from a historical perspective by outlining the international and national positions on benefit sharing agreements in health research. The chapter will also propose a definition of what a benefit entails and will critically discuss different models of benefit sharing.

2.1.1 Defining benefit

Benefit sharing is considered to be one of the most important factors in conducting ethical research in LMICs.¹¹⁰ The concept of benefit sharing, though necessary, is often accompanied by controversies and contradictions about what it truly comprises and how it should be defined.¹¹¹ It is evident that a benefit does not need to be monetary-based, as suggested by the Nagoya Protocol¹¹² on Benefit Sharing of Non-human Genetic Resources.

The Nagoya Protocol suggests that benefits could take the following forms (CBD 2010a, annex):¹¹³

- 1. Monetary benefits may include, but are not limited to:
- (a) Access fees/fee per sample collected or otherwise acquired;
- (b) Up-front payments;
- (c) Milestone payments;
- (d) Payment of royalties;
- (e) Licence fees in case of commercialisation,;
- (f) Special fees to be paid to trust funds supporting conservation and sustainable use of biodiversity:
- (g) Salaries and preferential terms where mutually agreed;
- (h) Research funding;
- (i) Joint ventures; and
- (j) Joint ownership of relevant intellectual property rights.

¹¹⁰Emmanuel EJ, Wendler D, Killen J and Grady C "What makes clinical research in developing countries ethical? The benchmarks of ethical research" 2004 *J Infect Dis* 189:930–937.

¹¹¹Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1)1–8.

¹¹²Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 26 May 2023).

¹¹³Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 26 May 2023).

- 2. Non-monetary benefits may include, but are not be limited to:
- (a) Sharing of research and development results;
- (b) Collaboration, cooperation and contribution in scientific research and development programmes, particularly biotechnological research activities, where possible in the Party providing genetic resources;
- (c) Participation in product development;
- (d) Collaboration, cooperation and contribution in education and training;
- (e) Admittance to ex situ facilities of genetic resources and databases;
- (f) Transfer to the provider of the genetic resources of knowledge and technology under fair and most favourable terms, including on concessional and preferential terms where agreed, in particular, knowledge and technology that make use of genetic resources, including biotechnology, or that are relevant to the conservation and sustainable utilisation of biological diversity;
- (g) Strengthening capacities for technology transfer;
- (h) Institutional capacity-building;
- (i) Human and material resources to strengthen the capacities for the administration and enforcement of access regulations;
- (j) Training related to genetic resources with the full participation of countries providing genetic resources, and where possible, in such countries;
- (k) Access to scientific information relevant to conservation and [the] sustainable use of biological diversity, including biological inventories and taxonomic studies;
- (I) Contributions to the local economy;
- (m) Research directed towards priority needs, such as health and food security, taking into account domestic uses of genetic resources in the Party providing genetic resources;
- (n) Institutional and professional relationships that can arise from an access and benefit sharing agreement and subsequent collaborative activities;
- (o) Food and livelihood security benefits;
- (p) Social recognition; and
- (q) Joint ownership of relevant intellectual property rights. 114

This list serves the purpose of showing that benefits need not always be monetary compensation.

¹¹⁴Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

Reference is often made to a fair benefit in the context of the use of human biological material. What constitutes a fair benefit in human health research? Schroeder¹¹⁵ suggests that the problem of realising what this actually means is related to the lack of a proper definition of the concept. The phrase is used in various fields where it is prominent, often with a definition that suits the concept as used in that context.¹¹⁶

As there is no uniform definition of a benefit, a comparative analysis of some definitions may offer some guidance.

The Nagoya Protocol¹¹⁷ defines benefit sharing as:

Benefits arising from the utilization of genetic resources as well as subsequent applications and commercialization shall be shared in a fair and equitable way with the Party providing such resources that is the country of origin of such resources or a Party that has acquired the genetic resources in accordance with the Convention. Such sharing shall be upon mutually agreed terms.¹¹⁸

The HUGO's Committee on Ethics, Law and Society define a benefit regarding human genetic resources as follows:

A benefit is a good that contributes to the well-being of an individual and/or a given community (e.g. by region, tribe, disease-group. [...] Benefits transcend avoidance of harm (non-maleficence) in so far as they promote the welfare of an individual and/or of a community. Thus, a benefit is not identical with profit in the monetary or economic sense. Determining a benefit depends on needs, values, priorities and cultural expectations.¹¹⁹

In the framework of the access and use of genetic resources in terms of the CBD, benefit sharing is defined as:

30

-

¹¹⁵Schroeder D "Benefit sharing: it's time for a definition" 2006 *J Med Ethics* 205–209.

¹¹⁶Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

 ¹¹⁷ Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 23 May 2023).
 118 CBD 2010a, Article 5.1.

¹¹⁹HUGO Ethics Committee Statement on benefit sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,the%20issue%20of%20benefit%2Dsharing. (Date of use : 23 May 2023).

The action of giving a portion of advantages or profits derived from the use of genetic resources or traditional knowledge to resource providers in order to achieve justice in exchange.¹²⁰

It is worth noting that the HUGO definitation of benefit sharing is contradictory to the definition in the CBD. The HUGO definition provides that a benefit need not be profit in a monetary or economic sense because its definition of benefit is based on the argument that we all share a common genetic heritage, which supports the notion that all health research should be altruistic in nature. However, the HUGO committee goes on to recommend that profit-making entities dedicate a percentage of their annual net profit to healthcare infrastructure and/or humanitarian efforts. ¹²¹ In a world where health research is increasingly profit-driven by the private sector, ¹²² it is imperative that we apply background conditions and their impact on what is beneficial in a given location to address social injustices. ¹²³ It is fair that all benefits monetary or otherwise should be considered a fair benefits in LMICs.

In the context of international human health research, the definition of benefit sharing in the framework of access and use of genetic resources in the CBD is useful. The said definition focuses on resource providers and by extension, may apply to research participants providing biological samples, including denoting what benefits resource providers (and communities) in developing countries ought to receive in compensation for their participation in research. This is in accordance with updated CIOMS research ethics guidelines that call for sponsors and researchers from HICs to negotiate research prioraties and benefits with LMIC hosts.

In South Africa, the concept of benefit is mentioned in several documents, such as the Department of Health's Ethics in Health Research Guidelines, which do not define what a

¹²⁰Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 23 may 2023).

¹²¹HUGO Ethics Committee Statement on benefit sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20recommen ds,the%20issue%20of%20benefit%2Dsharing. (Date of use : 23 May 2023).

¹²² Moodley K, Blockman M, Hawkridge AJ et al., "Hard choices: Ethical Challenges in phase 1 of COVID-19 vaccine roll-out in South Africa" 2021 *SAMJ* 111(6):554-558

¹²³Macpherson CC "Research ethics guidelines and moral obligations to developing countries: Capacity-building and benefits" 2019 *Bioethics* 33:389-385.

¹²⁴Simm K "Benefit sharing: a look at the history of an ethics concern" 2007 *Nat Rev Genet* 8(7):496–496. ¹²⁵CIOMS and WHO International Ethical Guidelines for Health-related Research Involving Humans https://cioms.ch/wp-content/uploads/2017/01/WEB-CIOMS-EthicalGuidelines.pdf. (Date of use: 23 May 2023).

"benefit' is, but state that the ratio of risk of harm to likelihood of harm shoud be favourable;¹²⁶ the HPSCA Guidelines of 2016 which also do not define what a 'benefit' entails but recommend that there should be a 'balance of burdens and benefits of research within different population groups';¹²⁷ and the Material Transfer Agreement of 2018.¹²⁸ The national MTA defines a benefit as:

Amongst others, the sharing of information; use of research results; royalties; acknowledgement of the Provider as the source of the Materials; publication rights; transfer of technology or Materials; and capacity building.

The definition of a benefit in the MTA provides a guide to what may be negotiated as a benefit between researchers and research partipants. The MTA further defines benefit sharing in section 2(3) as "the process or act of sharing in the benefits that derive from the Project in a manner that is fair and equitable." However, no specifics are given on how the arrangement should be structured and how benefit sharing should occur, leaving the researcher and the research participant in a difficult legal conundrum.

2.2 A brief overview of the recognition of benefit sharing globally and in South Africa

2.2.1 Background

Conventionally, benefit sharing has been regarded as a technical term used in human and non-human genetic resource research between the providers of the resource and those providing compensation and reward for the resource.¹³⁰

The concept of benefit sharing has gained prominence in international law, research ethics and political philosophy in the last three decades .¹³¹ Simms argues that there is a universal

¹²⁶Department of Health Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 26 May 2023).

¹²⁷HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s_for_Health_Researchers.pdf (Date of use: 26 May 2023).

¹²⁸Proclamation No 719 Government Gazette 41781 of 20 July 2018.

¹²⁹Proclamation No 719 Government Gazette 41781 of 20 July 2018.

¹³⁰Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioethics* 6: 135–143.

¹³¹Macpherson CC "Research ethics guidelines and moral obligations to developing countries: Capacity-building and benefits" 2019 *Bioethics* 33:389-395.

consensus regarding the need to share the benefits of collaborative research.¹³² Benefit sharing in health research should be the process or act of sharing the benefits that derive from the research fairly and equitably.¹³³ To reiterate, questions arise as to what exactly constitutes a benefit and with whom it should be shared.

When benefit sharing is considered in terms of justice, a fair benefit would fall under the principle of justice.¹³⁴ The principle of justice manifests in many forms. *Justice in exchange* establishes the fairness or equity of transactions¹³⁵ and is meant to address the micro level of justice (relating to individuals directly participating in the research).¹³⁶ *Distributive justice*, on the other hand, deals with the division of scarce resources between qualifying recipients,¹³⁷ which addresses the macro level of justice (for the common good of all).¹³⁸

2.2.2 Benefit sharing relating to *non-human biological resources*

Globally, the idea of benefit sharing emerged from a *distributive justice* notion.¹³⁹ Underlying this notion is the concept of the common heritage of humankind, founded upon the concept of *res communis*.¹⁴⁰ This concept directs that resources obtained from common heritage territories are not meant to be monopolised, possessed or owned by individuals, communities or the state, but rather, should all be subjected to the rights and interest of humankind.¹⁴¹ It is worth noting that this idea, while promoting the equal sharing of all resources to help balance the inequity between HICs and LMICs, seeks to clarify that

¹³²Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

¹³³Mohamed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8:60–64.

¹³⁴Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

¹³⁵Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

¹³⁶Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics*, *Society and Policy* 1(2):29–40.

¹³⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue Inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

¹³⁸Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics*, *Society and Policy* 1(2):29–40.

¹³⁹Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

¹⁴⁰Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:16–70.

¹⁴¹Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170.

a benefit need not be tangible but could be in another form, such as technology transfer and capacity building.¹⁴²

International instruments deriving from *res communis* include the United Nations Convention on Law of the Sea (UNCLOS 1982)¹⁴³ and the International Undertaking on Plant Genetic Resources (IUPGR 1983), adopted by governments at the Food and Agriculture Organisation (FAO) of the United Nations in 1981.¹⁴⁴ UNCLOS stipulates that all nations should share benefits from the seas. In Article 140, Paragraph 1, 13¹⁴⁵ UNCLOS acknowledges the vulnerability of LMICs in accessing benefits from the sea and urges HICs to be aware of this discrepancy.

The IUPGR seeks to promote the use of plant genetic material to benefit all humankind, including present and future generations.¹⁴⁶ The lack of implementation of this treaty in conjugation increasing cases of biopiracy and concern from developing countries regarding the exploitation and depletion of their indigenous genetic resources without just compensation, this undertaking has, in effect, been rendered defunct.¹⁴⁷

The International Treaty on Plant Genetic Resources for Food and Agriculture (ITPGRFA) of 2004 was the first legally binding international agreement focusing on the conservation and sustainable use of plant genetic resources for food and agriculture. The ITPGRFA seeks to promote the conservation of biodiversity while controlling private or sovereign control of plant genetic resources, which is inappropriate for food and agriculture.

¹⁴²Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170. ¹⁴³UN convention on Law of the Sea

https://www.imo.org/en/ourwork/legal/pages/unitednationsconventiononthelawofthesea.aspx#:~:tex t=The%20United%20Nations%20Convention%20on,the%20oceans%20and%20their%20resource s. (Date of use: 23 May 2023).

¹⁴⁴FAO: *International Undertaking on Plant and Genetic Resources*. Rome: Electronic publishing 1983:10. ¹⁴⁵UN Convention on Law of the Sea.

https://www.imo.org/en/ourwork/legal/pages/unitednationsconventiononthelawofthesea.aspx#:~:tex t=The%20United%20Nations%20Convention%20on,the%20oceans%20and%20their%20resource s. (Date of use: 23 May 2023).

¹⁴⁶FAO: *International Undertaking on Plant and Genetic Resources*. Rome: Electronic publishing 1983:10.

¹⁴⁷De Jonge B and Korthals M "Vicissitudes of benefit sharing of crop genetic resources: downstream and upstream" 2006 *Dev World Bioeth* 6: 144–157.

¹⁴⁸Gerstetter C, Gorlach B, Neumann K and Schaffrin D "The international treaty on plant genetic resources for food and agriculture within the current legal regime complex on plant genetic resources" 2007 *The Journal of World Intellectual Property* 10(3/4): 259–283.

¹⁴⁹Halewood M and Nnadozie K "Giving priority to the commons: the international treaty on plant genetic resources for food and agriculture" in Tansey G and RajotteT (eds) The future control of food (London, Earthscan 2008).

The term benefit sharing was popularised at the Convention on Biological Diversity (CBD) and adopted at the 1992 Earth Summit in Rio de Janeiro, Brazil. This was the first treaty which recognised that although the conservation of biodiversity is a common concern for all humankind, the genetic resources of a nation are sovereign. The parties to this treaty all adopted the 2002 Bonn Guidelines on Access to Genetic Resources and Fair and Equitable Sharing of the Benefits Arising out of their Utilization (Bonn Guidelines). These guidelines were to provide signatories with a "transparent framework to facilitate access to genetic resources and ensure fair and equitable sharing of benefits". In a supplementary agreement to the CBD, through the Nagoya Protocol, benefit sharing has been established as a principle of international law in the context of non-human genetic resources. Signatories to the CBD are legally obliged to share the benefits arising from access to biological resources. The resources covered by the CBD include animal, plant, microorganisms and traditional knowledge.

In the context of the principle of justice, the benefit sharing arrangements in the CBD and the Bonn guidelines rely on a mutually beneficial exchange, dealing with *justice in exchange*, be it of a monetary or non-monetary nature.¹⁵⁶ This entails that the provider of resource and user of the resource should benefit in some way from the research project.

At this juncture, it is necessary to briefly refer to the Convention for the Protection of Human Rights and Dignity of the Human Being with regard to the Application of Biology and Medicine (known as the Convention of Human Rights and Biomedicine or the Oviedo Convention).¹⁵⁷ This Convention of the Council Europe was open for signature in Oviado

¹⁵⁰The Convention on Biological Diversity 1992 https://www.un.org/en/observances/biological-diversity-day/convention#:~:text=The%20Convention%20on%20Biological%20Diversity,been%20ratified%2 0by%20196%20nations (Date of use: 23 May 2023).

¹⁵¹Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioethics* 6(3):135–143.

¹⁵²The Convention on Biological Diversity 2002. https://www.cbd.int/doc/publications/cbd-bonn-gdls-en.pdf (Date of use: 23 may 2023).

¹⁵³Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioethics* 6(3): 135–143.

¹⁵⁴Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 03 June 2020).

¹⁵⁵Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

¹⁵⁶Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioethics* 6(3):135–143.

¹⁵⁷Council of Europe Convention of Oviedo 1997 found at https://rm.coe.int/168007cf98 (Date of use:10 October 2023).

by the member States, the non-member States which have participated in its elaboration and by the European Union, and for accession by other non-member States in 1997.¹⁵⁸ The Convention is the first legally-binding international text designed to preserve human dignity, rights and freedoms, through a series of principles and prohibitions against the misuse of biological and medical advances.¹⁵⁹ Article 21 of the Convention asserts that "[t]he human body and its parts shall not, as such, give rise to financial gain."¹⁶⁰ This Convention, although regarded as one of the most authoritative reference documents on subjects of bioethics, ¹⁶¹ makes no reference to benefit sharing in health research.

2.2.3 Benefit sharing relating to *human biological resources*

The concept of benefit sharing in human health research is less established.¹⁶² Health research participation generally relies on altruism and strives to avoid undue inducement, which, in turn, could lead to the exploitation of vulnerable participants in LMICs.¹⁶³ Another reason for caution may be attributed to the concept of ownership of human biological resources.¹⁶⁴ The human body and its parts are classified as *res extra commercium* (that which cannot be commodified).¹⁶⁵ This classification has led to challenges in human tissue health research and the regulation of tissue banks and biobanks.¹⁶⁶

In addition to the arguments advanced in chapter one above, and also considering the number of recorded exploitation cases in health research globally as exemplified by the avian flu virus samples case in Indonesia, 167 it becomes clear that a legal framework

¹⁵⁸ https://www.coe.int/en/web/conventions/full-list?module=treaty-detail&treatynum=164 (Date of use: 10 October 2023)

¹⁵⁹https://www.coe.int/en/web/conventions/full-list?module=treaty-detail&treatynum=164 (Date of use: 10 October 2023).

¹⁶⁰https://rm.coe.int/168007cf98 (Date of use :10 October 2023).

¹⁶¹Petrini C, and Ricciardi W "The Convention on Human Rights and Biomedicine twenty years later: a look at the past and a step towards the futur" 2018 *Annali dell'Istituto Superiore di Sanitā* 1(54)(3):17.

¹⁶²Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

¹⁶³Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

¹⁶⁴Mahomed S, Nahrens K, Slabbert M and Sanne I "Managing human tissue transfer across national boundaries—an approach from an institution in South Africa" 2016 *Dev World Bioeth* 16:39–35.

¹⁶⁵Mahomed S 2018 *An ethico-legal framework for the regulation of biobanks in South Africa* (Bioethics and Health Law PhD 2018 University of the Witwatersrand) Available at: https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 23 May 2023).

¹⁶⁶Labuschagne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediments to accessing health care" 2019 *SAJBL* 12(1): 27–31.

¹⁶⁷Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook

governing benefit sharing for human biological material must be established. Avian flu (H5N1 influenza type A) became a global pandemic in 2005, with most death casualties occurring in Indonesia.¹⁶⁸

In the event of a world pandemic, the World Health Organisation (WHO) collects virus samples to distribute to affiliated laboratories for vaccine production purposes. Between 2005 to 2006, the Avian flu re-emerged in Indonesia, whereupon the country provided the WHO with a large number of virus specimens which were distributed to various US laboratories, as well as to Hong Kong University. 169 It emerged in 2006 that the WHO had shared the Indonesian virus samples with independent laboratories and that non-Indonesian researchers were presenting research using Indonesian samples without permission from the Indonesian government. Indonesia hence decided to withhold their virus samples from the WHO¹⁷⁰ which led to a rift developing between the WHO and Indonesia. This situation further deepened when reports confirmed that members of the WHO Global Influenza Surveillance Network (GISN) had shared information with private firms and its own member institutions (all situated in HICs), with the latter filing patent applications using the Indonesia virus samples. 171 It is patently unfair and inequitable for private firms, mostly from HICs, to use virus samples sourced from LMICs for the development of vaccines and subsequently to sell these therapies back to LMICs—often at inflated prices.

After new agreements were concluded by the WHO's Open-Ended Working group of Member States on Pandemic Influenza in 2011,¹⁷² granting LMICs guaranteed access to therapies or royalties originating from their donated virus samples, Indonesia resumed

Lucas J (eds) Benefit Sharing From Biodiversity to Human Genetics (Springer Science 2013) 95–127.

¹⁶⁸Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 95–127.

¹⁶⁹Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 95–127.

¹⁷⁰Sedyaningsih ER, Isfandari S, Soendoro T and Supari SF "Towards mutual trust, transparency and equity in virus sharing mechanism: the avian influenza case of Indonesia" 2008 *Ann Acad Med Singap* 37(6):482–488.

¹⁷¹Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–127.

¹⁷²Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–127.

sharing its virus samples with the WHO.¹⁷³ This case demonstrates the negative consequences when a trust relationship is violated, which, in turn, illustrates the need for pre-arranged benefit sharing agreements between the sample providers (in this instance, the Indonesian government and end users of samples). It further underscores the requirement for transparency throughout the research process.

The next section will discuss global instruments, recommendations and guidelines relating to benefit sharing arrangements when human biological materials (HBMs), including human DNA, RNA, blastomeres, polar bodies and human tissues, are used in health research.

2.2.3.1 UNESCO Universal Declaration on Bioethics and Human Rights

The United Nations Educational, Scientific and Cultural Organisation (UNESCO) was founded in November 1945. It is an organ of the United Nations which sets out to promote building a culture of peace, the eradication of poverty, sustainable development and intercultural exchange through education, the sciences, culture, communication and information.¹⁷⁴

One of the first international pronouncements on benefit sharing and the ethics of genetic research was the Declaration on the Human Genome and Human Rights, issued and adopted by UNESCO in 1997.¹⁷⁵ Unfortunately, as a soft law instrument, the Declaration is not legally binding on member states. South Africa is a signatory to this declaration.

The statements of this declaration appear to be aligned with the idea of sharing benefits based on common property and distributive justice, ¹⁷⁶ as may be concluded from Article 12(a), which states that "benefits from advances in biology, genetics and medicine, concerning the human genome, shall be available to all with due regard for the dignity and human rights for (sic) each individual." Article 19a(ii) further provides that:

In the framework of international cooperation with developing countries, States should seek to encourage measures enabling: countries to benefit from the

¹⁷³Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–127.

¹⁷⁴UNESCO Universal Declaration on Bioethics and Human Rights https://en.unesco.org/about-us/legal-affairs/universal-declaration-bioethics-and-human-rights#:~:text=and%20human%20rights-

^{,1.,}interest%20of%20science%20or%20society (Date of use: 23 May 2023).

¹⁷⁵UNESCO: UNESCO Declaration of Human Genome and Human Rights. Paris: 1997.

¹⁷⁶UNESCO: UNESCO Declaration of Human Genome and Human Rights. Paris: 1997.

achievements of scientific and technological research so that their use in favour of economic and social progress can benefit all.¹⁷⁷

These statements emphasise the idea that the human genome is the common heritage of mankind and must be available to all. 178

In 2005, the UNESCO General Conference adopted the Universal Declaration on Bioethics and Human Rights.¹⁷⁹ Article 2(f) of the Declaration provides that the Declaration aims to:

Promote equitable access to medical, scientific and technological developments as well as the greatest possible flow and the rapid sharing of knowledge concerning those developments and the sharing of benefits, with particular attention to the needs of developing countries.¹⁸⁰

Article 15(1) of the Declaration deals with benefit sharing and provides that:

Benefits resulting from any scientific research and its applications should be shared with society as a whole and within the international community, in particular with developing countries.¹⁸¹

Article 15 (1) furthermore identifies some of the forms that a benefit may take, which are: 182

- a) Special and sustainable assistance to, and acknowledgement of, the persons and groups that have taken part in the research;
- b) Access to quality health care;
- c) Provision of new diagnostic and therapeutic modalities or products stemming from research;
- d) Support for health services;
- e) Access to scientific and technological knowledge;
- f) Capacity-building facilities for research purposes;
- g) Other forms of benefit consistent with the principles set out in this Declaration.

Emphasising equal participation, Article 21(4) of the Declaration advises that:

¹⁷⁷UNESCO: UNESCO Declaration of Human Genome and Human Rights. Paris: 1997.

¹⁷⁸Resnik DB "The human genome common resource but not common heritage" 2005 *ResearchGate* found at:

https://www.researchgate.net/publication/255581513_The_human_genome_common_resource_b ut not common heritage (Date of use: 23 May 2023).

¹⁷⁹UNESCO Universal Declaration on Bioethics and Human Rights https://en.unesco.org/about-us/legal-affairs/universal-declaration-bioethics-and-human-rights#:~:text=and%20human%20rights-

^{,1.,}interest%20of%20science%20or%20society (Date of use: 23 may 2023).

¹⁸⁰Article 2(f) of UNESCOs' Universal Declaration on Bioethics and Human Rights.

¹⁸¹Article 15(1) of UNESCOs' Universal Declaration on Bioethics and Human Rights.

¹⁸²Article 15(1) of UNESCOs' Universal Declaration on Bioethics and Human Rights.

[W]hen negotiating a research agreement, terms for collaboration and agreement on the benefits of research should be established with equal participation by those party to the negotiation.¹⁸³

As stated above, benefit sharing need not be of a monetary nature. In LICs, the suggestions in Article 15(1) of what a benefit could entail seem reasonable and just. Capacity building, access to healthcare and infrastructure development are welcome benefits for LICs. In addition, participants involved in the research should be able to engage in discussions with researchers regarding benefit sharing arrangements either personally or via community representatives.

2.2.3.2 The Human Genome Organisation: Project Ethics Committee Statement

The mapping of the entire human genome brought about a proliferation of genomic and biobank research globally as well as in South Africa.¹⁸⁴ The Human Genome Organisation (HUGO) was founded in 1988 to coordinate the genomic research that was underway in different nations.¹⁸⁵

The HUGO Ethics Committee set out to suggest whether and how to distribute profits that may accrue to commercial enterprises, governments or academic institutions based on the participation of particular communities.¹⁸⁶ The Committee published the following six recommendations in their Statement on Benefit Sharing in 2000:¹⁸⁷

- 1) All humanity shares in and have (sic) access to the benefits of genetic research.
- 2) Benefits should not be limited to those individuals who participated in the research.
- 3) Prior discussions with groups or communities on the issue of benefit sharing should take place.
- 4) Even in the absence of profits, immediate health benefits as determined by community needs should be provided.
- 5) At a minimum, all research participants should receive information about general research outcomes and an indication of appreciation.

¹⁸³Article 21(4) of UNESCOs' Universal Declaration on Bioethics and Human Rights.

¹⁸⁴Guardasani D, Carstensen T, Tekola-Ayele FR *et al* "The African genome variation project shapes medical genetics in Africa" 2015 *Nature* 517:327–32.

¹⁸⁵The Human Genome Organisation http://www.hugo-international.org/history/ (Date of use: 23 May 2023).
¹⁸⁶HUGO Ethics Committee on benefit sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,the%20issue%20of%20benefit%2Dsharing (Date of use: 23 May 2023).

¹⁸⁷HUGO Ethics Committee: HUGO urges genetic benefit sharing 2000 Clinical Genetics 58:364-366.

6) Profit-making entities should dedicate a percentage (e.g., 1%–3%) of their annual net profit to healthcare infrastructure and/or to humanitarian efforts.

The HUGO Project Ethics Committee defines a benefit as:

A good that contributes to the well-being of an individual and/or a given community (e.g., by region, tribe, disease-group ...). Benefits transcend avoidance of harm (non-maleficence) insofar as they promote the welfare of an individual and/or of a community. Thus, a benefit is not identical with(sic) profit in the monetary or economic sense. Determining a benefit depends on needs, values, priorities and cultural expectations.¹⁸⁸

Although the HUGO Ethics Committee acknowledges in its statement that "expenditures by private industry for genetic research now exceeds the contributions of governments", 189 it still advises that the benefits of genetic research should be available to all. Benefit sharing should be discussed with the relevant participants in the research and gratitude should be extended to them. Benefits arising from the research should be in the in line with the principle of distributive justice for the good of all. This last statement in my view may find application in the context of genetic research undertaken in HICs where the principle of distributive justice can be implemented for the sharing the benefits of the products of health research.

The HUGO Statement on Benefit Sharing of 2000 calls for private companies involved in health research to observe the "moral obligations" they have in the context of health research. This is sadly not always the case. Private companies are profit-driven and are usually not interested in concluding a benefit sharing arrangement with research participants. The HUGO Statement is predicated on the premise that all humans share a common genetic heritage, an idea that is contradicted by the many gene patents that exist and are held by private entities. The question rightly arises as to how research participants in LMICs may benefit from these patented genes when they are used to create expensive therapies.

¹⁸⁸HUGO Ethics Committee https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec

ommends,the%20issue%20of%20benefit%2Dsharing (Date of use: 23 May 2023).

189HUGO Ethics Committee: HUGO urges genetic benefit sharing 2000 *Clinical Genetics* 58:364–366.

¹⁹⁰HUGO Ethics Committee: HUGO urges genetic benefit sharing 2000 *Clinical Genetics* 58:364–366.

¹⁹¹Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: Will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 *Developing Word Bioeth* 21:125-130. ¹⁹²Resnick DB "The human genome: common resource but not common heritage" 2005 *Frontis* 197-210.

The HUGO Statement advocates for benefit sharing a form of distributive justice, which advocates for the equal sharing of benefits and burdens in society. Regarding collaborative research between LMICs and HICs health researchers, this is not easily achievable as power imbalances are operative and negotiation powers are consequently limited.

2.3.3.3 The Declaration of Helsinki

The World Medical Association was established in 1947 with the intention to discuss and provide guidance on the problems of practising medicine across borders.¹⁹³

The Declaration of Helsinki¹⁹⁴ was first adopted by the World Medical Association (WMA) General Assembly in Helsinki. The declaration aims to provide ethical principles for medical research involving humans and human biological materials and data.¹⁹⁵ Since its adoption in 1964, the Declaration has undergone seven revisions, the latest of which was in 2013.¹⁹⁶

At the Tokyo 2004 WMA General Assembly, a note of clarification was added to Article 30 of the Declaration, which reads as follows:

The WMA hereby reaffirms its position that it is necessary during the study planning process to identify post-trial access by study participants to prophylactic, diagnostic and therapeutic procedures identified as beneficial in the study or access to other appropriate care.¹⁹⁷

¹⁹³The World Medical Association available at: https://www.wma.net/what-we-do/ (Date of use: 26 September 2023).

¹⁹⁴World Medical Association Declaration of Helsinki - Ethical Principles for Medical Research Involving Human Subjects. https://www.wma.net/policies-post/wma-declaration-of-helsinki-ethical-principles-for-medical-research-involving-human-subjects/ (Date of use: 26 September 2023).

¹⁹⁵World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 26 September 2023).

¹⁹⁶World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 26 September 2023).

¹⁹⁷World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 26 September 2023).

This clarification meant that other benefits, such as "appropriate care" could be afforded to participants, unlike the previous rigidity of post study access to successfully tested drugs.¹⁹⁸

The present 2013 Declaration directly addresses issues pertaining to benefit sharing. Article 17 and Article 20 of the Declaration relate to aspects of benefit sharing. Article 19¹⁹⁹ provides that:

Some groups and individuals are particularly vulnerable and may have an increased likelihood of being wronged or of incurring additional harm. All vulnerable groups and individuals should receive specifically considered protection.²⁰⁰

Article 20²⁰¹ cautions that medical research with a vulnerable group is justified only when the research cannot be conducted using a non-vulnerable group:

Medical research with a vulnerable group is only justified if the research is responsive to the health needs or priorities of this group and the research cannot be carried out in a non-vulnerable group. Moreover, the group should stand to benefit from the knowledge, practices or interventions that result from the research.²⁰²

Article 34²⁰³ of the 2013 version of the Declaration (with regard to post-trial access) states that:

In advance of a clinical trial, sponsors, researchers and host country governments should make provisions for post -trial access for all participants who still need an intervention dentified as beneficial in the trial.

43

¹⁹⁸Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing:from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

¹⁹⁹World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁰⁰World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁰¹World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁰²World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁰³World Medical Association (WMA) Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

2.2.3.4 Council for International Organisations and Medical Sciences (CIOMS)

The international ethical guidelines for biomedical research involving human subjects by the Council for International Organisations of Medical Sciences (CIOMS) was established jointly by the WHO and UNESCO in 1949.²⁰⁴ This council aims to promote and assist in international activities regarding research in biomedical sciences.²⁰⁵ The CIOMS International Ethical Guidelines for Biomedical Research Involving Human Subjects were published in 1993 and updated in 2016²⁰⁶ to include explanations of what constitutes benefits and who is responsible for providing them. The CIOMs call for sponsors and researchers of HICs to negotiate research priorities and benefits with LMIC hosts.²⁰⁷

Benefits and benefit sharing feature extensively in the 2016 CIOMS Guidelines. Guideline 1²⁰⁸ states that "scientific and social value cannot legitimate (sic) subjecting study participants or host communities to mistreatment, or injustice". It also explains that all stakeholders in health research have "a moral obligation that all research is carried out in ways that uphold human rights, and respect, protect, and is fair to research participants and communities."

Regarding research conducted in low-resource settings, Guideline 2²⁰⁹ states that as part of their obligations, sponsors and researchers should "make every effort together with governments and other stakeholders to make available to the population or community any intervention or product developed as soon as possible and help build research capacity and infrastructure". The Guideline also states that the sponsors and researchers are to "consult with and engage communities in making plans for any intervention or product developed available, including the responsibilities of all relevant stakeholders". Additionally, Guideline 2²¹⁰states that benefits other than those directly associated with

²⁰⁴CIOMS and WHO International Ethical Guidelines for Health-related Research Involving Humans https://cioms.ch/wp-content/uploads/2017/01/WEB-CIOMS-EthicalGuidelines.pdf. (Date of use: 23 May 2023).

²⁰⁵CIOMS International Ethical Guidelines for Biomedical Research Involving Human Subjects https://www.fhi360.org/sites/all/libraries/webpages/fhi-retc2/Resources/CIOMS02.pdf (Date of use: 23 May 2023).

²⁰⁶CIOMS International Ethical Guidelines for Biomedical Research Involving Human Subjects https://www.fhi360.org/sites/all/libraries/webpages/fhi-retc2/Resources/CIOMS02.pdf (Date of use: 23 May 2023).

²⁰⁷CIOMS International Ethical Guidelines for Biomedical Research Involving Human Subjects https://www.fhi360.org/sites/all/libraries/webpages/fhi-retc2/Resources/CIOMS02.pdf (Date of use: 23 May 2023).

²⁰⁸Guideline 1 of the CIOMS.

²⁰⁹Guideline 2 of the CIOMS.

²¹⁰Guideline 2 of the CIOMS.

study participation, such as capacity building and improving health infrastructure, may be made available in resource-poor settings but must be determined after negotiations with the communities or local population.

Guideline 3²¹¹ reiterates that the benefits and burdens of research should be equitably distributed among all stakeholders in the research project.

Guideline 7²¹² suggests that researchers, sponsors, health authorities and other stakeholders should engage communities in the early stages of a project and the dissemination of its results.

Guideline 8,²¹³ while acknowledging that it is the responsibility of national governments to have competent ethical boards to review the feasibility of health-related research in their countries, specifically refers to capacity building being the responsibility of researchers and sponsors who conduct health research in low-resource countries. Capacity building is explained to include but is not limited to:

- Research infrastructure building and strengthening research capacity;
- Strengthening research ethics review and oversight capacity in host communities [...];
- Developing technologies appropriate to healthcare and health-related research; educating research and health care personnel and making arrangements to avoid undue displacement of healthcare personnel;
- Engaging with the community from which research participants will be drawn [...] arranging for joint publication consistent with recognized authorship requirements and data sharing; and
- Preparing a benefit sharing agreement to distribute eventual economic gains from the research.

In reference to collaborative partnerships, Guideline 8²¹⁴ provides that "to overcome power differences to give collaborators equal negotiating strength, steps to promote inclusion, mutual learning and social justice" must be taken.

²¹¹Guideline 3 of the CIOMS.

²¹²Guideline 7 of the CIOMS.

²¹³Guideline 8 of the CIOMS.

²¹⁴Guideline 8 of the CIOMS.

Guideline 11²¹⁵ refers to biological materials and its related data. The guideline states that "if specimen[s] and data are stored outside the original setting, there should be plans to return all materials to its(*sic*) origins and share results, benefits and burdens with all the stakeholders involved in the research". Regarding governance of the reuse of samples, the guideline implores the custodians of samples to establish to whom any benefits should accrue.

Guideline 13²¹⁶ suggests that research participants should be appropriately reimbursed for participating in research and that such compensation could be monetary or non-monetary but should not be exorbitant, to prevent undue inducement.

The Guidelines emphasise the need for benefit sharing and for negotiations to occur on an equal level between HICs and LMICs, which is not always a possibility.

2.2.4 Benefit sharing in South Africa: non-human biological resources

As mentioned earlier, the concept of benefit sharing was coined at the Convention on Biological Diversity (CBD) and adopted in 1992 at the Rio de Janeiro Earth Summit in Brazil.²¹⁷ Consequently, benefit sharing for non-human biological materials was established as a principle of international law via a supplementary agreement to the CBD, namely the Nagoya protocol.²¹⁸ South Africa is a signatory to the Nagoya protocol, a coherent international legal framework regulating the access to and sharing of benefits arising from non-human genetic materials.²¹⁹

2.2.4.1 Biodiversity Act 10 of 2004

The South African Biodiversity Act 10 of 2004²²⁰ is the only binding piece of legislation that regulates benefit sharing agreements in terms of illustrating and defining what the

²¹⁵Guideline 11 of the CIOMS.

²¹⁶Guideline 13 of the CIOMS.

²¹⁷Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²¹⁸Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 23 May 2023).

²¹⁸The Convention on Biological Diversity 1992 http://www.cbd.int/abs/about (Date of use: 23 May 2023).

²¹⁹Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

²²⁰Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

agreement should contain.²²¹ Section 80(i)²²² in Chapter 6 of the Act defines the mandate of the Act as follows:

- a. To regulate bioprospecting involving indigenous biological resources;
- b. To regulate the export from the Republic of indigenous biological resources for the purpose of bio-prospecting or any other kind of research;
- c. To provide for a fair and equitable sharing by stakeholders in benefits arising from bioprospecting involving indigenous biological resources.²²³

In terms of Section 80(2)(b)(i) of the Biodiversity Act, genetic material of human origin is clearly excluded²²⁴ and the Act only applies to bioprospecting and research involving indigenous biological resources.

Section 81(1)(b)²²⁵ of the Act prohibits any bioprospecting in SA and the removal of any biological material from SA unless a permit has been granted. Further, for this permit to be granted, section 82 of the Act requires the completion of a material transfer agreement (MTA) between the applicant and stakeholder (which could be a person, organ of the state, community or indigenous community). The MTA should regulate the provision of or access to those biological resources and a benefit sharing agreement that provides for sharing by the stakeholder in any future benefits that may be derived from the relevant bioprospecting.²²⁶ The sitting minister responsible for national environmental management must approve of the benefit sharing arrangement. Chapter 6 of the Act also allows the issuing authority to mediate negotiations between the applicant and stakeholder fairly and equitably.

The concept of benefit sharing concerning non-human genetic resources has received considerable attention in SA, as it provides an example of how two of South Africa's first benefit sharing arrangements were negotiated between a local indigenous community and one of the world's largest publicly trading pharmaceutical companies, in the absence of an enabling domestic legal environment.²²⁷ The San *Hoodia* case led to two benefit sharing arrangements for the San community, the first with a pharmaceutical company and the

²²¹Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

²²²Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

²²³S 80(1).

²²⁴S 80 (2).

²²⁵S 81.

²²⁶S 82(2).

²²⁷Mohamed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8:60-64.

second with *Hoodia* growers of Southern Africa.²²⁸ The San people are among the oldest communities in Southern Africa. They have considerable traditional knowledge of the *Hoodia gordonii*, a moist plant which is indigenous to the Kalahari Desert that has been used by the San people to curb hunger for centuries.²²⁹

The Council for Scientific and Industrial Research (CSIR), established by an act of Parliament, is one of the largest research institutes in Africa. After secretly bioprospecting the *Hoodia* plant upon learning of its beneficial traditional uses, the CSIR lodged a patent in 1995 in South Africa (patent number 983170) regarding the active appetite suppressant ingredients in the *Hoodia*.²³⁰ This was followed by international patent applications. In 1996, the CSIR patented a utility patent. No domestic access and benefit sharing laws or frameworks existed at the time.²³¹ Without establishing any agreements with the San community from whom the traditional knowledge derived, the CSIR further negotiated an exclusive license with Phytopharm, a company specialising in the development of phytomedicines, to transfer its research rights and commercial use of the patent for the development of Hoodia products for profit-making purposes.²³² Phytopharm, in turn, granted licenses to Pfizer and Unilever.²³³ This also occurred without the consent of the San community, who were excluded from all of these negotiations. Furthermore, there were no negotiations surrounding the sharing of benefits resulting from the commercialisation of *Hoodia*.²³⁴

The San people had suffered centuries of marginalisation.²³⁵ To advocate for their rights and access to land, the San leaders formed the Working Group for Indigenous Minorities

²²

²²⁸Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²²⁹The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

²³⁰Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²³¹Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²³²The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

²³³The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

²³⁴The San people and the Hoodia plant available at https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

²³⁵Suzman J "An introduction to the regional assessment of the status of the San in Southern Africa" (Windhoek: Legal Assistance Centre 2001).

in Southern Africa (WIMSA).²³⁶ In 2003, WIMSA, with the help of NGOs and the CSIR, negotiated the first *Hoodia* benefit sharing agreement.²³⁷ The San were to receive six per cent of the CSIR's royalties from licenses and eight per cent of the milestone payments.²³⁸ This benefit sharing agreement received a lot of criticism and the San people never received the windfall they had expected.²³⁹ Although far from perfect, the benefit sharing arrangement supplies a platform for future benefit arrangements that will enable communities to reap the benefits of their traditional knowledge and to share in the commercialisation of products based on such knowledge.²⁴⁰ The San people have since negotiated subsequent benefit sharing agreements with commercial *Hoodia* farmers and others, covering the uses of different plants.²⁴¹

It is submitted that the Biodiversity Act's benefit sharing provisions, albeit excluding human biological resources, could provide some direction in the development of a template for a legally binding benefit sharing arrangement for research involving human biological resources.

2.2.5 Benefit sharing in South Africa: human biological resources

Health research in South Africa is governed by the National Health Act 61 of 2003.²⁴² Although benefit sharing is an established principle of international law and adopted in SA through the Biodiversity Act, there exists no legally binding law in research that speaks directly to benefit sharing when using human biological materials.²⁴³ A discussion of the

²³⁶Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²³⁷The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

²³⁸Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit Sharing From Biodiversity to Human Genetics* (Springer Science 2013) 33–64.

²³⁹Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²⁴⁰The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 may 2023).

²⁴¹Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

²⁴²61 0f 2003, Chapter 9.

²⁴³Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

relevant guidelines and documents that currently guide research that involves human biological resources in South Africa follows below.

2.2.5.1 Department of Health Ethics in Health Research Guidelines, 2015

Section 69(1) of the National Health Act mandates the NHA to establish the National Health Research Ethics Council (NHREC), which is tasked with the responsibility of determining guidelines for the functioning of health research ethics committees.²⁴⁴

In terms of the NHA, the NHREC must: 245

- 1) set norms and standards for health research involving humans and animals, as well as for conducting clinical trials;
- 2) determine guidelines to facilitate best practices for research ethics committees;
- 3) register and audit research ethics committees;
- 4) adjudicate complaints about research ethics and Animal Research Ethics committees;
- 5) refer matters concerning violations of ethical or professional rules to the relevant health professions council; recommend disciplinary action against persons found to have violated the norms and standards set for the responsible and ethical conduct of health research; and
- 6) advise the national and provincial departments of health on ethical matters concerning research.

As stated already, the 2014 Regulations relating to Research with Human Participants, as well as paragraph 1.8.1²⁴⁶ of the Department of Health's Ethical Guidelines, stipulate that the Department's Guidelines are the minimum national benchmark of norms and standards for conducting responsible and ethical research, including human research, making them legally binding.

The Guidelines set out ethical principles that should govern all health research involving human participants as well as HBMs and data collected from living or deceased persons.

²⁴⁵Department of Health (DoH) Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

²⁴⁴Section 72(6) (a) of the NHA.

²⁴⁶DoH Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

The concept of benefit sharing is not directly addressed in the Guidelines. The Guidelines require that a fair balance of risks and benefits must apply to all who participate in research²⁴⁷ and state that "the population from which the participants are drawn will benefit from the research results if not immediately, then in the future".²⁴⁸ Furthermore, the Guidelines advise that the benefit to the participant or community in the research should outweigh the risk of harm.²⁴⁹In addition, the Guidelines outline that in obtaining informed consent, research participants should be informed of any potential benefits of their participation, both during and after the research.²⁵⁰ However, there is no reference in the Guidelines on how a benefit sharing arrangement should be formulated.

Despite suggesting that benefit sharing is necessary for health research, the Guidelines do not comment on how benefit sharing arrangements should be structured and how these arrangements should be negotiated. There is a clear need for providing guidance on how to structure benefit sharing arrangements that are agreeable to both researchers and research participants. Since the guidelines have some legal force, it may be deduced that the Guideline's reference to benefit sharing points to an ethico-legal imperative in the context of health research.

2.2.5.2 Health Professions Council of South Africa (HPCSA) Guidelines

In 2016, the HPCSA published revised Guidelines in sixteen booklets to guide health practitioners and health researchers in their mandate as health professionals.²⁵¹ The Guidelines were to "offer more precise guidance and direction for action in concrete situations" and to enable the HPCSA to implement sanctions against transgressors.²⁵²

²⁴⁷DoH Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

²⁴⁸DoH Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

²⁴⁹DoH Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

²⁵⁰DoH Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

²⁵¹HPCSA mandate https://headroom.co.za/wp-content/uploads/2020/03/HPCSA-Ethics_Booklet-03_2020-1.pdf (Date of use: 26 May 2023).

²⁵²HPCSA mandate https://headroom.co.za/wp-content/uploads/2020/03/HPCSA-Ethics_Booklet-03_2020-1.pdf (Date of use: 26 May 2023).

Booklet 13 specifically deals with ethical guidelines for health researchers. The guidelines recognise that biomedical research has advanced very swiftly in the last century and caution that with South Arica's Apartheid past, which led to the marginalisation of specific racial groups, the misuse of power in health research cannot be ignored.²⁵³

Section 6 of the guidelines relating to the duties of research participants cautions against the undue inducement of research participants and remarks that research participants should be fairly compensated for their time. Such compensation is to be specified in the relevant research protocol or proposal.²⁵⁴

The Guidelines also provide that the balance of burdens and benefits should be equalised within different population groups. Guideline 6.6.3 proposes that at the end of a study, the research participants are entitled to benefit from the study by accessing the best proven prophylactic, diagnostic and therapeutic methods identified by the study.²⁵⁵

Despite alluding to the need for benefit sharing, the guidelines do not suggest how this process should occur.

2.2.5.3 National Material Transfer Agreement for Human Biological Material

Although the NHA Act was proclaimed in 2003, certain regulations in Chapter 8 of the NHA Act dealing with Human Biological Materials (for example, blood, blood products, tissues and gametes) were only enacted in 2012.²⁵⁶ The National Department of Health only gazetted the national template of a Material Transfer Agreement of Human Biological Materials (HBMs) under the NHA in 2018.²⁵⁷

The MTA in section $2(2)^{258}$ defines a benefit as:

²⁵³HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s_for_Health_Researchers.pdf (Date of use: 26 May 2023).

²⁵⁴HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s for Health Researchers.pdf (Date of use: 26 May 2023).

²⁵⁵HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s_for_Health_Researchers.pdf (Date of use: 26 May 2023).

²⁵⁶Proclamation No 11 *Government Gazette* 35081 of 27 February 2012.

²⁵⁷Proclamation No 719 Government Gazette 41781 of 20 July 2018.

²⁵⁸Proclamation No 719 Government Gazette 41781 of 20 July 2018.

Amongst others, the sharing of information; use of research results; royalties; acknowledgement of the Provider as the source of the Materials; publication rights; transfer of technology or Materials; and capacity building.²⁵⁹

The MTA further defines benefit sharing in section 2(3) as "the process or act of sharing in the benefits that derive from the Project in a manner that is fair and equitable". ²⁶⁰ Section 7 of the MTA expressly provides that "the sharing of benefits should be discussed and negotiated between the Provider and Recipient before Materials are transferred to the Recipient". ²⁶¹

The rapid commercialisation of health research has precipitated increased traffic of HBMs across national boundaries. In South Africa, universities and research institutions are increasingly experiencing financial hardships, with the result that there is a dire need to procure funds from sources other than government funds. This financial burden may often lead to unequal partnerships in research. Evidence of the exploitation of research participants exists, as well as documented proof that HBMs have left South Africa without ethical clearance, export permits and MTAs during collaborative research. This is why the gazetted MTA is of particular importance for collaborative health research in the current environment. If collaborative research is to equally benefit all involved, a need for a legally binding contract that protects all participants involved in the study is warranted. The gazetted MTA attempts to set out "a national, uniform template within which the Parties in research collaboration will engage in the transfer, use and other processing of the Materials". At the time of writing this thesis, some shortcomings in the gazetted MTA have been identified, as well as the need to revise the template. This is discussed in more detail elsewhere in this thesis.

The MTA specifically refers to benefit sharing in section 7.²⁶⁵ In principle, the MTA recognises the need for a benefit sharing agreement between research collaborators at the start of the research. However, the MTA does not provide a template of a benefit sharing agreement, which is one of the aims of this thesis, i.e., to propose a template that

²⁵⁹Section 2 (2) of the National MTA.

²⁶⁰Proclamation No 719 Government Gazette 41781 of 20 July 2018.

²⁶¹Proclamation No 719 Government Gazette 41781 of 20 July 2018.

²⁶²Labuschaigne M, Dhai A, Mahomed S *et al* "Protecting participants in health research: the South African Material Transfer Agreement" 2019 *S Afr Med J* 109(5):353–356.

²⁶³Labuschaigne L, Dhai A, Mahomed S *et al* "Protecting participants in health research: the South African Material Transfer Agreement" 2019 *S Afr Med J* 109(5):353–356.

²⁶⁴Labuschaigne L, Dhai A, Mahomed S *et al.* "Protecting participants in health research: the South African Material Transfer Agreement" 2019 *S Afr Med J* 109(5):353–356.

²⁶⁵Proclamation No 719 Government Gazette 41781 of 20 July 2018.

can be routinely used to guide on benefit sharing arrangements for health research in South Africa.

2.3 The ethics of benefit sharing

The notion of justice as an ethical principle has existed for centuries in many communities, ²⁶⁶ whereas the concept of benefit sharing, which serves to implement justice, only emerged towards the end of the 20th century. ²⁶⁷ As discussed already, the notion of benefit sharing has been accepted in many national and international ethics guidelines, not to mention its inclusion in the CBD, the Declaration of Helsinki (WMA 2008) and in UNESCO's Universal Declaration on Bioethics and Human Rights (UNESCO 2005). ²⁶⁸

Since it has been recognised that scientific advancement without benefit sharing is unjust, it follows that those who contribute to the advancement of science need to benefit in return.²⁶⁹ These are the foundational principles informing the philosophical principle of benefit sharing; however, just because the idea of benefit sharing exists, does not mean it is actually realised. If benefit sharing does not occur, it is warranted to assert that an injustice has occurred, possibly resulting in the exploitation of another party.²⁷⁰

The argument against benefit sharing states that it may lead to the undue inducement of participants in a study and that the more vulnerable a population, the more probable that this undue inducement would result in the exploitation of the population.²⁷¹ An argument that benefit sharing should legally be constrained by legal prohibition on the commercialisation of human tissue, is legally and ethically unsound, as benefit sharing

²⁶⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue Inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

²⁶⁶Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 217–230.

²⁶⁸Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 217–230.

²⁶⁹Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook J Lucas (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 217–230.

²⁷⁰Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 217–230.

²⁷¹Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 217–230.

cannot be equated to the sale of human biological material. Research participants donating biological material do so altruistically, whilst their participation in the research is the contribution that benefit sharing aims to redress.

It is impossible to consider the ethics of benefit sharing without detailing its relationship with the ethical concepts of exploitation, vulnerability and undue inducement,²⁷² which is addressed in the next section.

2.3.1 Vulnerability

The protection of vulnerable persons is central to research ethics; however, how to define vulnerable persons or populations is not very clear.²⁷³ Vulnerability is an important concept in bioethics because a vulnerable person or population is more prone to exploitation, which is morally wrong.²⁷⁴ Problems arise when considering health research conducted in LMICs by sponsors or institutions from HICs causing the necessity to determine what constitutes exploitation as well as when or if harm has occurred.²⁷⁵ All humans are vulnerable and can experience intrinsic harm (due to mental illness or old age) or extrinsic harm (due to external circumstances).²⁷⁶

The Oxford English dictionary describes vulnerability as exposure to the risk of being attacked or harmed, either physically or emotionally.²⁷⁷ The CIOMS Guidelines of 2002 refer to vulnerability as:

A substantial incapacity to protect one's own interest owing to such impediments as lack of capability to give informed consent, lack of alternative means of obtaining medical care or other expensive necessities, or being a junior or subordinate member of a hierarchical group.²⁷⁸

²⁷²Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 217–230.

²⁷³Hurst S "Vulnerability in research and healthcare; describing the elephant in the room?" 2008 *Bioethics* 22:191–202.

²⁷⁴Macklin R "Bioethics, vulnerability, and protection" 2003 *Bioethics* 17(5-6):472–486.

²⁷⁵Macklin R "Bioethics, vulnerability, and protection" 2003 *Bioethics* 17(5-6):472–486.

²⁷⁶Dhai A "Vulnerability exploited and a population betrayed" 2012 SAJBL 5(2): 62–63.

²⁷⁷Oxford Dictionary 12th Ed Oxford University Press.

²⁷⁸CIOMS Guidelines https://cioms.ch/wp-content/uploads/2016/08/International_Ethical_Guidelines_for_Biomedical_Research_Involving_H uman_Subjects.pdf (Date of use: 26 May 2023).

This definition is broad. Guideline 13 of CIOMS explains that vulnerable persons cannot protect their own interests because of a lack of power, intelligence, education, resources and strength.²⁷⁹

Current ethical guidelines, such as the amendment to the Declaration of Helsinki of 2013. refer to several groups in the definition of vulnerable persons.²⁸⁰ In article 19, the Declaration provides that "some groups and individuals are particularly vulnerable and may have an increased likelihood of being wronged or incurring additional harm. All vulnerable groups should receive specially considered protection". 281 It would not be wrong to assume from this definition that all persons in LMICs are vulnerable persons when exposed to specific circumstances, yet this is not necessarily true. When defining vulnerability, the Declaration focuses on the ability of an individual or persons to give informed consent, for the protection of the autonomy of a person or society which seems to link vulnerability to medical vulnerability and legal age of consent.²⁸² The Declaration seeks to fulfil the notion of distributive justice by recognising that the vulnerable needs to be protected and should stand to benefit from medical research.²⁸³ The Declaration however, is silent on contextual vulnerability, for example, research done in LMICs, conflict areas or refugee camps where there is a broad range of vulnerable populations and where benefit sharing should be a pre-requisite for health research. The Declaration furthermore lacks clarity on a description of vulnerable persons or populations during disaster situations like floods, earthquakes and pandemics. Evans and others²⁸⁴ correctly argue that there is no adequate direction for responders, researchers and organisations to guide benefit sharing during infectious disease outbreaks. During the recent COVID-19 pandemic, the world has witnessed (despite calls for solidarity and social justice) unfair access to vaccines to LMICs, with HICs

-

²⁷⁹CIOMS Guidelines https://cioms.ch/wp-content/uploads/2016/08/International_Ethical_Guidelines_for_Biomedical_Research_Involving_H uman_Subjects.pdf (Date of use: 26 May 2023).

²⁸⁰WMA 2013 https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁸¹WMA 2013 https://www.wma.net/wp-content/uploads/2016/11/DoH-Oct2013-JAMA.pdf (Date of use: 23 September 2023).

²⁸²Article 25 of the WMA Declaration of Helsinki 2013

²⁸³Article 20 of the WMA Declaration of Helsinki 2013.

²⁸⁴Evans NG, Hills K and Levine AC "How should the WHO Guide Acess and Benefit Sharing During Infectious Disease Outbreaks?" 2020 *AMA Journal of Ethics* 22(1):28-35.

unconscionably nationalising and stock-piling limited COVID-19 vaccines, and this despite LMICs having participated in research that lead to the development vaccines.²⁸⁵

Schroeder and Gefanas²⁸⁶ argue that the definition of vulnerability should merge the traditional dictionary definition, which focuses on extrinsic factors (like harm from outside), with the CIOMS guidelines' definition, which focuses on intrinsic factors (e.g., the inability to protect oneself). The authors also suggest that special protection cannot be offered to every person, only to persons who are at risk of exploitation during medical research. In this regard, they offer another definition of vulnerability in medical research. They propose that these definitions, in combination, may be paraphrased as follows: "To be vulnerable means to face a significant probability of incurring an identifiable harm, while substantially lacking the ability or means to protect oneself". ²⁸⁷

Levine *et al*²⁸⁸ suggest that in health research collaborations in LMICs, the principal concerns surrounding the issue of vulnerability revolve around inequalities of power and resources. It should be acknowledged that even within one country or setting, not all persons share the same level of vulnerability. According to a World Bank report,²⁸⁹ poverty in South Africa in 2022 was estimated at 62.6%, based on the upper-middle-income country poverty line.²⁹⁰ In the previous World Bank survey conducted 2014/2015, approximately 55% of the population were reported to be living below the national upper-bound poverty line.²⁹¹ Despite the advent of democracy in South Africa, most of the previously marginalised citizens remain exposed to abject poverty,²⁹² making the inequality

²⁸⁵ Moodley K, Blockman M, Hawkridge AJ et al., "Hard choices: Ethical challenges in phase 1 of COVID-19 vaccine roll-out in South Africa" 2021 SAMJ 111(6):554-558.

²⁸⁶Schroeder D and Gefanas E "Vulnerability: too vague and too broad?" 2009 *Cambridge Quarterly of Healthcare Ethics* 18:113–121.

²⁸⁷Schroeder D and Gefanas E "Vulnerability: too vague and too broad?" 2009 *Cambridge Quarterly of Healthcare Ethics* 18:113–121.

²⁸⁸Levine C, Faden R, Grady C *et al* "The limitations of "vulnerability" as protection for human research participants" 2004 *AJOB* 4(3):44–49.

²⁸⁹ https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpo-zaf.pdf (Date of use: 23 september 2023).

 $^{^{290}\}mbox{https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpozaf.pdf (Date of use: 23 september 2023).$

²⁹¹https://databankfiles.worldbank.org/public/ddpext_download/poverty/987B9C90-CB9F-4D93-AE8C-750588BF00QA/current/Global_POVEQ_ZAF.pdf (Date of use: 23 September 2023).

²⁹²Dhai A "Vulnerability exploited and a population betrayed" 2012 SAJBL 5(2):62-63.

among the highest in the wolrld.²⁹³ The government has not been able to eradicate distributive inequality which has led to distinct inequalities in the income, education and health of its citizens.²⁹⁴ These inequalities, coupled with ongoing electricity supply shortages, socio-economic disparities and the COVID-19 pandemic,²⁹⁵ expose many South Africans to specific vulnerabilities because they may lack education, adequate access to health care resources and financial freedom.

Considering that the level of vulnerability may vary for every person and that different kinds of health research may involve different levels of risk, identifying vulnerable persons should be based on what harm could occur to such persons during the specific type of health research. There are many parameters of possible harm to research participants in studies; these range from but are not limited to, an unfavourable risk—benefit ratio, breach of privacy, invalid consent and the lack of access to the benefits of health research.

Health research involving genetic material usually only requires a swab to obtain DNA and is of minimal harm to research participants from a physical or health-related view. Harm may occur on many other levels, for example if the research participants did not enjoy access to health care services or access to the benefits that followed from their participation in research. It is in this regard that benefit sharing agreements should be mandatory for research collaborations to address the issue of justice, which requires that all who participate in research should benefit from that research.

2.3.2 Exploitation

According to the Oxford English Dictionary, the ordinary meaning of exploitation is to make use of and unfairly and unjustly benefit from the work of another.²⁹⁶ Anarson and Schroeder propose another meaning, specifically relating to health research in LMICs, by describing wrongful exploitation as "a failure to benefit others as some norm of fairness requires".²⁹⁷

²⁹³https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpozaf.pdf (last accessed 23 september 2023).

²⁹⁴https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpo-zaf.pdf (last accessed 23 september 2023).

 $^{^{295}\}underline{\text{h}}$ ttps://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpozaf.pdf (last accessed 23 september 2023).

²⁹⁶Oxford Dictionary Twelfth Edition Oxford University Press.

²⁹⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

When considering human health research, the word *exploitation* is often used freely when referring to an injustice, a seeming lack of fairness or a wrong inferred on a person, community or population.²⁹⁸ Exploitation is supposedly immoral and wrong, but sometimes its wrongfulness cannot easily be explained or described.²⁹⁹

Where collaborative research occurs between HICs and LMICs, the term becomes more prominent because of the assumed vulnerabilities of research participants in LMICs, yet exploitation of the vulnerabilities of research participants is not always deemed wrong. Schroeder and Anarson³⁰⁰ attempt to explain this by referring to the following hypothetical drug trial: If the only way for an HIV-positive pregnant woman to obtain antiretrovirals to halt HIV transmission to her unborn baby is to participate in a placebo-controlled trial (where some participants are given the active drug and the control group are given an inactive pill); then, despite access to healthcare, it could be said that she has indirectly been forced or compelled to enrol in the study. Although seemingly unfair to the research participants in the placebo arm of the research, it may be said that no actual harm has been done to them.

There is no consensus on what makes an instance of exploitation wrong, even when there is agreement that it has indeed occurred.³⁰¹ Schwartz³⁰² suggests that exploitation is wrong because it is coercive, as with the hypothetical HIV drug trial. It is an injustice because the research is only used to benefit others. Some claim that it is wrong because it degrades the participants treating them as a means to an end,³⁰³ while others claim that it is wrong because the vulnerable should be protected.³⁰⁴

In 2007, Mayer suggested that fundamental wrongdoing could be found with exploitation if it is grouped into a family of inequities to which it partially belongs.³⁰⁵ Exploitation is one type of wrongful gain because it has something in common with robbery and theft.³⁰⁶

²⁹⁸Chennels R "Equitable access to human biological resources in developing countries: benefit sharing without undue inducement" (PhD Thesis School of Health University of Lancaster 2014).

²⁹⁹Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³⁰⁰Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9–32.

³⁰¹Mayer R "What's wrong with exploitation?" 2007 Journal of Applied Philosophy 24(2):137–150.

³⁰²Schwartz J "What's wrong with exploitation?" 1995 Nous 29(2):158–188.

³⁰³Wood A "Exploitation" 1995 Social Philosophy and Policy 12(2):136–158.

³⁰⁴Macklin R "Bioethics, vulnerability and protection" 2003 *Bioethics* 17(5-6):472–486.

³⁰⁵Mayer R "What's wrong with exploitation?" 2007 *Journal of Applied Philosophy* 24(2):137–150.

³⁰⁶Mayer R "What's wrong with exploitation?" 2007 Journal of Applied Philosophy 24(2):137–150.

However, exploiters are not criminals like thieves—they gain wrongfully but not necessarily illegally.307

Mayer classifies three groups of exploitation of wrongful gain based on fairness. Cass 1 exploiters do not benefit the disadvantaged party at all, even if they ought to. They are 'freeloaders' who benefit from the input of others while contributing nothing. This is exemplified by persons who are capable of paying taxes but avoid doing so, yet still want to access roads and hospitals funded by other taxpayers' money. This class of exploitation involves no exchange of any kind and is thus irrelevant to this discussion. Class 2 exploiters typically never benefit others sufficiently; they only provide the bare minimum. In this form of exploitation, an exchange occurs but not to the fair benefit of both parties. This form of exploitation involves reciprocal exchanges where fairness is questioned. 308 It is also the most common form of exploitation that typically occurs in research collaborations between HICs and their institutions, and LMICs and their institutions. This type of classification of exploitation is depicted by the genetic harvest project example below. With increased health research occurring in LMICs like SA where communities were historically marginalised, 309 there are justified concerns about the exploitation of vulnerable communities. This type of exploitation is illustrated by an international collaborative research project that was conducted with villagers in Anhui, a poor province in China from 1995 to 2000,310 by Harvard University, the Chinese government and local Chinese research institutes.311

In 2002 it was reported that a Chinese American scientist, jointly funded by the National Institutes of Health (NIH) of America and biopharmaceutical companies had been collecting blood samples from the Anhui villagers over a period of five years.³¹² This occurred without the prior informed consent of the villagers, yet their blood samples were

Schroeder D, Cook Lucas J et al (eds) Ethics dumping, case studies from North-South research collaborations (Springer International Publishing Cham 2018) 71-80.

³⁰⁷Chennels R "Equitable access to human biological resources in developing countries: benefit sharing without undue inducement" (PhD Thesis School of Health University of Lancaster 2014).

³⁰⁸Chennels RS "Equitable access to human biological resources in developing countries: benefit sharing without undue inducement (PhD Thesis School of Health University of Lancaster 2014).

³⁰⁹Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440. ³¹⁰Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in

³¹¹Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in Schroeder D. Cook Lucas J et al (eds) Ethics dumping, case studies from North-South research collaborations (Springer International Publishing Cham 2018) 71-80.

³¹²Pomfret Nelson D "An isolated region's genetic mother lode" 2000-12-20 https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-motherlode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

transferred to a US biobank for research into several diseases.³¹³ No research ethics committee had also sanctioned this genetic harvesting. The study had 16 686 participants instead of 2 000, as stipulated in the approved protocol for the asthma study; the monetary compensation promised to the farmers was significantly reduced and the bronchodilators used were different from those approved in the protocol.³¹⁴ Class 2 exploitation is undeniably portrayed in this scenario since an unfair exchange had occurred between the researchers and participants. The injustice exists in that Harvard University benefitted significantly from the monies received from the NIH and the biopharmaceutical companies, whereas the villagers received very little benefit.

Finally, in Class 3 exploitation, the exploitation also involves an exchange between two parties, but the exchange does not benefit the vulnerable individual. It is also an exchange that under societal norms should not occur at all. An example of this would be where a drug dealer sells drugs to an addict, ostensibly providing the addict with what he or she needs, yet, in reality is exploiting the addict's addiction in an exchange that society would frown upon.³¹⁵

Mayer argues that although exploitation is wrong, when there is an exchange, it is usually a wrongful action that benefits the exploiter as well as their victim, as is typically the case in Class 2 exploitative transactions. In the case of the Anhui villagers in China, an exploitative transaction undoubtedly occurred. The villagers received something, albeit less than promised. It may be said that their participation in the study left them better off than before. This presents a challenge to policymakers because if successful asthma drugs are indeed developed from the research results of this collaboration, why would the study have needed to stop? In my view, it remains morally wrong that the villagers did not receive fair benefits from the study. Perhaps if there had been a legally binding benefit sharing arrangement between the villagers and the researchers, the villagers would have been awarded more favourable benefits.

Accordingly, to avoid Class 2 and 3 exploitative transactions in human health research, it is evident that there is a need for authentically negotiated benefit sharing arrangements

³¹³Pomfret J, Nelson D "An isolated region's genetic mother lode" 2000-12-20 https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-mother-lode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

³¹⁴Xiong L and Wang Y 2002 "Harvard University's genetic research in China is illegal" *Outlook Weekly* 15:48–50.

³¹⁵Mayer R "What's wrong with exploitation?" 2007 Journal of Applied Philosophy 24(2):137–150.

³¹⁶Mayer R "What's wrong with exploitation?" 2007 *Journal of Applied Philosophy* 24(2):137–150.

with *fair* benefits for the resource providers. The case in Anhui, China could easily have been a case in a village in Limpopo in South Africa, confirming the need for drafting a national benefit sharing agreement for health research in South Africa.

When benefit sharing does not take place, exploitation becomes possible, which is more troublesome when it occurs in vulnerable populations. Critics of benefit sharing in health research suggest that the undue inducement of vulnerable populations should not take place since the more vulnerable the population, the more a small benefit may constitute an undue inducement.³¹⁷ This statement begs the question: what is an undue inducement and how does it relate to fair benefit sharing?

2.3.3 Undue Inducement

When research partipants, who may be vulnerable persons or communities, are coerced—either covertly or overtly—to participate against their better judgement in a health research project, the coercions may be construed as undue inducements. Emphasis is placed on informed consent of research participants in ethical guidelines relating to human research. Coercion and undue inducement are both unethical methods of obtaining consent.³¹⁸

The term undue inducement was coined to protect participants in health research from harm to themselves in return for payment or some benefit. The CIOMS guidelines caution that "the payments should not be so large [...] or the medical services so extensive as to induce prospective subjects to consent to participate in the research against their better judgment (undue inducement)."³¹⁹

Undue inducements are problematic for the following reasons:320

- 1. They jeopardise the voluntary nature of informed consent.
- The research participant might accept a risk that would not otherwise be acceptable.

³¹⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³¹⁸Chennels RS "Equitable access to human biological resources in developing countries: benefit sharing without undue inducement" (PhD Thesis School of Health University of Lancaster 2014).

³¹⁹CIOMS 2002: Guideline 7.

³²⁰Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

- 3. Research participants might participate in research against their better judgment.
- 4. The entire practice of rewards or inducements is alien to the noble ideals and altruistic intent of the medical practice.

Inducements are an intrinsic part of life and not all are necessarily undue. Yet, there is rising concern that health research conducted in LMICs where economic challenges are prevalent, may affect the voluntariness of informed consent owing to undue inducements such as access to proven efficacious but unlincenced technologies, ancillary care that is better than local standards of care, financial reimbursements for participation and other unidentended benefits if participants choose to share or sell investigational drugs. 321 Research participants in LMICs are often subjected to poverty and have minimal or hardly any access to healthcare services. In such an instance, even the benefit of better healthcare during research participation could be construed as an undue inducement. Emmanuel $et\ a^{g22}$ refer to an example of HIV research participants receiving antiretroviral treatments that would improve their health. Is it unethical for these participants to participate in a study when their status quo may improve? How can this incentive diminish their freedom to choose to participate in the study? Anarson and Van Niekerk 323 suggest that "desperate need is not sufficient to undermine consent" and that *denying* people the right to take inducements may be infringing upon their freedom of choice.

The argument that research participants would take more risks when offered undue inducements is not convincing. A research subject freely chooses to participate in a study, which is mandated to be cleared by a research ethics committee or institutional review board globally, as is the case in South Africa. These bodies are tasked, *inter alia* with evaluating the risk–benefit ratio of the research.³²⁴ Some risks are of such a nature that they outweigh any potential benefits that may arise from the research. With research involving HBMs, such as genome research or tissue research, the risk of physical harm may be minimal; whereas serious ethical considerations may arise due to privacy issues

³²¹Mngadi KT, Singh JA, Mansoor L and Wassenaar DR "Undue inducement:a case study in CAPRISA 008" 2017 *J Med Ethics* 43:824-828.

³²²Mngadi KT, Singh JA, Mansoor L and Wassenaar DR "Undue inducement:a case study in CAPRISA 008" 2017 *J Med Ethics* 43:824-828.

³²³Anarson G and Van Niekerk A "Undue fear of inducements in research in developing countries" 2009 Cambridge Quarterly of Healthcare Ethics 18(2):122–129.

³²⁴Emmanuel EJ, Currie XE and Herman A "Undue inducement in clinical research in developing countries: is it a worry?" 2005 *The Lancet* 366:336–340.

and potential social harms, such as stigmatisation and discrimination of participants and their families.

According to Emmanuel,³²⁵ the claim that undue inducements will persuade research participants to make decisions against their better judgement is unfounded. The purpose of incentives is to induce people to do what they would otherwise not. Emmanuel³²⁶ points out that if research is ethical, excessive inducements would not cause people to take unreasonable risks. Accordingly, if a research project has been granted ethical clearance by the relevant IRB and REC, it should follow that the risks of harm to the research participants will be either negligible, and if moderate or high, that these have been carefully considered by the IRB or the REC, together with relevant precautions and interventions to ensure the continuing safety of participants.

The idea that research should be altruistic and that there should thus not be a need for inducements is unsound. Lee argues that with monetary inducements, it is not that offering any amount is wrong, but that the problem is rather that offering too much money which could result in undue inducement, is wrong. HICs and LMICs do not share the same resources. Suggesting that sharing benefits with participants should be discouraged when research is conducted in LMICs because of the undue inducement prohibition is akin to an exploitative transaction. According to Macklin, this is another form of the double standard in medical research, where ethical standards formulated in HICs have the potential to disadvantage LMICs.

As stated earlier, the NHREC in South Africa is tasked with developing national ethical guidelines for the functioning of health research ethics committees (RECs), as well as registering and auditing RECs.³³⁰ Section 1.4.3 of the Guidelines requires that all RECs reviewing research involving human participants must register with the NHREC.³³¹ This is

³²⁵Emmanuel EJ, Currie XE and Herman A "Undue inducement in clinical research in developing countries: is it a worry?" 2005 *The Lancet* 366:336–340.

³²⁶Emmanuel EJ, Čurrie XE and Herman A "Undue inducement in clinical research in developing countries: is it a worry?" 2005 *The Lancet* 366:336–340.

³²⁷Lee E "Our flawed approach to undue inducement in medical research" 2019 *Bioethics* 33:13-18.

³²⁸Lee E "Our flawed approach to undue inducement in medical research" 2019 *Bioethics* 33:13-18.

³²⁹Macklin R "Double standards in medical research in developing countries" 2004 *Cambridge University Press*, Cambridge.

³³⁰Department of Health Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

³³¹Department of Health Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

important because it means the individual research participant should be protected from research with an unfavourable risk-benefit ratio, owing to REC oversight.

The Guidelines refer in Section 3.1.7³³² specifically to the issues of reimbursements and inducements for participants. The guidelines stipulate that all research participants are entitled to reimbursement for travel and time. Regarding inducements, the guidelines state that "inducements encourage participation and may be offered in circumstances where[,] e.g.[,] recruitment, especially of healthy participants, is anticipated to be difficult". This is useful and positive because as previously stated, not all inducements are necessarily undue in nature.

The notion of undue inducements is historically rooted in a very literalist interpretation of the notion that health research should be altruistic and that all ethical research should avoid harm. The Department of Health's Ethics in Health Research Guidelines refer in Section 2.1³³³ to distributive justice by stating that "there should be a fair balance of risks and benefits amongst all role-players involved in research". It is necessary to briefly canvass what altruism in health research entails, as discussed next.

2.4 Altruism and justice in health research

2.4.1 Altruism in health research

Altruistic participation in health research means that the research will benefit all. It is founded on the common heritage of humankind with the general view that the underlying purpose of all research is the creation of generalised knowledge. Considering that health research is routinely conducted in LMICs whose participants may be vulnerable to exploitation, doubts have been cast on the altruism and solidarity model in health research. There is a substantial distinction between practising altruism in health

³³²Department of Health Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

³³³Department of Health Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

³³⁴Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18(2):165–170.

³³⁵Lairumbi G, Parker M, Fitzpatrick R and English M "Forms of benefit sharing in global health undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities in medicine* 7(1):1–8.

³³⁶Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioeth* 5(3):135–143.

research in HICs and doing so in LMICs.³³⁷ The benefits in HICs are numerous and for all citizens in HICs, altruistic health research possibly leads to the following:338

- 1. Ever-increasing numbers of medical interventions to achieve and maintain health tailored to local health needs and, in principle, accessible to all.
- 2. Increased knowledge about human health is made available to citizens through general education or health campaigns.
- 3. The availability of jobs in a high-tech industry (pharmaceutical research) and various related sectors (e.g., academia) and indirectly the very infrastructure and institutions that make such jobs possible.
- 4. Profits for commercially oriented research companies the and pharmaceutical production and retail industry.

These benefits are not readily available in LMICs and if one views benefit sharing in terms of justice, the absence of these benefits for LMICs may be considered an injustice.

2.4.2 Justice in health research

The principle of justice takes many forms. *Justice in exchange* establishes the fairness or equity of transactions, whereas distributive justice espouses that the risks and benefits are shared equitably in society. Corrective justice redresses an injustice; retributive justice determines which punishment is appropriate for a given crime. 339 According to Article 25 of the 1948 Universal Declaration of Human Rights, all humans have the right to healthcare through the notion of distributive justice;340 yet it is an added injustice to take something

³³⁸Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9-32.

³³⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9-32.

³³⁹Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 9-32.

³⁴⁰UN Universal Declaration on Human Rights https://www.ohchr.org/en/human-rights/universaldeclaration/translations/english (Date of use: 23 May 2023).

from another person without allowing the sharing of benefits, which, in turn, defeats the purposes of *justice in exchange*.³⁴¹ Consequently, this is a form of exploitation.

However, this does not necessarily imply that health research in HICs is always altruistic and void of exploitation. When healthcare research becomes privately funded enterprises, it becomes difficult to shield communities from exploitation.³⁴² This is evident in health research involving human tissue.³⁴³ For example, in the USA, biotechnology has brought about a revolution in human tissue research that has turned human tissue into a commodity and an object of property. In terms of this model, human tissue may be modified and hence commodified as material, leading to big business in the USA as well as globally.³⁴⁴ Hospitals are in the unique position of being able to supply huge amounts of tissue and medical information to the biotechnology industry to generate revenue.³⁴⁵ For example, the Beth Israel Deaconess Medical Centre (BIDMC), the teaching hospital of Harvard Medical School, is one such hospital. BIDMC specialises in patient care, biomedical research and teaching.³⁴⁶ In 2002, BIDMC began a collaboration with Ardais Corporation.³⁴⁷ Ardais Corporation is a private company which offers "healthcare services with a focus on tissue samples for use as clinically relevant and statistically valid human disease models such as cancer, inflammation, and vascular diseases". 348 BIDMC provides Ardais with residual human tissue from biopsies or other HBMs, together with their associated medical information (typically data from a tissue bank or biobank), for which BIDMC would then be financially compensated.

The human genetic resources that BIDMC trades to Ardais derive from altruistically donated samples. The donors of these resources sign an informed consent agreement

³⁴¹Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³⁴²Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

³⁴³Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁴⁴Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁴⁵Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁴⁶Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁴⁷Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁴⁸Ardais Corp https://www.bloomberg.com/profile/company/551614Z:US (Date of use: 23 May 2023).

which primarily serves to protect their identity and privacy.³⁴⁹ This agreement, *inter alia*, stipulates that donors:

Understand and agree that the tissue they donate as participants in the research program becomes the permanent property of the BI Medical Centre, and that BI Medical Centre does not make provision for compensation to donors in the event of product testing of commercial development.³⁵⁰

BIDMC treats the donated samples and their associated data as its legal property and then transfers the use of these samples via a license to Ardais. This license permits Ardais to sublicense the tissue to other companies that can provide revenue.³⁵¹ The BIDMC consent form expressly states that "there will be no direct benefit" under the guise of distributive justice which would accomplish healthcare for all. One may rightly ask who is truly benefitting from this altruism and if so, to what extent?³⁵² It may be argued that by using reputable hospitals and medical centres that people trust, the researchers coerce patients to sign informed consent forms requiring altruistic donation of tissue for the broader aim of serving research that would benefit the collective, yet this situation in reality may be said to lead to the unfair allocation of benefits, which could be considered exploitation.³⁵³

2.4.2.1 Fair benefit: the case of the Majengo sex workers

To date, a good example of a fair benefit agreement in human health research carried out in a LMIC is the Majengo sex workers study that was conducted in the slums of Nairobi in Kenya.³⁵⁴ AIDS has been endemic in sub-Saharan Africa for a number of decades, and has killed and infected millions of people, yet a vaccine to suppress it remains elusive.³⁵⁵ In 1980, an infectious disease expert from Canada, Francis Plummer, noticed that a percentage of the 2000 sex workers recruited for a study on sexually transmitted diseases

³⁴⁹Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁵⁰Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁵¹Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁵²Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁵³Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁵⁴Andada P, Cook Lucas J. *Majengo HIV/AIDS research case. A report for GenBenefit* 2007 http://www.uclan.ac.uk/genbenefit (Date of use: 27 May 2023).

³⁵⁵Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioeth* 5(3):135–143.

(STDs) remained uninfected with HIV, despite their high-risk behaviour.³⁵⁶ It was thought that this observation could lead to the development of a vaccine.

A collaborative study by the universities of Oxford (UK), Nairobi (Kenya) and Manitoba (Canada), including the UK Medical Research Council and the AIDS Vaccine Initiative and Uganda Virus Research Institute, commenced in 1998.³⁵⁷ Funded by the Canadian government and the public health of Nairobi City Council, vaccine trials started in 2001 and progressed to clinical trial Stages I and II.³⁵⁸

A clinic that also served as a research facility was set up in the Majengo slums to provide basic outpatient medical care for female sex workers. With the sex workers' informed consent, blood, cervical, vaginal and saliva samples were obtained from these women and used to study the epidemiology and immunology of HIV. In 2005, national guidelines for the research and development of HIV/AIDS vaccines were developed in Kenya, specifically in response to this case.³⁵⁹ Although no effective vaccine against HIV was developed from these studies, the research in this study provided the foundation for understanding the epidemiology of HIV and the risk factors associated with its spread.³⁶⁰

The Kenyan guidelines provide for the "fair and equitable sharing of benefits" arising from the research results attained from biological materials.³⁶¹ In this regard, the fair benefit was that the socioeconomically disadvantaged sex workers have received access to healthcare and free antiretrovirals since 2005, whereas the researchers obtained sound research

³⁵⁶Cook Lucas J, Schroeder D, Arnason G, Andanda P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–128.

³⁵⁷Cook Lucas J, Schroeder D, Arnason G, Andanda P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–128.

³⁵⁸Andada P and Cook Lucas J *Majengo HIV/AIDS research case. a report for GenBenefit* 2007. http://www,uclan.ac.uk/research/explore/projects/assets/cpe_genbenefit_nairobi_case.pdf (Date of use: 20 October 2019).

Phase I clinical trials determine the best way to administer a drug, its frequency and the maximum tolerated dose. Also, possible side effects and if the treatment is safe.

Phase II trials have a larger cohort of research participants than Phase I. These trials determine the most successful dose, evaluate potential efficacy and characterise the treatment benefit for the disease.

³⁵⁹Kenyan Ministry of Health Kenyan Ministry of Health National Guidelines 2005 available at: https://www.globalgiving.org/pfil/1108/projdoc.pdf (Date of use: 23 May 2023).

³⁶⁰Bandewar SVS, Kimani J and Lavery JV "The origins of a research community in the Majengo observational cohort study" BMC Public Health available at: https://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-10-630 (Date of use: 23 May 2023).

³⁶¹Ministry of Health National Kenyan Guidelines for Research and Development of HIV/AIDS Vaccines Appendix 5: Biological Material Transfer Agreement. Kenya: Ministry of Health 2005. http://www.globalgiving.org/pfil/1108/projdoc.pdf (Date of use: 23 may 2023).

results. Considering the stigma women in LMICs suffer in the sex work trade, and the lack of access to adequate healthcare, this is regarded as a fair exchange.

One cannot ignore the fact that the concepts of vulnerability, exploitation, benefit sharing and undue inducements are inevitably entwined with human health research.³⁶² However, although a very fine line exists between unethical inducement and appropriate benefit sharing, this should not present an obstacle to the development of appropriate benefit sharing mechanisms, especially in LMICs.³⁶³

2.5 Benefit sharing models

The aforementioned discussion has determined that to avoid the exploitation of research participant as a sample donor, even when the donation is altruistic, some benefit sharing in the advantages or profits arising from the use of their human biological resource must take place.³⁶⁴ The next section will evaluate some of the existing benefit sharing models.

2.5.1 The private benefit sharing model

In private biobanks or private tissue banks, private biotechnology companies act as brokers of tissue and health data for many researchers.³⁶⁵ However, as seen in the collaboration between BIDMC and ARDAIS Inc. above,³⁶⁶ it is evident that the role of academic teaching institutions as suppliers of HBMs to these private biobanks needs close monitoring.³⁶⁷ Winickoff and Winickoff³⁶⁸ suggest that in the USA, where many such collaborations exist, several challenges arise in the existing system of federal oversight, despite compliance with federal regulations.

³⁶²Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013).

³⁶³Thambisetty S "Human genome patents and developing countries" http://www.iprcommission.org/papers/pdfs/study_papers/10_human_genome_patents.pdf (Date of use: 23 May 2023).

³⁶⁴Schroeder D "Benefit sharing: it's time for a definition" 2006 *J Med Ethics* 205–209.

³⁶⁵Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12):1180–1184.

³⁶⁶Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁶⁷Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 *New Engl J Med* 349(12): 1180–1184.

³⁶⁸Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12): 1180–1184.

South Africa has unique shortcomings pertaining to tissue banks and biobanks. Biobanks, which can store millions of HBMs, including human tissue and their associated data, are not specifically regulated in terms of South African law.³⁶⁹ Despite the increase in demand for human tissue in the medical context, SA does not have a national tissue procurement agency, nor a national tissue authority.³⁷⁰ Biobanks are regulated by a very flexible legal framework broadly based on regulations relating to tissue banks and the Department of Health's Ethics in Health Research Guidelines, in contrast to human tissue banks, which are governed by a specific dedicated set of regulations promulgated in terms of Chapter 8 of the National Health Act.³⁷¹

The acknowledgement of benefit sharing poses unique challenges in the case of human biobanks and tissue banks that store and distribute human biological materials (DNA, RNA, blastomeres, polar bodies and human tissue) and potentially their associated medical data in perpetuity.³⁷² For example, who should benefit from the primary and secondary use of these samples?

2.5.2 The altruistic benefit sharing model

The constitutional republic of Iceland proposed an ambitious model of benefit sharing in 1998. Driven by the notion that all health research should be altruistic, the Icelandic government passed the Act on a Health Sector Database (HSD Act).³⁷³ This law allowed the Minister of Health to grant a license to a private American company, deCODE Genetics, to construct and operate a centralised database linking medical records with genealogical and genetic information.³⁷⁴ The biggest problem with the Act was that it provided for presumed consent, meaning that consent would be assumed and that a person could only be excluded from the database if they specifically objected. This health sector database, however, never materialised as the Supreme Court of Iceland found the

³⁶⁹Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediments to accessing health care" 2019 *SAJBL* 12(1):27–31.

³⁷⁰Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediments to accessing health care" 2019 S*AJBL* 12(1):27–31.

³⁷¹Department of Health, South Africa *Ethics in health research: principles, processes and structures* 2nd edition Pretoria DoH.

³⁷²Dhai A, Mahomed S and Sanne I "Biobank and human health research: balancing progress and protections" 2015 *S Afr J BL* 9:55–59.

³⁷³Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁷⁴Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

HSD Act to be unconstitutional for violating citizens' privacy. The Act itself did not require any benefit sharing arrangements, but Article 4 of the Act allowed for negotiations of any benefit sharing agreements to be conducted between the health minister and the licensee. Two specific benefit sharing arrangements were subsequently concluded, one between the pharmaceutical company, Roche, and deCODE, for free pharmaceutical and diagnostic products for the Icelandic people, and a second between the Ministry of Health and deCODE as part of the operating licence. Despite these benefits, it is has been argued that the use of presumed consent and the government laying claim to all medical records of the Icelandic population, points to the Icelandic government exploiting its own population.

2.5.3 The charitable trust benefit sharing model

A trust involves a fiduciary relationship in which trustees hold the title to property but are obligated to keep or use said property for the benefit of the beneficiary.³⁷⁹ This model suggests that if hospitals solicit altruistic donations of HBMs, then the hospitals should act as trustees of the HBMs donations rather than brokers.

Winickoff and Winickoff advance the argument that "in order to protect rights and scientific value", biobanks should be based "on a new form of agreement among the medical institution, the researcher, and the donor community: one modelled on the charitable trust". This model could also be used for tissue banks. This model proposes that informed consent should be a continuous process of communication with the donor of the genetic resource or tissue. Since a trust involves a fiduciary relationship, the trustee holds legal fiduciary duties towards the human biological resource but must keep or use the

_

³⁷⁵Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 95–127.

³⁷⁶Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

³⁷⁷Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M Donating human samples: who benefits? cases from Iceland, Kenya and Indonesia in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

³⁷⁸Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁷⁹Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁸⁰Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12):1180–1184.

property for the benefit of the specified party, i.e., the beneficiary/donor.³⁸¹ All researchers wishing to use the samples in the tissue bank or biobank would require to propose the project to both an IRB or a REC, as well as the trust's ethics committee. This model is thought to present the best alternative to modern biobanks and tissue banks because it presents clear ethical, legal and scientific advantages.³⁸²

Such a model is used by various rare disease groups that have constructed tissue banks to enable researchers to control research design, implementation and benefit.³⁸³ One such example is PXE International, a non-profit foundation that collects HBMs and their associated health data from approximately 1000 volunteers affected by Pseudoxanthoma elasticum (PXE).³⁸⁴ PXE is a rare, hereditary connective tissue disorder that affects the skin, eyes and cardiovascular system. ³⁸⁵ The goal of the organisation is to induce researchers to study the genetic basis of PXE to develop therapies.³⁸⁶ PXE International serves as the trustee of all collected samples; all research using PXE tissue must be cleared by its own IRB and an ethics committee of PXE International.³⁸⁷ PXE International informs all its donors of all the research projects and is also in constant communication with its collaborating research groups.³⁸⁸ Potential benefits to the group are linked to specific research and these are communicated. PXE International has been extremely successful as it was already collaborating with as many as 17 laboratories by 2003.³⁸⁹

Another successful organisation based on the charitable trust model of benefit sharing is the United Kingdom's Human Tissue Authority (HTA). The HTA was created by the Human Tissue Act of 2004. It serves to regulate activities concerning the removal, storage, use

³⁸¹Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 *New Engl J*

Med 349(12):1180–1184. ³⁸²Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J

Med 349(12):1180–1184.

383Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003

Jurimetrics 43(2):187–228.

³⁸⁴Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁸⁵Laube S and Moss C "Pseudoxanthoma elasticum" 2005 *Archdischild* available at: https://adc.bmj.com/content/archdischild/90/7/754.full.pdf (Date of use: 23 May 2023).

³⁸⁶Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁸⁷Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁸⁸Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

³⁸⁹Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

and disposal of human bodies, organs and tissue.³⁹⁰ The HTA serves as the trustee of all human tissue collected in the UK.

2.6 Conclusion

Benefit sharing in human health research is considered to be one of the benchmarks for ethical research in LMICs.³⁹¹ It has been argued by some scholars that health research should be altruistic to avoid exploitation by incentivising potential participants with undue inducement.³⁹² Altruism is a concept that is easily attained in research within HICs where health systems function and health research can be of benefit to all. However, benefit sharing for health research in LMICs is a central concern and is difficult to attain in the absence of legal frameworks. In LMICs, it is innately exploitative to allow participation in health research without participants receiving any benefits, especially where no prospect exists that participants may have access to the benefits (such as pharmaceuticals) once a study ends.³⁹³

A situation where research participants in LMICs bear all the risks and burdens of participating in research without accruing any benefits, is a failure of *justice in exchange*.³⁹⁴ The question of undue inducements is pertinent when harm to participants is likely. It is assumed that with most genetic research, where only a swab is required (which is of minimal physical risk), the question of undue inducements is less relevant. However, because genetic swabs involve social risks which may trigger genetic discrimination, stigmatisation and stereotyping, the last-mentioned all high risks, the question of undue inducements become highly pertinent.

Schroeder's³⁹⁵ suggestion that benefit sharing is a technical term that is used as a tool to achieve communicative justice should be accepted. When research participants are

³⁹⁰UK Human Tissue Authority mandate https://www.hta.gov.uk/policies/human-tissue-act-2004 (Date of use: 26 May 2023).

³⁹¹Emmanuel EJ, Wendler D, Killen J and Grady C "What makes clinical research in developing countries ethical? The benchmarks of ethical research" 2004 *J Infect Dis* 189:930–937.

³⁹²Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducemant" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³⁹³Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducemant" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³⁹⁴Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducemant" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

³⁹⁵Schroeder D "Benefit sharing: it's time for a definition" 2006 *J Med Ethics* 205–209.

vulnerable, it is necessary to correct such injustice, which makes benefit sharing instrumental in preventing exploitation.³⁹⁶

Benefit sharing is indeed a proportionally modest and indirect way to achieve respect for the human rights of health for all. It is of grave concern that many governments, including SA, do not have templates for benefit sharing agreements in their legal frameworks for health research.

The next chapter explores the question of ownership and the commodification of HBMs, as well as the ethico-legal framework relevant to benefit sharing in South Africa, with particular emphasis on biobanks, tissue banks and genomic research benefit sharing arrangements.

_

³⁹⁶Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducemant" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 9–32.

CHAPTER 3 ETHICO-LEGAL FRAMEWORK RELEVANT TO BENEFIT SHARING

3.1 Introduction

The notion of benefit sharing in health research requires the identification of the parties that are to share the benefits. In health research involving human biological materials (HBMs), this identification may be challenging³⁹⁷ and is further compounded by the issue of ownership of the materials. The issues and controversies regarding ownership of HBMs that are donated for medical research have become long-standing and ongoing legal disputes³⁹⁸ which are key to discussions around benefit sharing. The commercialisation of the products resulting from research using HBMs, brought about by the buying and selling of HBMs, is a complex and often controversial issue in health policy.³⁹⁹ The proliferation of health research in LMICs⁴⁰⁰ and the boom in biobanks⁴⁰¹ in these countries, including South Africa, have challenged perceptions regarding the human bodies in law.⁴⁰²

It is globally understood that the sale of and trade in human tissue is prohibited.⁴⁰³ In South Africa, section 60(4)(b) of the National Health Act enforces this prohibition.⁴⁰⁴ According to Mahomed *et al*,⁴⁰⁵ this universally understood principle stems from the traditional belief that persons cannot own their bodies and accordingly their HBMs too, since this owning of oneself implies the objectification of oneself and in so doing, allows the objectification of one's body by others.

³⁹⁷ Mahomed S "An ethico-legal framework for the regulation of biobanks in South Africa" (PhD Bioethics and Health Law University of the Witwatersrand Johannesburg 2018) Available at: https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-

⁶⁵e36ff9563e/content (Date of use: 26 May 2023).

398Mahomed S, Nöthling-Slabbert, M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

³⁹⁹Bjorkman B and Hansson SO "Bodily rights and property rights" 2006 J Med Ethics 32:209–214.

⁴⁰⁰Lairumbi G, Parker M, Fitzpatrick R and English M "Forms of benefit sharing in global health research undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities in medicine* 7(1):1–8.

⁴⁰¹Schroeder D and Lucas J "Benefit sharing: from biodiversity to human genetics-an introduction" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 1–8.

⁴⁰²Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁰³Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁰⁴Section 60(4) of the NHA.

⁴⁰⁵Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

The law has generally not been consistent in clarifying the issue of treating persons and their HBMs as property. However, in efforts to promote health research, the law has been clear regarding the issue of intellectual property emanating from research involving HBMs.⁴⁰⁶

Because the concept of benefit sharing aspires to protect and benefit all participants in health research, it requires that the notion of ownership of HBMs, as well as to whom this might apply, must be explored.

To address this, aligned with the objective of developing a standard framework for benefit sharing agreements in South Africa, this chapter will first discuss the current ethico-legal framework governing health research involving HBMs in South Africa. Where relevant, foreign judgements dealing with the so-called ownership of HMBs will be referred to, as there is presently no legal precedent concerning the ownership of HBMs in South Africa.⁴⁰⁷

Gibson⁴⁰⁸ rightly argues that the law relies on tradition and legal precedent, however, in the current age of genomics, tissue engineering and biotechnology, novel situations arise, leading to legal disputes involving these technologies. The challenge is that while the law must be seen to both follow tradition and precedent when formulating public policy or legislation, it should also be responsive and flexible enough to respond to changes requiring legal regulation. Before the ethico-legal framework is discussed, the notion of ownership in HBMs will first be canvassed.

3.2 Ownership of HBMs in the era of biotechnology

As stated above, it is universally recognised that no person can own another person since this would constitute objectification by another, which is tantamount to slavery. However, the question of a person's ownership of his or her own body is more complicated. In law, the human body and its parts have traditionally been classified as *res extra commercium* (a thing outside the commercial sphere), while separated bodily materials are considered *res nullius* (belonging to no one) until they are brought under the control of the

⁴⁰⁶Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁴⁰⁷Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁰⁸Gibson SF "The Washington University v Catalona: Determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁴⁰⁹Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

first person who obtains possession of the separated HBMs.⁴¹⁰ If the recipient has processed the HBM in some way, the recipient gains a proprietary interest or an additional series of rights and in some cases, a right of ownership in these HBMs.⁴¹¹

The commercialisation of products emanating from medical research using HBMs has resulted in difficult legal and ethical questions, which often lack definite answers. It is the lack of precise answers that adds to the ongoing debate regarding who owns human tissue donated for medical research. The question of treating the human body and HBMs as property presents complex legal, ethical and philosophical dimensions. The legal meaning of property asserts individual autonomy, describing a legal relationship between persons and objects, enabling the proprietor to exercise exclusive control over a said object. According to the Oxford Dictionary, property is synonymous with ownership in law.

Some scholars maintain that patients have ownership rights of their samples in perpetuity, including an unfettered right to determine what happens to these samples.⁴¹⁶ This is contrasted with the view that giving patients property rights to their samples will turn the human body into a commodity and hinder research.⁴¹⁷

3.2.1 Ethico-legal issues relating to the ownership of HBMs and human dignity

The Bill of Rights of the Constitution of South Africa, 1996, states in section 10 that "everyone has inherent dignity and the right to have their dignity respected". ⁴¹⁸ There is also international consensus that research using HBMs should respect human dignity and human rights. ⁴¹⁹

⁴¹⁰Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

⁴¹¹Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁴¹²Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁴¹³Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴¹⁴Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

⁴¹⁵Concise Oxford English Dictionary 12th ed 2011.

⁴¹⁶Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁴¹⁷Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁴¹⁸ The Constitution of South Africa 1996.

⁴¹⁹Beyleveld D and Bronsword R "Human dignity, human rights, and human genetics" 1998 *Modern Law review* 61(5): 661–680.

The concept of human dignity demands that human beings have intrinsic value and cannot be commodified, as it violates their dignity and worth. Moreover, if human beings can commodify themselves and have a market value, they can be treated as mere objects by themselves and others. As stated above, the human body and its parts are traditionally said to be *res extra commercium*, (things outside of the commercial sphere) but once HBMs are separated from the body, they are deemed *res nullius* (belonging to no one) up until the first person who takes possession of them and intends to use them, who then acquires exclusive legal control by *occupatio*. There are numerous examples in law that point to the law's uneasiness with classifying HBMs in terms of property, ownership or possession. It appears that the law accepts that when an excised HBM has been subjected to labour and skill, such as the cultivation of tissue or the manipulation of the tissue, the persons who performs these actions with regard to the tissue may claim property rights in the tissue product that is developed.

In terms of South African law, ownership is a real right. The law entitles the recipient of HBMs to a real right and ultimate control over the HBMs.⁴²⁵ The common law description of ownership as found in case law is described thus as:

The most complete real right which gives the owner the most complete and absolute entitlements to a thing. Even so, it is a right which can be limited by objective law and by the rights of others (limited real rights or creditor's rights).⁴²⁶

Because ownership is a real right, it is often defined based on entitlements.⁴²⁷ Thus, the holder of the real right may control, use, encumber, alienate/transfer and vindicate the thing in which they hold a real right. Should an institution or organisation procure an individual's donated HBMs, they would be regarded as owners of the samples and would, unless the contrary is indicated and agreed upon by the parties, not be required to inform

⁴²⁰Resnick DB "The commodification of human reproductive material" 1998 *Journal of Medical Ethics* 24:388–393.

⁴²¹Resnick DB "The commodification of human reproductive material" 1998 *Journal of Medical Ethics* 24:388–393.

⁴²²Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

⁴²³Nöthling-Slabbert M "Human bodies in law: Arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

⁴²⁴Swain MS and Marusyk RW "An alternative To Property Rights in Human Tissue" 1990 Hastings Centre Report 20(5):12–16.

⁴²⁵Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 S Afr J BL 6(1):16–20.

⁴²⁶Van der Walt AJ and Pienaar GJ *Introduction to the law of property* 7th ed (Juta 2017) 43–56.

⁴²⁷Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

the donor of any subsequent use of the sample and/or provide any information on the progress of the research process. This situation prompts the question whether such an arrangement may withstand legal and ethical scrutiny, particularly with regard to informed consent requirements, as well as the provisions of the Regulations relating to Research with Human Participants, and to mention potential violations of constitutional rights. The need to protect and respect inherent human dignity and a person's right to bodily and psychological integrity, including the prerequisite that a person should not be subjected to medical or scientific experiments without their informed consent, are outlined in the Bill of Rights of the Constitution.

Given the common law position that no man is *dominus membrorum suorum* (master of his bodily members) and the universally accepted maxim that the human body is *res extra commercium* (a thing outside of the commercial sphere), the question arises as to whether and under which circumstances South African law would recognise ownership of HBMs. Should such proprietary rights be found to exist, what would the consequences of such a classification in terms of South African law be. Put simply, this comes down to the question regarding what rights individuals have over their HBMs.⁴³²

There are abundant examples in law that demonstrate the law's contradictions in making sense of the human body in the context of ownership and property.⁴³³ Examples in this regard relate to how the law regulates the various states of human transition (in other words, relating to a developing foetus and pregnant woman), activities that commodify the body and body parts, and more recently, in issues relating to developments in biotechnology and genetics.⁴³⁴

There is also a need to build trust with communities that value and attach great importance to blood and tissue collected for use as research samples, since they play an important

⁴²⁸Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 S Afr J BL 6(1):16–20.

⁴²⁹National Department of Health. Notice 177: Regulations Relating to Human Biological Material. Pretoria: Government Gazette, 2012.

⁴³⁰The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁴³¹The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁴³²Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 S Afr J BL 6(1):16–20.

⁴³³Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴³⁴Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

cultural role in traditional ceremonies.⁴³⁵ Indigenous communities⁴³⁶ are reluctant to participate in biobanking and genomic research because they are concerned with the handling, treatment and ownership of HBMs and the knowledge gained from the specimen analysis.⁴³⁷

It is evident that the outright classification of HBMs as property gives rise to complex ethical and legal questions.⁴³⁸ To promote fair and just altruistic research, researchers have proposed frameworks for ethical consent models which would mitigate potential conflict between biobanks, research participants, investigators and sponsors, especially when conflicting claims of ownership in HBMs arise.⁴³⁹ These ethical consent models obviate the need to determine proprietary rights in HBMs.

3.2.2 Proposed ethical models when considering the regulation of HBMs

One of the proposed models is premised on *custodianship*, involving the caretaking obligation for HBMs from their initial collection to the final dissemination of research findings. ⁴⁴⁰ The custodianship model approaches these issues from a broader perspective than the framework of ownership which views HBMs as property with exclusive rights vested in the owner. The custodian model recognises the altruism of research participants and deems their donated HBMs as 'gifts', and with the informed consent of the donors, promotes free and fair research for the good of society. Custodianship rejects the notion that HBMs are for-profit commodities.

Another proposed ethical framework model is that of *stewardship*, which implies that everyone involved in research is responsible for protecting human subjects' interests and well-being to the best of their ability.⁴⁴¹ This model seeks to decrease ethical conflict emerging from the ethical challenges that may arise for all persons involved in research

⁴³⁵Aramoana J and Koea J "An integrative review of the barriers to indigenous peoples' participation in biobanking and genomic research" 2020 JCO Global Oncology, 6: 83–91. On behalf of the CommNETS Collaboration. doi: 10.1200/JGO.18.0015.

⁴³⁶Indigenous peoples are the ethnic groups which are the original inhabitants of a given region, in contrast to groups who have settled, occupied, or colonised the area more recently.

⁴³⁷Aramoana J and Koea J "An integrative review of the barriers to indigenous peoples' participation in biobanking and genomic research" 2020 *JCO Global Oncology*, 6: 83–91. On behalf of the CommNETS Collaboration. doi: 10.1200/JGO.18.0015.

⁴³⁸Skene L "Ownership of human tissue and the law" 2002 Nature Reviews Genetics 3:145–147.

⁴³⁹Yassin R, Lockhart N, Del Riego MG *et al* 2010 "Custodianship as an ethical framework for biospecimen-based research" *Cancer Epidemiology Biomarkers Prev* 19(4):1012–1015.

⁴⁴⁰Yassin R, Lockhart N, Del Riego MG *et al* 2010 "Custodianship as an ethical framework for biospecimenbased research" *Cancer Epidemiology Biomarkers Prev* 19(4):1012–1015.

⁴⁴¹Jeffers BR "Human biological materials in research: ethical issues and the role of stewardship in minimizing research risks" 2001 *ANS Adv. Nurs. Sci 24*(2):32–46.

using HBMs. Stewardship arises from respect for human dignity and recognition of common humanity. Stewardship suggests that for researchers to retain public trust, investigators should design research protocols that respect and protect donors of the HBM and their communities. The responsibilities of a steward may include protecting research results and not sharing confidential genetic and linked health information that could promote the stigmatisation of particular populations.

Private biobanks are storing millions of HBM samples amassed from collaborations with medical institutions. A proposed consent model for private biobanks is that of a *charitable trust*. This model seeks to protect rights and maximise scientific value. The model suggests that informed consent should be a process of continuous communication between researchers and research participants, while a biobank operates as a trustee or steward of HBMs to ensure the protection of the samples. The trust agreement would see the HBMs' donor or settlor transfer their property interest in their donation to the trustee, who would assume legal fiduciary duties to keep the HBMs for the benefit of the named beneficiary in which the public acts as the beneficiary. The charitable trust model has clear ethical, legal and scientific advantages that accommodate altruism, good governance and benefit to the public.

It is worth noting that when dealing with indigenous populations, there exists a cultural link between an ancestral past to the present through blood and tissue. Genetic manipulation of their HBMs and the immortalisation of cell lines is troubling to some of these communities and thus, whichever ethical consent model is adopted, it must accommodate the indigenous community with ongoing input and the ability to influence specimen use and disposal.⁴⁴⁶

_

⁴⁴²Jeffers BR "Human biological materials in research: ethical issues and the role of stewardship in minimizing research risks" 2001 *ANS Adv. Nurs. Sci* 24(2):32–46.

⁴⁴³Fullerton SM, Nicholas RA, Guzauskas G *et al* "Meeting the governance challenges of next-generation biorepository research" 2010 *Science Translational Medicine* 2(15): 15cm3–15cm3.

⁴⁴⁴Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 *New Engl J Med* 349(12):1180–1184.

⁴⁴⁵Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 *New Engl J Med* 349(12): 1180–1184.

⁴⁴⁶Aramoana J and Koea J "An integrative review of the barriers to indigenous peoples' participation in biobanking and genomic research" 2020 *JCO Global Oncology*, 6: 83–91. On behalf of the CommNETS Collaboration. doi: 10.1200/JGO.18.0015.

3.3 South African legislative framework on the ownership of HBMs, intellectual property rights and benefit sharing

In South Africa, the issue of the ownership of HBMs is no less confusing than the universal confusion on this topic. This confusion mostly arises from an ambiguous definition of certain HBMs in the legislation.⁴⁴⁷ Mahomed *et al*⁴⁴⁸ address this issue and conclude that the ambiguity in the definition of HBMs, such as the definition of human tissue in South African legislation, makes it difficult to provide a clear and consistent message regarding any proprietary claims in respect of human tissue.

Some legal scholars argue that a close inspection of the common law and statutory law supports the conclusion that in the research context, HBMs are indeed susceptible to ownership.⁴⁴⁹ Other scholars, on the other hand, maintain that any attempt to conceptualise the ownership of HBMs inevitably leads to the objectification of a person, with the result of reducing such person to a state of mere property.⁴⁵⁰

It is clear from the aforementioned discussion that the issue of ownership in HBMs is fraught with complex and conflicting views. To gain a better understanding of the scope of the opposing views, the discussion below will turn to the existing regulatory framework relating to the ownership of HBMs, as well as the related issue of benefit sharing around such materials.

3.3.1 Constitution of the Republic of South Africa, 1996

The Constitution of South Africa was adopted in 1996⁴⁵¹ and is deemed the supreme law of South Africa, whereby any law or conduct inconsistent with it is invalid.⁴⁵² In Chapter 2 of the Bill of Rights of the Constitution, the right to human dignity is stated as a right that should be protected entirely.⁴⁵³ In addition to the protection of the right to human dignity in

⁴⁴⁷Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁴⁸Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁴⁹Thaldar DW and Shozi B "The legal status of human biological material used for research" 2021 *South African Law Journal.*

⁴⁵⁰Mahesh KP "Laws and regulations associated with ownership of human biological material in South Africa" 2015 *S Afr J BL* 15:8(1)11–18.

⁴⁵¹The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁴⁵²The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁴⁵³The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 March 2021).

section 10 of the Constitution, the Bill of Rights further recognises rights that are closely linked to dignity in section 12(2), which states that:

Everyone has the right to bodily and psychological integrity, which includes the right to; make decisions concerning reproduction; security and control over their body and not to be subjected to medical and scientific experiments without their informed consent.⁴⁵⁴

Transposing the provisions of the Bill of Rights to apply in a research setting becomes challenging, especially considering the mistrust that communities have about the motives of research when the historical abuse of African resources is not confined to the past.

Considering a case from as recently as 2018, discussed next, it is evident that the exploitation of African research participants and research institutions by research institutions in HICs is an ongoing trend that results in justifiable mistrust. The Welcome Sanger Institute is based in the United Kingdom and specialises in genome research. This institute was accused by, among others, the Universities of KwaZulu-Natal and Stellenbosch, of commercialising a gene chip developed using DNA donated by African donors. Allegedly, some of the MTAs used in the research did not give permission for the commercialisation of the gene chip, nor did the research participants consent to restrict commercialisation. The accusations resulted in a legal dispute and although an external legal investigation concluded that there was no "unlawful exploitation of scientific work" nor any breach of contract or intellectual property rights, the 75 000 gene arrays stored at Sanger were never used and expired at the end of 2019.

Unfortunately, it is not always foreign institutions that cultivate a culture of mistrust in research using HBMs with research participants; sometimes local institutions in LMICs play a major role. A case in point is that of Discovery, a South African health insurer which, in 2015, partnered with Human Longevity Incorporation, a company owned by Craig Venter and based in the United States of America (USA).⁴⁵⁷ Under this partnership, Discovery clients, all of whom were SA citizens, were offered genetic testing at a reduced cost (US\$)

⁴⁵⁴Chapter 2, Section 12(2) of the Constitution of South Africa 1996.

⁴⁵⁵ Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136– 138.

⁴⁵⁶Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136– 138.

⁴⁵⁷ Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136— 138.

250 per person) which was provided by the Human Longevity Incorporation. In return, Discovery members' samples and anonymised information would be transferred to the USA and stored at the Incorporation, supposedly for "clinical care and the aim of advancing medical research". Thousands of Discovery members' saliva samples were transferred to further this aim. 458 Subsequently, it became apparent that the real aim of this partnership was to create one of the world's biggest databases for whole genome, phenotype and clinical data by taking advantage of a legal *lacuna* that existed in South Africa at the time, specifically pertaining to a regulatory framework for the use, storage and export of biological samples. 459 This case illustrates the difficulty in applying the provisions of the Bill of Rights in the absence of legally binding frameworks for research using HBMs. In this particular case, scholars pointed out the need to protect research participants in SA.⁴⁶⁰ In addition to the mistrust created by such research partnerships between participants and researchers, and sponsors and researchers, other ethical issues also arose: collection of samples and data, and the storage, use and sharing of HBMs from African research participants without regard for culture, privacy, confidentiality, autonomy and the feedback of findings are creating a difficult research environment for South African researchers.461

The right to privacy is outlined in section 14 of the Bill of Rights in the Constitution. 462 On 1 July 2020, the Protection of Personal Information Act 4 of 2013 (POPIA) was enacted. 463 The Act provides a high-level, principle-based approach to the use of personal information—including genomic data—by outlining strict requirements that must be attained before personal information is processed and transferred outside of SA. It is worth noting that while the actual specimens derived from HBMs fall outside the remit of POPIA, any data derived from such samples would fall under the remit of POPIA as it is considered

_

⁴⁵⁸ Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136– 138.

⁴⁵⁹Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 *South African Medical Journal* 106(2):136–138.

⁴⁶⁰Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 *South African Medical Journal* 106(2):136–138.

⁴⁶¹Academy of Science of South Africa Consensus Study *Human genetics and genomics in South Africa:* ethical. legal and social implications 2018 found at: http://bit.lv/3sH15hH (Date of use: 23 May 2023).

⁴⁶²The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁴⁶³POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 23 May 2023).

to be personal information.⁴⁶⁴ Even when anonymised for research purposes, genetic information requires consent for future use. POPIA requires personal information to be collected for specific, explicitly defined and lawful purposes.⁴⁶⁵ As such, there exists a fine line between the exact point at a which a genetic sample becomes personal information.⁴⁶⁶

3.3.2 National Health Act 61 of 2003 and relevant regulations

The National Health Act 467 (NHA) came into effect in 2005. At that time, Chapter 8 of the Act, which deals with human tissue (such as blood, blood products, tissues and gametes) had not been enacted and the regulation of human tissue was governed by the former Human Tissue Act, which has since been repealed. 468 The NHA defines tissue as "human" tissue, and includes flesh, bone, a gland, an organ, skin, bone marrow or body fluid, but excludes blood or a gamete".469 Yet in the regulations relating to tissue banks, also promulgated in terms of Chapter 8 of the NHA, tissue is defined as "a functional group of cells used collectively in the regulations to indicate both cells and tissue". 470 The discrepancies in definitions become more confusing if the definitions of a cell in the NHA and regulations are compared. The regulations relating to the use of human biological material describe a cell as "the smallest structural and functional unit of an organism, consisting of cytoplasm and a nucleus enclosed in a membrane in living things". 471 This is in contrast with the definition of a cell in the Regulations relating to the Artificial Fertilisation of persons, also promulgated in terms of Chapter 8 of the NHA, where a cell is said to be "the basic structural and functional unit in people and all living things and is a small container of chemical and water wrapped in a membrane". 472 The lack of clear and consistent definitions in the NHA, illustrated by these examples, compromises the determination of the nature and status of these concepts, which in turn complicates clear inferences regarding possible uses and limitations regarding these entities.

⁴⁶⁴Personal information is loosely defined in the Act as information relating to an identifiable, living natural person (Section 1).

⁴⁶⁵POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf Section 13 (Date of use: 23 May 2023).

⁴⁶⁶Mahomed S and Staunton C "Ethico-legal analysis of international sample and data sharing for genomic research during COVID-19: a South African perspective" 2021 *BioLaw* 1 Journal available at www.biodiritto.org ISNN2284-4503 (Date of use: 23 May 2023).

 $^{^{467}61}$ of 2003.

⁴⁶⁸65 of 1983.

⁴⁶⁹61 Of 2003.

⁴⁷⁰South Africa. National Health Act No.61 of 2003. Regulations relating to tissue banks. Government Gazette No. 35099, 2012 (Published under Government Notice R182).

⁴⁷¹GN R177 in Government Gazette 35099 of 2 March 2012.

⁴⁷²GN R 1165 in Government Gazette 40312 of 30 September 2016.

In 2012, the Regulations Relating to the Use of Human Biological Material, ⁴⁷³ an important set of Regulations promulgated in terms of Chapter 8 of the NHA, were enacted. These regulations define HBMs as "material from a human being including DNA, RNA, blastomeres, polar bodies, cultured cells, embryos, gametes, progenitor stem cells, small tissue biopsies and growth factors from the same". Mahomed *et al*⁴⁷⁴ argue that this definition results in ambiguity, compared to the NHA's definition of tissue, which excludes gametes and the definition of tissue in the regulations relating to tissue banks, referring to tissue as a term used to collectively indicate both cells and tissues. The effect of these uncoordinated definitions is that the classification of human tissue remains open-ended. ⁴⁷⁵

3.3.2.1 Regulations Relating to the Use of Human Biological Materials (HBMs)

These Regulations provide in regulation 5 that HBMs can be removed or withdrawn from living persons for the following medical and dental purposes:

- a) DNA-, RNA- and chromosome-based genetic testing
- b) Health research referred to in Section 69(3) of the NHA
- c) Training referred to in Section 64(1)(a) of the NHA
- d) Studies of archaeological, medical or heritage value on DNA obtained from human genetic material, conducted in terms of the National Heritage Resource Act.

Although the issue of ownership is not addressed in these Regulations, the emphasis on the requirement of informed consent as a prerequisite for the removal and use of HBMs suggests that the law recognises that HBMs have an origin or source which needs to be recognised. Regulation 2(a) of the same regulations states that only a competent person (trained and registered by the Health Professions Act) is eligible to remove biological material for genetic testing, genetic health research or therapeutic purposes. Regulation 3 deals with the removal of biological materials from living persons for the purposes stated in Regulation 2(a) and explicitly states that no competent person may remove any HBMs unless it is done with the written informed consent of the person from whom the HBMs are removed.

⁴⁷³GN R177 in Government Gazette 35099 of 2 March 2012.

⁴⁷⁴Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁷⁵Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

Regulation 3(1)(b) of the Regulations provides that if a child is younger than 18 years but older than 12, written consent is required, provided the child is of sufficient maturity and has the mental capacity to understand the benefits, risks and social implications of the procedure. However, if a child is immature but above the age of 12, a parent, caregiver or guardian must provide informed consent similarly to if the child were below the age of 12. In the case of an emergency, informed consent could be provided by the head of the health establishment if the removal of HBMs from a person is required. Regulation 3 stipulates that the consent for the removal of HBMs may also be provided by the Minister (of Health), if a parent, guardian or caregiver of a child unreasonably refuses or is incapable of giving consent to assist a child, either because they are deceased or cannot be traced. In the case of mentally ill patients, consent may be provided by the patient if they are capable of doing so. In the case of a mentally ill patient who is incapable of consenting, a curator, spouse, next of kin, parent, major child, brother or sister, partner or associate can give consent. In the case of an emergency, the head of the health establishment could provide the relevant consent. Regulation 3(2) states that no person shall carry out genetic health research unless the research has been approved by a registered health research ethics committee.476

Regulation 4 addresses the removal of HBMs from deceased persons. Regulations 4(1) and 4(2) provide that any organisation, institution or person that intends to use tissue from a deceased person for purposes of genetic testing, health research and therapeutics, where no consent has been given by the deceased person before their death and where there is no evidence that the removal of the tissue or cells would be contrary to a directive given by the deceased before their death, such an organisation, institution or person must take steps to locate the spouse, partner, major child, parent, guardian, major brother or major sister of a deceased person, *in this specific order mentioned*, in order to obtain consent.⁴⁷⁷ Regulation 4(3) states that if none of the persons referred to is located, an application, including evidence that the above steps have been taken, must be submitted to the Director General with a request to remove the HBMs from the deceased.⁴⁷⁸

Regulation 9 refers to the use of adult, embryonic and umbilical cord cells for the purposes of stem cell therapy. It categorically determines that any competent person wishing to

⁴⁷⁶GN R177 in Government Gazette 35099 of 2 March 2012.

⁴⁷⁷GN R177 in Government Gazette 35099 of 2 March 2012.

⁴⁷⁸GN R177 in Government Gazette 35099 of 2 March 2012.

utilise these stem cells for stem cell therapy must obtain written informed consent from the donor of such stem cells.

With respect to compensation for the withdrawal of these HBMs, Regulation 11 cautions that a person whose HBMs are withdrawn may only be reimbursed for reasonable expenses incurred by them in order to effect the donation concerned, as defined in section 60(4) of the NHA.⁴⁷⁹ Accordingly, no payment may be made to a person donating HBMs, except to reimburse the donor for reasonable expenses relating to the donation. This prohibition, as was argued earlier in this thesis, does not preclude benefit sharing by participants in health research, as it is strictly limited to the donation of the HBMs only.

All of the regulations discussed in this chapter demonstrate that South African legislation is lacking on issues regarding benefit sharing. Even though the regulations adequately deal with the issue of informed consent for the removal of HBMs from living and deceased persons for purposes of research and or study, the regulations are silent on the issue of consent relating to any future use of the extracted HBMs. ⁴⁸⁰ Unlike other African countries such as Kenya and Nigeria, ⁴⁸¹ the South African National Health Act and its regulations do not clarify the issue of informed consent for the future use of extracted HBMs after these have been procured for research purposes. This omission is a further issue that complicates the determination of who has ownership interests as regards the extracted HBMs. ⁴⁸² Despite the omission in these regulations, the Department of Health's Ethics in Health Research Guidelines indeed attempt to regulate the secondary use of HBMs, as will be discussed below.

3.3.2.2 Regulations Relating to Artificial Fertilisation of Persons

The Regulations relating to the Artificial Fertilisation of Persons came into effect in 2016.⁴⁸³ These regulations are the only set of regulations that specifically address the issue of the

⁴⁷⁹GN R177 in Government Gazette 35099 of 2 March 2012.

⁴⁸⁰Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁴⁸¹Nienaber A "Consent to and authorisation of the export and use of human biological specimens for future research–perspectives from three African countries" 2011 *CILSA* 63:225–254.

⁴⁸²Nienaber A "Consent to and authorisation of the export and use of human biological specimens for future research–perspectives from three African countries" 2011 *CILSA* 63:225–254.

⁴⁸³Notice 1165 in Government Gazette 40312 of 30 September 2015.

ownership of human biological material in the context of gametes in fertilisation. In this regard, regulation 18⁴⁸⁴ provides as follows:

- (1) Before artificial fertilisation, the ownership of a gamete donated for the purpose of artificial fertilisation is vested -
 - (a) in the case of a male gamete donor but -
- (i) before receipt of such gamete by the authorised institution to effect artificial fertilisation by the authorised institution which removed or withdrew the gamete; and (ii) after receipt of such gamete by the authorised institution that intends to effect artificial fertilisation, in that institution;
- (b) in the case of a male gamete donor for the artificial fertilisation of his spouse, in that male gamete donor; and
- (c) in the case of a female gamete donor, for the artificial fertilisation of a recipient, in that female gamete donor.
- (2) After artificial fertilisation, the ownership of a zygote or embryo effected by donation of male and female gametes is vested -
 - (a) in the case of a male gamete donor, in the recipient; and
 - (b) in the case of a female donor, in the recipient.

Gametes are included in the definition of HBMs in the Regulations relating to the Use of Human Biological Materials, as referred to above. However, unlike the aforementioned regulations, the Regulations relating to Artificial Fertilisation expressly provide for ownership of the gametes and the product of the gametes. The above provision is the closest acknowledgement in South African law of some form of proprietary rights in a specific category of HBMs.⁴⁸⁵ Mahomed⁴⁸⁶ points out that it appears that whoever is in possession of the human biological material used in the process of artificial fertilisation would be regarded as the owner of the said material.

3.3.2.3 Regulations relating to Tissue Banks

These Regulations were enacted in 2012 in terms of section 68 of the National Health Act.⁴⁸⁷ The issue of ownership does not appear in these Regulations since regulation 1 emphatically states that no person shall remove, acquire or import human tissue from any living person and use said tissue or its products for therapeutic, research or educational

⁴⁸⁴Regulation 18 of the Regulations Relating to the Artificial Fertilisation of Persons Notice 175 in Government Gazette 35099 of 2 March 2012.

⁴⁸⁵It is suggested that due to discrepancies in the definition of an embryo in South African law, there is no guidance on whether an embryo may fulfil the requirements to be classified as property.

⁴⁸⁶Mahomed S "*An ethico-legal framework for the regulation of biobanks in South Africa*" (Philosophy PhD University of the Witwatersrand Johannesburg 2018) Available at: https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 26 May 2023)

⁴⁸⁷Notice 182 in Government Gazette 35099 of 2 March 2012.

purpose unless he or she is granted permission by the Department.⁴⁸⁸ The Director-General of the Department of Health may, on the application, authorise the use, import and export of tissue by an applicant (who will become an authorised tissue bank).⁴⁸⁹

The origins of the tissue are "donors" in the Regulations, which define a donor as "a person from who[m] tissue, blood, blood products or stem cells is donated". ⁴⁹⁰ According to the Regulations, the only obligations to the donor of the tissue—as stated in Regulation 6—is for the recipient of their tissue to register their particulars and the identity and relationship of the consenting person, including their name, address and telephone number to satisfy the tissue bank's or organisations' reporting obligations. ⁴⁹¹ Section 15 (iv) of the Regulations calls for the anonymity and privacy of the donors. ⁴⁹²

3.3.2.4 Regulations relating to Stem Cell Banks

These regulations were also enacted in 2012 under section 68 of the National Health Act.⁴⁹³ In the Regulations relating to stem cells,⁴⁹⁴ stem cell banks are obligated to the donors to preserve and protect the donors' right to privacy and anonymity.⁴⁹⁵ Informed consent is also a prerequisite for donation.⁴⁹⁶ Regulation 3 states that no person shall use stem cells or its products for therapeutic, research or educational purposes unless they are authorised by the Health Department of Health via the Director General.⁴⁹⁷ Regulation (5)(2)(b) requires that stem cell banks provide a health officer for the area in which the stem cell donation was supplied.⁴⁹⁸ Regulation 7⁴⁹⁹ outlines the additional duties of the health officer:

- (1) A health officer may, as far as stem cells or any matter relating thereto is considered -
- (a) take samples, or direct that such samples be forwarded or delivered to whom so ever or wherever she or he deems fit, in such

⁴⁸⁸Regulation 1 of the Regulations Relating to Tissue Banks Notice 182 in Government Gazette 35099.

⁴⁸⁹Regulation 3(3)(c) 1 of the Regulations Relating to Tissue Banks Notice 182 in Government Gazette 35099.

⁴⁹⁰Notice 182 in Government Gazette 35099 of 2 March 2012.

⁴⁹¹Regulation 6 of the Regulations Relating to Tissue Banks Notice 182 in Government Gazette.

⁴⁹²Regulation 15(iv) of the Regulations Relating to Tissue Banks Notice 182 in Government Gazette.

⁴⁹³Notice 183 in Government Gazette 35099 of 2 March 2012.

⁴⁹⁴Stem cells are defined as cells that have both the capacity to self-generate as well as to differentiate into mature specialised cells.

⁴⁹⁵Regulation 10 of the Regulations Relating to Stem Cell Banks Notice 183 in Government Gazette 35099. ⁴⁹⁶Regulation 5 (1)(a)(viii) of the Regulations Relating to Stem Cell Banks Notice 183 in Government Gazette

⁴⁹⁷Regulation 3 of the Regulations Relating to Stem Cell Banks Notice 183 in Government Gazette 35099.

⁴⁹⁸Regulation (5)(2)(b) of the Regulations Relating to Stem Cell Banks Notice 183 in Government Gazette 35099.

⁴⁹⁹Regulation 7 of the Regulations Relating to Stem Cell Banks Notice 183 in Government Gazette 35099.

quantities as she or he may consider necessary and adequate for testing purposes, of tissue or any tissue product or of any device or test reagent or other material used in the testing or preparation of such tissue or tissue product;

- (b) mark or seal any container with stem cell[s] or any device, test reagent or substance;
- (c) request information or registers from the management of the authorised stem cell bank and interrogate any member of the staff of the authorised stem cell bank in connection with -
- i) any premises, equipment or methods used or being used by the authorised stem cell bank;
- ii) any tissue or tissue product or any test reagent or substance referred to in these regulations; or
- iii) any applicable standards operating procedures;
- (d) place under embargo or seize any stem cells; or
- (e) documentation if in her or his opinion it may produce evidence of an offence in terms of the Act and these regulations.
- (2) a health officer shall exhibit the written authority by virtue of which she or he was authorised, to any person affected by the exercise or performance, of any power, duty or function under the Act, when called upon to do so.

This specific Regulation effectively makes the health officer the custodian of the stem cells.

3.3.2.5 Regulations Regarding the General Control of Human Bodies, Tissue, Blood, Blood Products and Gametes

Regulation 180 was enacted in March 2012 in terms of section 68 of the National Health Act. For Regulation 2 requires written consent from the living person from whom tissue, blood or gametes are removed. The tissue, blood and gametes removed from a living person are considered donations to be used for medical and dental purposes or the artificial fertilisation of another person. For a section 68 of the National Health Act

Regulation 26⁵⁰² states that exclusive rights to the bodies of deceased persons, tissue, blood and gametes are acquired by any person who obtains the body of the deceased person or the HBMs, subject to informed consent and the provisions and restrictions of the NHA or any other law, on condition that the body, tissue, blood or gametes are used for the purposes for which they have been donated. This Regulation implies that exclusive

⁵⁰⁰Notice No. R. 180 in Government Gazette 35099 of 2 March 2012.

⁵⁰¹Regulation 3 of the Regulations Regarding the General Control of Human Bodies, Tissue, Blood, Blood Products and Gametes in Government Gazette 35099 of 2 March 2012.

⁵⁰²Regulation 26 of the Regulations Regarding the General Control of Human Bodies, Tissue, Blood, Blood Products and Gametes in Government Gazette 35099 of 2 March 2012.

property rights are transferred from donor to the donee of HBMs, where a donee could be a researcher, doctor or research institute.⁵⁰³ However, despite receiving 'exclusive rights', limitations on the donation of the HBMs are found in the requirement that these may only be donated or used for the specific purposes defined in the NHA.

The notion that coded and anonymised HBMs samples (inclusive of genetic material) ceases to be identified with the donor once donated is troublesome because the genetic information within those materials is susceptible to identification.⁵⁰⁴

3.3.2.6 National Material Transfer Agreement for Human Biological Materials

In 2018, the government of South Africa gazetted a Material Transfer Agreement (MTA) template for the transfer of HBMs.⁵⁰⁵ The MTA is a guidance template document to be used by all providers and recipients of biological material for use in research or clinical trials, under the oversight of Health Research Ethics Committees.⁵⁰⁶

Section 3.3 of the MTA states that the provider⁵⁰⁷ of the HBMs remains the custodian of the materials and that the donor remains the owner of the material until such materials are destroyed.⁵⁰⁸ The significance is that this clause introduces the concept of a custodianship between the donor and the provider of the HBMs. Custodianship means the provider has a caretaking obligation to the HBMs from the initial collection of the HBMs to the publication of the research results, which is different from the legal understanding of ownership as it does not involve exclusive rights of and complete control over the HBMs in relation to proprietary rights, as would be in the case of ownership.⁵⁰⁹ Section 4 of the MTA lists the provider's obligations, one being the obligation to obtain informed consent from the donor of the HBMs, where reasonably possible, as well as approval from the HREC for any further

⁵⁰³National Health Act No. 63. Government Gazette No. 350099 2012.

⁵⁰⁴Mahesh KP "Laws and regulations associated with ownership of human biological materials in South Africa" 2015 *S Afr J BL* 8(1):11–18.

⁵⁰⁵Proclamation No 11 Government Gazette 41781 of 20 July 2018.

⁵⁰⁶Proclamation No 11 Government Gazette 41781 of 20 July 2018.

⁵⁰⁷Provider being the providing institution of the HBM.

⁵⁰⁸Section 3.3 of the MTA 2018.

⁵⁰⁹Yassin R, Lockhart N et al "Custodianship as an ethical framework for

biospecimen-based research" 2010 Cancer Epidemiology, Biomarkers and Prevention: A Publication of the American Association for Cancer Research, Cosponsored by the American Society of Preventive Oncology 19(4):1012–1015.

uses of the material. ⁵¹⁰ Section 10 of the MTA⁵¹¹ reinforces the onus on the custodian to obtain informed consent from the owner of the HBMs, which provides that "the provider must submit the informed consent form for Secondary Uses of the Material to the HREC should the need arise for Secondary Use". This section is also significant as neither the NHA nor the Regulations relating to the Use of Human Biological Materials addresses the issue of the secondary use of HBMs.

The MTA addresses the issue of benefit sharing in section 7.1⁵¹² by requiring that "the sharing of benefits should be discussed and negotiated between the Provider and Recipient⁵¹³ before materials are transferred to the Recipient". The MTA does not mandate that benefit sharing must occur but does provide an Annexure B where the terms of a benefit sharing arrangement agreed upon by the Provider and Recipient institutes may be recorded. The MTA is the first legal document to define benefit and benefit sharing in the context of health research.⁵¹⁴

I propose that the benefit sharing model that provides the best 'fit' for the benefit of all participants in health research is the Charitable Trust model, proposed by Winickoff and Winikoff. ⁵¹⁵ A "trust" involves a fiduciary relationship in which trustees hold title to property but are obligated to keep or use the said property for the benefit of the beneficiary. ⁵¹⁶ This model suggests that if hospitals solicit altruistic donations of HBMs, the hospitals should act as *trustees* of the HBM donations rather than brokers. This model creates a fiduciary relationship where the trustee has legal fiduciary duties over the HBMs, but must keep or use the property for the benefit of the specified party, which in the case of the HBMs, is the beneficiary/donor. ⁵¹⁷

-404

⁵¹⁰Section 4 of the MTA 2018.

⁵¹¹Section 10 of the MTA 2018.

⁵¹²Section 7.1 of the MTA 2018.

⁵¹³Recipient being the recipient institution.

⁵¹⁴In the MTA, a benefit is described as "amongst others, the sharing of information; use; of research results; royalties; acknowledgement of the Provider as the source of the Materials; publication rights; transfer of technology or the process materials; and capacity building", while benefit sharing is described as "the process or act of sharing in the benefits that derive from the Project in a manner that is fair and equitable."

⁵¹⁵Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12):1180–1184.

⁵¹⁶Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

⁵¹⁷Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12):1180–1184.

The gazetted MTA received some criticism from scholars whom, whilst recognising the need to respect local culture when promoting research collaborations between South Africa and international organisations and the need for fair benefit sharing, suggest that the MTA has failed to align with the broader legal environment in clarity and practicality.⁵¹⁸

Despite the criticism, the MTA is a first step in the right direction.⁵¹⁹ South African institutions are currently involved in multi-national research with both LMICs' and HICs' institutions.⁵²⁰ Considering the historical injustices brought about by lack of integrity, cultural insensitivities and unfair collaborations,⁵²¹ and the fact that previous MTAs that have been used in SA often lacked ethical safeguards and failed to address specific South African problems,⁵²² a standardised MTA gazetted by the National Department of Health to serve as a framework for the transfer of HBMs and related data, is a critical necessity.

3.3.3 Protection of Personal Information Act 4 of 2013

POPIA⁵²³ was enacted on 1 July 2020 and took full effect on 1 July 2021.⁵²⁴ The Act is set to give effect to section 14 of the right to privacy in the Constitution.⁵²⁵ The Act has far-reaching implications for all research activities that involve the collection, processing and storage of personal information. POPIA is designed to work like other data protection regulations such as the European Union's General Data Protection Regulation (GDPR).⁵²⁶ Section 2(a) of the Act states that the purpose of the Act is to:⁵²⁷

⁵¹⁸Thaldar DW, Botes M and Nienaber A "South Africa's new standard material transfer agreement: proposals for improvement and pointers for implementation" 2020 *BMC Medical Ethics* 21(1):1–13.

⁵¹⁹Labuschaigne M, Dhai A *et al* "Protecting participants in health research: the South African Material transfer agreement" 2019 *S Afr Med J* 109(5):353–356.

⁵²⁰Thaldar DW, Botes M and Nienaber A "South Africa's new standard material transfer agreement: proposals for improvement and pointers for implementation" 2020 *BMC Medical Ethics* 21(1):1–13.

⁵²¹Moodley K and Singh S "It's all about trust": reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

⁵²²Mahomed S, Nahrens K, Slabbert M and Sanne I "Managing human tissue transfer across

national boundaries—an approach from an institution in South Africa" 2016 Dev World Bioeth 16:39-35.

⁵²³POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 22 May 2023).

⁵²⁴Mahomed S and Staunton C "Ethico-legal analysis of international sample and data sharing for genomic research during COVID-19: a South African perspective" 2021 *BioLaw* 1 Journal available at www.biodiritto.org ISNN2284–4503 (Date of use: 23 May 2023)

⁵²⁵Section 14 of The Constitution.

⁵²⁶Mahomed S and Staunton C "Ethico-legal analysis of international sample and data sharing for genomic research during COVID-19: A South African perspective" 2021 *BioLaw* 1 Journal available at www.biodiritto.org ISNN2284–4503 (Date of use: 23 May 2023).

⁵²⁷POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 23 May 2023).

- (a) give effect to the constitutional right to privacy, by safeguarding personal information when processed by a responsible party, subject to justifiable limitations that are aimed at:
 - (i) balancing the right to privacy against other rights, particularly the right of access to information; and
 - (i) protecting important interests, including the free flow of information within the Republic and across international borders.⁵²⁸

Physical HBM samples do not fall into the category of personal information, which is defined in POPIA as "including information relating to identifiable, living, natural person". However, the data derived from the HBMs falls within the ambit of personal information in POPIA. Biobanks and/or tissue banks which store and distribute HBMs and often associated data for health research in perpetuity, are now legally obliged to adhere to the POPIA.

Section 26(a) of the POPIA prohibits a responsible party from processing personal information concerning the ethnic origin, health or biometric information of a data subject⁵³⁰ unless, as stated in section 27(1), the data subject consents, and/or the processing of the personal information of the data subject is for research purposes that serve the public interest and consent cannot be obtained.⁵³¹ Section 27(1) additionally provides that guarantees have to be in place to ensure that the data processing does not adversely affect the individual privacy of the subject to a disproportionate extent.⁵³²

Section 29 (b) of the POPIA states that processing personal information concerning a data subject's race or ethnic origin is permitted if the processing complies with laws and other measures designed to protect or advance persons or categories of persons who have been disadvantaged by unfair discrimination.⁵³³

Section 32(5) of the Act stipulates that personal information concerning inherited characteristics (such as genetic or genomic information) may not be processed in respect

⁵²⁸ Section 2 (a) of the POPIA Act No 4 of 2013.

⁵²⁹POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 23 May 2023).

 ⁵³⁰ For health research purposes, a data subject under the POPIA is referred to as a research participant.
 531 Section 26 and Section 27 of the POPIA Act https://www.gov.za/sites/default/files/gcis document/201409/3706726-

¹¹act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 23 May 2023).

⁵³²Section 27 (1) of the POPIA Act https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 23 May 2023).

⁵³³Section 29 of the POPIA Act https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf_(Date of use: 23 May 2023).

of the data subject unless a serious medical interest prevails or the data processing is necessary for historical, statistical or research activity.⁵³⁴

Thus, although the POPIA places certain restrictions on the processing of personal information which extends to data derived from HBMs, certain exceptions exist regarding the processing of personal information for research purposes.⁵³⁵

3.3.4 Intellectual Property Rights from Publicly Financed Research and Development Act (IPR Act)

The IPR Act promulgated in 2008 only came into effect in 2010.⁵³⁶ The Act is based on the Bayh–Dole Act (the United States Patent and Trademark Laws of 1980).⁵³⁷ The IPR Act extends only to Research and Development (R&D) that relies on public funds. The IPR Act states in section 2(1) that its aim is to⁵³⁸:

Make provision that intellectual property emanating from publicly financed research and development is identified, protected, utilised and commercialised for the benefit of the people of the Republic, whether it be for social, economic, military or any other benefit.

The IPR Act does not define what constitutes research and development (R&D). The *Interpretation Note 11: State-Owned Enterprises and the Interface with the IPR Act*⁵³⁹ of the National Intellectual Property Management Office (NIPMO) explains that because of the lack of a definition of R&D in the IPR Act, NIPMO decided to adopt the definition by the *Frascati Manual* (2015) of the Organisation for Economic Co-operation and Development (OECD).⁵⁴⁰ In this definition, R&D include basic research, applied research and experimental development, as well as specialised healthcare research that is conducted in university hospitals.⁵⁴¹ Research using human biological materials is also conducted in government hospitals, university hospitals, private hospitals and private doctors rooms.

⁵³⁴Section 32(5) of the POPIA Act https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf_(Date of use: 23 May 2023).

⁵³⁵Sections 14(2); 15(3); 18(4); 27(1) 32(5); 35(1) of POPIA.

⁵³⁶The IPR Act https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 23 May 2023).

⁵³⁷The IPR Act https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 23 May 2023).

⁵³⁸Section 2(1) of the IPR Act No 51.

⁵³⁹The IPR Act 51 https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 23 May 2023).

⁵⁴⁰Section 5 of the IPR Act 51 https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 23 May 2023).

⁵⁴¹Definition of Research and Development as per *Frascati Manual* of the OECD https://www.oecd-ilibrary.org/docserver/9789264199040-en.pdf (Date of use: 23 May 2023).

Along with the collected samples that are to be processed and analysed, the sample donors' relevant medical data and social information are also collected. All of these activities fit into the definition of R&D.

In the IPR Act, a recipient of public funds is defined as any person, juristic or non-juristic that undertakes R&D using funding from a funding agency (the state or an organ of the state) and includes an institution.⁵⁴² Section 4 of the IPR Act provides that intellectual property resulting from publicly funded research and development shall belong to the recipient of the public funds.⁵⁴³ If the recipient of the funds does not wish to retain ownership, NIPMO must be informed of the decision and reasons for such a decision. Section 4(4) elaborates that if NIPMO does not have an intention to acquire ownership of the intellectual property, the recipient must be notified of this decision, who must then provide the intellectual property creator⁵⁴⁴ with an option to acquire ownership of and patent the intellectual property, provided that the R&D was entirely financed by public funds.⁵⁴⁵

Section 2(a) of the Act tasks the recipient of the public funds to assess, record and report on the benefit of publicly funded research and development for society. The South African government recognises that there should be benefit sharing between society and the recipient using public funds for R&D. The recipient is also obliged in accordance with section 2(c) of the IPR Act to identify commercialisation opportunities for intellectual property that would emanate from the publicly funded research.

Section 5(1)(a) of the Act reiterates that a recipient must:

Put in place mechanisms for the identification, protection, development, management of intellectual property, intellectual property transactions and, where applicable, the commercialisation of intellectual property and appropriate capacity-building relating thereto.⁵⁴⁶

If the recipient of the funds is an institution, section 5(1)(f) furthermore provides that it should:

⁵⁴²IPR Act 51 of 2008.

⁵⁴³Section 4(1) 0f the IPR Act 2008.

⁵⁴⁴An "intellectual property creator" is defined as the person involved in the conception of intellectual property in terms of this Act and identifiable as such to obtain statutory protection and enforcement of intellectual property rights, where applicable in the IPR Act 51 of 2008.

⁵⁴⁵Section 4(4) (b) (a) of the IPR Act 51 of 2008.

⁵⁴⁶Section 5(1) (a) of the IPR Act of 2008.

Manage revenues due to it from intellectual property transactions and the commercialisation thereof, including managing the benefit sharing arrangements with intellectual property creators at the institution.

Section 5 stresses the issue of benefit sharing arrangements between recipients of public funds and intellectual property creators. Although the focus of this Act is not the regulation of HBMs *per se*, the question arises on the position of the donors of HBMs whose contribution of their materials enabled the R&D to occur.

In section 5(10)(j), the Act requires institutions to set up mechanisms to annually assess, record and report to NIPMO on the benefits of publicly financed research conducted in that institution for society.

The IPR Act acknowledges that recipients of public funds can co-own intellectual property resulting from R&D and undertaken with public funds with a private entity or organisation, as stipulated in section 15(2) of the IPR Act.⁵⁴⁷ The Act defines a private entity or organisation as "a private sector company, a public entity, an international research organisation, an educational institution or an international funding or donor organisation".⁵⁴⁸ However, this co-ownership requires that stringent conditions between the parties are met, which include among others, benefit sharing agreements. The requirements listed in section 15(2) of the Act include, that (1) a contribution of resources should have occurred (such as background intellectual property brought in by the private entity); (2) the parties involved jointly contributed to the creation of the intellectual property; (3) a benefit sharing arrangement with the intellectual property is agreed upon by the parties.⁵⁴⁹

The default sole ownership by the recipient is then voided when all these requirements have been met and co-ownership of the intellectual property is possible. In the final instance, section 15(4) stipulates that any R&D that occurs at an institution and which is fully funded by a private entity or an organisation will not be seen as publicly funded R&D and would thus be exempt from the application of the IPR Act. 550

⁵⁴⁷Section 15(2) of the IPR Act 51 of 2008.

⁵⁴⁸Section 15(5) of the IPR Act 51 of 2008.

⁵⁴⁹Section 15(2) of the IPR Act 51 of 2008.

⁵⁵⁰Section 15(4) of the IPR Act 51 of 2008.

In the era of biotechnology, research collaborations between institutions in LMICs and HICs have drastically increased. Mahomed points out that the exchange of human biological materials and associated data between institutions, both within and across countries, has become a prominent feature of biomedical research and biobanking. Section 12 of the IPR Act addresses issues pertaining to offshore intellectual property transactions, stating that a recipient has a duty to inform NIPMO of its intention to conclude an intellectual property transaction offshore, bearing in mind that the Republic of South Africa will benefit from this transaction. The recipient must also satisfy NIPMO that the Republic lacks sufficient capacity to develop and commercialise the intellectual property. S53

Intellectual property creators feature very prominently in the IPR Act, as is evident from the wording of section 2(2)(d), which states that human ingenuity and creativity must be recognised and rewarded. Section 10 of the IPR Act elaborates on the benefit sharing rights of intellectual property creators in institutions but remains silent on any benefit sharing arrangement with donors of HBMs.⁵⁵⁴ Section 10(1) specifically provides:

Intellectual property creators at an institution and their heirs are granted a specific right to a portion of the revenues that accrue to the institution from their intellectual property in terms of this Act until such right expires.⁵⁵⁵

Section 10(a)(b) next details how benefits to intellectual property creators and their heirs should be calculated, namely at least 20 per cent of the revenue accruing to the institution emanating from said intellectual property for the first one million rands of revenue, or any such higher amount as the Minister may prescribe and thereafter, at least 30 per cent of the net revenue accruing to the institution. The IPR Act further provides regarding benefit sharing, that intellectual property creators must be the first call for the applicable revenue, after which the recipient can allocate the balance of the revenue as it deems fit, which should include funding for more research and development.⁵⁵⁶ This arrangement once

⁵⁵¹H3frica Consortium, Rotimi C, Abayomi A *et al* "Research capacity. Enabling the genomic revolution in Africa" 2014 *Science* 344 (6190):1346–1348.

⁵⁵²Mahomed S and Labuschaigne M "The role of research ethics committees in South Africa where human biological materials are transferred between institutions" 2019 *S Afr J Bioethics Law* 12(2):79–83.

⁵⁵³Section 12(2) IPR Act 51 of 2008.

⁵⁵⁴Section 10 of IPR Act 51.

⁵⁵⁵ Section 10 of IPR Act 51.

⁵⁵⁶Section 10 of the IPR Act 51 of 2008.

again alludes to donors of human biological material in health research, especially in LMICs. The question rightly arising is why the donors of HBMs and their communities are not entitled to any benefits accruing from their donations in terms of the IPR Act.

3.3.5 Patent Act 57 of 1978

The age of biotechnology precipitated many biotechnology and pharmaceutical companies aggressively pursuing intellectual property rights to biological materials to protect their proprietary interests and gain profits from their research and development costs. In most research relating to human biological materials, there is a need to isolate, purify, concentrate, clone and even modify the HBMs; consequently, the issue of private companies, universities and government agencies patenting HBMs is not an uncommon endeavour. HBMs and indeed, the engineered products that arise from research and development on HBMs, are highly contested issues in healthcare research, especially if their development is vital for saving lives or curing a disease. Ownerships in HBMs would confer exclusive rights to the donor of the HBMs; however, as discussed earlier, this notion would contradict respect for human dignity and the inviolability of human beings.

A patent is a right granted by the government to exclude others from using, making or commercialising an invention for a limited period. Fatents protect intellectual property rights. In South Africa, the Patent Act provides that a patent may be granted for any new invention that involves an inventive step and which is capable of being used or applied in trade, industry or agriculture. Section 27(1) of the Patent Act stipulates that a patent application may be made by the inventor of the creation or by any other person who acquires the right to apply from the inventor or by both such inventor and such other

⁵⁵⁷Resnick DB "The human genome: common resource but not common heritage" 2005 *Frontis* 197–210.

⁵⁵⁸Resnick DB "The human genome: common resource but not common heritage" 2005 *Frontis* 197–210.

⁵⁵⁹Mahomed S 2018 *An Ethico-legal framework for the regulation of biobanks in South Africa* (Doctor of Philosophy PhD University of the Witwatersrand) Available at: http://wiredspace.wits.ac.za/handle/10539/25331.

⁵⁶⁰Resnick DB "The human genome: common resource but not common heritage" 2005 Frontis 197–210.
⁵⁶¹Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 South African Journal of Science 107(11/12):24–33.

⁵⁶²Section 25(1) of the Patent Act 57 of 1978.

person.⁵⁶³ Section 27(2) explains that in the absence of the agreement referred to in section 27(1), joint inventors may apply for a patent in equal, undivided shares.⁵⁶⁴

The Companies and Intellectual Property Commission (CICP), previously known as the Companies and Intellectual Property Registry Office (CIPRO), administers all parts of legislation related to intellectual property regulation, such as the Patent Act, and is also responsible for the registration of all intellectual property rights such as patents in South Africa. 565 Pouris and Pouris 566 rightly argue that the patent system appears to benefit the pharmaceutical and biotechnology industries in South Africa in terms of research, development and innovation. It is troubling to learn that of the 280 patents applied for by South African universities and researchers to CIPRO from 1996-2006, only 58 were protected abroad. Although some inventions require only local protection, exposing inventions abroad without protection could easily result in copyright infringements of the innovation abroad. It is worth noting that the Intellectual Property Rights for Publicly Financed Research Development Act was enacted to provide for more effective utilisation of IP emanating from publicly financed research and development. 567 However, whilst the IPR Act establishes the National Intellectual Property Management Office and the Intellectual Property Fund, it does not address issues related to CIPRO's activities, ⁵⁶⁸ meaning that researchers and their inventions could go unprotected, wasting government funds in the process.

In 2005, section 2 of the Patents Act was amended⁵⁶⁹ to insert certain definitions and to require the applicant of a patent to furnish information relating to any role played by an indigenous biological or genetic resource or traditional knowledge.

⁻

⁵⁶³Section 27(1) of the Patent Act 57 of 1978.

⁵⁶⁴Section 27(2) of the Patent Act 57 of 1978.

⁵⁶⁵Responsibilities of the Companies and Intellectual Property Commission http://www.cipc.co.za (Date of use: 23 May 2023).

⁵⁶⁶Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 *South African Journal of Science* 107(11/12): 24–33.

⁵⁶⁷The IPR Act 51 of 2008.

⁵⁶⁸Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 *South African Journal of Science* 107(11/12): 24–33.

⁵⁶⁹Patents Amendment act 2005 https://wipolex.wipo.int/en/text/179614 (Date of use: 23 May).

In human health research, proprietary rights are often granted to the intellectual property developer rather than the owner or donor of the HBMs.⁵⁷⁰ Although the concept of benefit sharing appears in the IPR Act, this arrangement is limited to apply between an institution and the intellectual property creators and their heirs.⁵⁷¹ Both the IPR Act and the Patent Act fail to provide a framework on how benefit sharing arrangements should be construed with the donors in research, as well as how development by the intellectual property creator should occur. In the current context where the number of human tissue banks and biobanks are rising sharply and have become integral to research facilities around the globe, including in South Africa,⁵⁷² the legislation does not provide for a framework for benefit sharing between research participants, investigators and investors.

3.3.6 Department of Health Ethics in Health Research Guidelines, 2015

The concept of the commodification of human biological materials is not directly addressed in the Department of Health's Ethics in Health Research Guidelines. However, section 2.1 of the Guidelines advises that all health research should consider distributive justice. This section states that the risks and benefits for all role players involved in research should be fairly balanced and that the population should not be denied any benefits resulting from the research. Section 2.1 explicitly states that the donors of human biological materials—who are also the participants in human health research—and their communities should have a reasonable likelihood that they will all benefit from the research, if not immediately, then in the future.

Section 3.3.7 of the Guidelines⁵⁷⁵ addresses the secondary use of biological materials that have originally been collected for other purposes, such as diagnostics or therapeutic purposes. Researchers often bank surplus samples in tissue banks and biobanks, leading to the dilemma of having to seek new informed consent from the donor of human biological materials for further and secondary uses of the HBMs again. Section 3.3.7 recommends

⁵⁷⁰See IPR Act 51 of 2008 and Patent Act 57 of 1978.

⁵⁷¹See Section 5 and Section 10 of the IPR Act 51 of 2008.

⁵⁷²Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediments to accessing healthcare" 2019 *SAJBL* 12(1): 27–31.

⁵⁷³DoH Ethics in Health Research Guideline https://www.health.gov.za/up-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

⁵⁷⁴Section 2.1 of the DoH 2015 Ethics in Health Research.

⁵⁷⁵Section 3.3.7 of the DoH 2015 Ethics in Health Research.

how to proceed with the secondary use of HBMs and data in the absence of broad consent from the donor. This requirement seemingly acknowledges that a person retains some rights to their HBMs and data.⁵⁷⁶ Mahomed⁵⁷⁷ points out that although the legal framework is unclear on the idea of the ownership of HBMs and intellectual property that result from the use of said HBMs and its data, a benefit sharing agreement incorporated into the MTA, coupled with the model of custodianship of HBMs would promote health research and build trust amongst the relevant stakeholders involved in a research project.

It is evident that in South African law, there is a general silence on the classification of HBMs, yet there is an unending and complex debate of which one contentious point is whether there should be a property or non-property approach towards HBMs.⁵⁷⁸ The law also does not address the different models, such as custodianship, stewardship and trusteeship of HBMs.

The Constitution of South Africa in section 39(1) directs that when interpreting the Bill of Rights, a court tribunal or forum must consider international law and may consider foreign law.⁵⁷⁹ For this reason, foreign judgements become relevant when human rights are interpreted. The next section will turn to some approaches by foreign courts when adjudicating issues around the legal ownership of HBMs and how benefit sharing—if any was to occur with the donor of the HBMs—is addressed.

3.4 Foreign case law

In an ideal society, health research should be altruistic and for the good of humanity, a society with free and adequate healthcare, as in some affluent nations.⁵⁸⁰ The case of

⁵⁷⁶Section 3.3.7 of the DoH 2015 Ethics in Health Research.

⁵⁷⁷Mahomed S 2018 "An ethico-legal framework for the regulation of biobanks in South Africa" (PhD in Bioethics and Health Law University of the Witwatersrand 2018) Available at https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 27 May 2023).

⁵⁷⁸Mahomed S, Nöthling-Slabbert M and Pepper MS "Ownership and human tissue-the legal conundrum: a response to Jordaan's critique" 2017 *SAMJ* 107(3): 196–198.

⁵⁷⁹The Constitution of South Africa https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 23 May 2023).

⁵⁸⁰ Lucas CJ, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 95–127.

Henrietta Lacks, who was the first source of the immortal cell line (HeLa),⁵⁸¹ makes it clear that health research is often conducted at the expense of others. The HeLa tissue cells that were used in the initial culture of the cell line were obtained from Ms Henrietta Lacks in 1951 at John Hopkins Hospital by physicians who were treating her for cervical cancer. The cell line was removed without her knowledge and consent soon after she had passed away.⁵⁸² The HeLa cells, still viable to date, became invaluable to medical research. It could be argued that the inventors of the cell line acted altruistically by readily distributing the cells across the research community while the cell line and its discoveries became extremely lucrative. However, all of this happened while the Lacks family experienced poverty and had no access to healthcare.⁵⁸³ This outcome is patently unfair.

Another illustrative case is that of *Moore v. Regents of the University of California*.⁵⁸⁴ In this case, Mr John Moore, a patient with hairy cell leukaemia, had a splenectomy in 1976 as part of his treatment at the University of California Los Angeles (UCLA) Medical Centre.⁵⁸⁵ The attending physician, Dr David Golde and his research assistant removed samples of Mr Moore's blood, bone marrow aspirate, sperm, skin, other tissues and fluid over a period of several years.⁵⁸⁶ The physician and assistant subsequently conducted research on the samples donated by Moore, without informing Mr Moore that his cells had great monetary potential.⁵⁸⁷ In 1979, Dr Golde established a patented cell line from Mr Moore's tissue.⁵⁸⁸ The patent was assigned to the Regents of the University of California, as they had invested in the commercial development of the cell line and future products. Dr Golde became a paid consultant and received stock in the company that acquired the developmental rights for the cell line. When Mr Moore discovered this, he sued Dr Golde,

_

⁵⁸¹Troug RD, Kesselheim AS and Joffe S "Paying patients for their tissue: the legacy of Henrietta Lacks" 2012 *Science* 337:37–38.

⁵⁸²Beskow LM "Lessons from HeLa cells: the ethics and policy of biospecimens" 2016 *Annu. Rev. Genom. Hum.Genet* 17:395–417.

⁵⁸³Beskow LM "Lessons from HeLa cells: the ethics and policy of biospecimens" 2016 *Annu. Rev. Genom. Hum.Genet* 17:395–417.

⁵⁸⁴Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁵⁸⁵Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 Jurimetrics J 481:167–191.

⁵⁸⁶Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 Jurimetrics J 481:167–191.

⁵⁸⁷Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 Jurimetrics J 481:167–191.

⁵⁸⁸Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

UCLA and two biotechnology companies,⁵⁸⁹ claiming that he had a right to share in the profits generated from his HBMs, which he asserted had a potential value of approximately \$3 billion.

The California Supreme Court rejected Mr Moore's property right's claim in his cells, concluding that this would hamper scientists from conducting medical research. However, the court found that Moore had a cause of action against Dr Golde for the physician's failure to obtain informed consent and for a breach of a fiduciary duty. However, in *dicta* the court observed that its decision left the issue whether the transfer of human tissue should be gift-based or market-based undecided, and that future controversies would have to be decided on a case-by-case basis. ⁵⁹⁰ As a result of the combination of the lack of case precedent, California legislation relating to the disposal of human tissue, as well as the fact that the patented cells were different from those excised from Mr Moore, Mr Moore's claim was rejected by the California Supreme Court. ⁵⁹¹

Greenberg v. Miami Children's Hospital Research Institute, Inc. is another case in the United States in which a court ruled that persons have no real property rights to HBMs that are donated for medical research.⁵⁹² The case involved families of children with Canavan disease.⁵⁹³ These families donated tissue, blood and other HBMs to Dr Matalon, a research physician (supported by the Miami Children's Hospital), whose objective was to identify the gene responsible for Canavan disease in order to develop a prenatal diagnostic test.⁵⁹⁴ Dr Matalon and his team used the donated tissue and relevant family histories and financial support from the families, successfully identified and isolated the gene causing Canavan disease in 1993.⁵⁹⁵ Subsequently, the researchers patented the gene sequence and its related applications, including the prenatal diagnostic test without informing the

-

⁵⁸⁹Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁵⁹⁰Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

⁵⁹¹Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁵⁹²Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁵⁹³Canavan disease is a gene-linked, neurological birth disorder where the white brain matter degenerates into spongy tissue.

⁵⁹⁴Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

⁵⁹⁵Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

families.⁵⁹⁶ The families sued and maintained that they would not have donated their biological material if they had known that Dr Matalon would exploit their genetics for commercial purposes and was going to restrict the use of the research results.⁵⁹⁷

The Florida Court concurred that the researchers should have provided information and obtained consent but determined that any property right in blood and tissue samples "evaporates once the sample is voluntarily given to a third party". ⁵⁹⁸ The Court also found that a research product developed from HBMs is factually and legally distinct from the original donated HBMs and, as such, becomes the property of the researcher. ⁵⁹⁹ It is worth noting that the Code of Medical Ethics of the American Medical Association, which came into effect after the research (in this case) had already begun, states that all physicians involved in research using human biological materials should:

- 1) Disclose potential commercial applications to the tissue donor before a profit is realized on products developed from biological materials.
- 2) Obtain informed consent to use biological materials in research from the tissue donor. Human biological materials and their products may not be used for commercial purposes without the consent of the tissue donor.
- 3) Share profits from the commercial use of human biological materials with the tissue donor in accordance with lawful contractual agreements.⁶⁰⁰

Although these court decisions are primarily meant to enable research to continue and prevent persons from commodifying themselves, it appears unjust that others, such as the researchers, are permitted to commodify and profit from research using another person's HBMs. It is only fair that research participants should also benefit from research using their HBMs. Moreover, research needs donors and does not only rely on the input of researchers.

An additional case in which a court had to decide on the ownership of research participants' donated HBMs was *Washington University v. Catalona*. This case involved Dr Catalona,

⁵⁹⁶Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

⁵⁹⁷Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

⁵⁹⁸Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁵⁹⁹Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

⁶⁰⁰Code of Medical Ethics Of the American Medical Association 7.3.9 https://www.ama-assn.org/delivering-care/ethics/commercial-use-human-biological-materials (Date of use: 23 may 2023).

⁶⁰¹Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

a urologist and researcher who was employed by Washington University (WU) from 1976 to 2003. While in the employment of WU, Dr Catalona and his colleagues conducted prostate cancer research. Dr Catalona habitually asked his patients to donate HBMs such as prostate tissue, blood and DNA (removed during prostate surgery) to research.⁶⁰² Through these efforts, Dr Catalona and other attending physicians at the WU built the largest biorepository for prostate cancer research in the world, regarded as a biobank strictly reserved for research purposes. Interestingly, Dr Catalona chose to refer to this collection as the "Catalona Collection", contrary to the reference used during litigation, namely the WU genito-urinary (GU) collection.⁶⁰³

The HBM donors signed several consent forms that labelled their donations as altruistic gifts for the benefit of medical research, waiving all rights to the HBM samples donated and any product that would result from research using the samples donated to the GU Biorepository. This biobank had accumulated thousands of samples and data from the donated HBMs and was operated and maintained mostly by funding and with the assistance of technical staff from WU. 605

The dispute commenced when Dr Catalona transferred a significant number of samples to a private laboratory. WU objected to this, claiming ownership of the samples, which led to a disagreement that was followed by the termination of Dr Catalona's employment at WU.⁶⁰⁶ Dr Catalona found new employment at the North-western School of Medicine, informed his patients of his decision and also asked for their permission to move their donated HBMs with him for the continuation of his prostate cancer research projects. Many of the patients consented, but WU refused to transfer the samples on the basis that they owned the samples and subsequently, sued Dr Catalona.⁶⁰⁷ The Court upheld a unanimous 2007 ruling by the Eighth US Circuit Court of Appeals, which stated that prostate tissue and serum samples donated to Washington University may continue to be

_

⁶⁰²Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁶⁰³Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

⁶⁰⁴Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁶⁰⁵Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008

Jurimetrics J 481:167–191.

⁶⁰⁶Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁶⁰⁷Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

used by the institution for cancer research.⁶⁰⁸ The Court dictated that the patients had altruistically donated the HBMs for research and no longer retained any ownership rights to the samples or the authority to transfer said samples to third parties.

The Court acknowledged that the patients were within their rights to cease participating in the research by declining to answer further questions, refusing to donate any more biological samples and refusing consent for their dated samples being used in future research.⁶⁰⁹ This ruling meant that the patients could retain some control over their donated samples and the university could not use the samples for research in perpetuity without consent from the patients.

As demonstrated by these foreign judgements, courts appear to have struggled to adapt the tradition and precedent of the law to the challenges arising from the biotechnology era. It would seem that the law recognises that (1) informed consent by research participants is a prerequisite and that information on the research and the potential commercial result using the donated samples should be a logical consequence; (2) the research participants' ownership of the donated HBMs ceases at the moment of donation and (3) the recipient of the donated HBMs has the right to commercially exploit any products that result from the research using the donated samples.⁶¹⁰

However, in other instances, courts have ruled that separated body parts can be owned by an individual. In the United Kingdom, the issue of theft by surgeons at the Royal College arose in the case of *R v. Kelley* in 1998.⁶¹¹ In this case, an artist, Mr Kelly, had obtained preserved dismembered body parts from the Royall College via a technical employee of the College. The artist used the body parts as moulds for sculptures that were exhibited in a London art gallery. To convict Mr Kelly and the technician of theft, the Courts had to recognise that the body parts were property. Mr Kelly and the technician countered the argument by stating that parts of corpses are not property and could therefore, not be stolen under the British Theft Act.⁶¹² The court ruled that the body parts were fit for

⁶⁰⁸Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 S Afr J BL 6(1):16–20.

⁶⁰⁹Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁶¹⁰Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

 ⁶¹¹Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.
 612The British Theft Act 1968.

proprietary rights as they had acquired different attributes by virtue of skill since the body parts had been preserved.⁶¹³

In another case relating to HBMs (specifically, male gametes), the Court of Appeal for England and Wales also found that proprietary rights could be assigned to sperm donors. The case was *Yearworth v North Bristol NHS Trust* ⁶¹⁴ in which said court ruled that the six men whose sperm had been stored prior to their cancer treatment indeed owned their sperm that was negligently destroyed in the storage process.⁶¹⁵

In Australia, the case of *Roche v Douglas*⁶¹⁶ which came about because of a paternity dispute where a DNA test was required, the Supreme Court of Western Australia had to rule on whether tissue samples obtained from a deceased person before his death for diagnostic purposes were susceptible to ownership. The court commented that the "world has moved on"⁶¹⁷ and ruled that tissue samples are indeed susceptible to ownership. The court explained the decision thus:

In the wider sense, it defies reason to not regard tissue samples as property. Such samples have a real physical presence. They exist and will continue to exist until some step is taken to effect the destruction. There is no purpose to be served in ignoring physical reality. To deny that the tissue samples are property, in contrast to the paraffin in which the samples are kept or the jar in which both the samples and paraffin are stored, would be in my view to create a legal friction. There is no rational or logical justification for such a result. ⁶¹⁸

These cases demonstrate that the issue of the ownership of HBMs, although very complex, merits the application of the notion of the ownership of human biological materials in special circumstances. Some courts appear willing to deal with the issue of ownership on a case-by-case basis, which may establish a precedent which would allow human biological materials to be viewed as legal property over time.⁶¹⁹

⁶¹³Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

⁶¹⁴Yearworth v North Bristol Trust [2009]3 WLR 118.

⁶¹⁵Yearworth v North Bristol Trust [2009]3 WLR 118.

⁶¹⁶Roche v Douglas 2000 WASC 146 (2002) 22WAR 331.

⁶¹⁷Roche v Douglas 2000 WASC 146 (2002) 22WAR 331 paragraph 22.

⁶¹⁸Roche v Douglas 2000 WASC 146 (2002) 22WAR 331 paragraph 24.

⁶¹⁹ Mahomed S "An ethico-legal framework for the regulation of biobanks in South Africa" (PhD University of the Witwatersrand 2018) Available at : https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 26 May 2023).

3.5 Conclusion

Biotechnology is new and exciting and has endless possibilities, whereas the law always lags behind technology and operates on tradition and precedent. The issues and problems around the ownership and proprietary rights of HBMs which have arisen from the commercialisation of research products that are the result of R&D using human biological materials, as discussed in this chapter, remain problematic and controversial. However, universally and in South Africa, the laws that relate to these important questions are currently inconsistent and unclear. It is also true that without distinct and proper definitions of specific human biological materials, e.g., human tissue, 1 it is impossible to provide sound legislation regarding any proprietary claims.

The law's uneasiness with the notion of ownership of the human body or human biological material may stem from the idea that the commodification of a person or their biological material offends notions of human dignity and the inviolability of the human body. However, the further uses of HBMs, which led to intellectual property rights and commercial gain, are not reconcilable with the requirement that a donor should donate their HBMs altruistically.

Regarding genomic research, Resnick⁶²² suggests that the human genome should be regarded as an important common resource and that everyone has a duty of stewardship and justice towards this resource, meaning we are obligated to share benefits fairly in genetic research and development, which equally applies to most HBMs. The SA MTA⁶²³ introduces the notion of the *provider* of the human biological materials as the custodian of said materials and states that before HBMs are transferred, a benefit arrangement should be agreed upon between the *provider* and the *recipient*. A custodian is not the same as an owner who, according to the SA MTA, is the donor of the HBMs. No benefit-agreement arrangement is mandated regarding the owner of the donated HBMs.

⁶²⁰Gibson SF "The Washington University V. Catalona: determining ownership of genetic samples" 2008 *Jurimetrics* 48(2):167–192.

⁶²¹Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 S Afr J BL 6(1):16–20.

⁶²²Resnik DB "The human genome: common resource but not common heritage" 2005 Frontis 197–210.

⁶²³Proclamation No 11 Government Gazette 41781 of 20 July 2018.

The classification of HBMs raises complex ethical and legal issues. In the context of private biobanks and collaboration between private and/or public medical institutions, the consent framework of the charitable trust—which this research proposes—while promoting altruism in research offers to remedy the ethical and legal challenges that the notion of proprietary rights imposes on HBMs. If biobanks act as trustees of HBMs, their fiduciary duty legally requires trustees to work for the benefit of the beneficiary of the trust where the public is the beneficiary. This would accommodate all the requirements of ethical research, altruism, good governance and benefit to the public.

In the absence of legislation for a benefit sharing model to apply in South Africa, the next chapter focuses on foreign benefit sharing models to compare best practices towards designing the most suitable and informed benefit sharing template for South Africa.

CHAPTER 4

A COMPARATIVE ANALYSIS OF ETHICO-LEGAL FRAMEWORKS FOR BENEFIT SHARING IN RESEARCH USING HUMAN BIOLOGICAL MATERIALS IN THE UNITED KINGDOM, AUSTRALIA, UGANDA AND SOUTH AFRICA

4.1 Introduction

The previous chapters canvassed the nature of benefit sharing in health research, which refers to the fair distribution of benefits and burdens arising from health research and development. Chapter 2 of this thesis established that there is no uniform definition of a benefit. A comparative analysis of existing definitions, presented in chapter 2, was found to be a constructive approach that may offer useful guidance. It is my submission that in the context of international human health research, the most appropriate definition of benefit sharing is the definition used in the framework of the access and use of genetic resources in terms of the CBD, which describes benefit sharing as:

The action of giving a portion of advantages or profits derived from the use of genetic resources or traditional knowledge to resource providers in order to achieve justice in exchange.⁶²⁵

Scholars have argued that the philosophical principle behind benefit sharing in research using HBMs is straightforward, since it is a matter of justice and in whose absence exploitation could occur.⁶²⁶ Legally, benefit sharing points to a technical term used to describe an exchange between an HBM source and those compensated for its use.⁶²⁷ The most pertinent argument against benefit sharing is that health research should be altruistic to avoid the commodification of the self.⁶²⁸

⁶²⁴Simm K "Benefit sharing frameworks-justifications for and against benefit sharing in human genetic research" 2007 www.uclan.ac.uk/genbenefit (Date of use: 23 May 2023).

⁶²⁵ Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 23 May 2023).

⁶²⁶Andanda P, Schroeder D, Chaturvedi S *et al.* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

⁶²⁷Schroeder D "Benefit sharing: it's time for a definition" 2007 J Med Ethics 33:205–209.

⁶²⁸ Lucas JC, Schroeder D, Anarson G *et al* "Donating human samples: who benefits?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 95–128.

Global injustices, however, dictate that the concept of altruistic research, as presented in developed countries propagating distributive justice that discourages the commodification of the self, cannot be translated to low- and middle-income countries (LMICs) without the emergence of serious exploitation issues. The ever-increasing health research collaborations between HICs and LMICs, and largely in historically disadvantaged communities such as those found in South Africa, have precipitated the need for global health research to provide fair benefits to all who contribute to said research to avoid exploitation.

Globally, no legally binding framework regarding benefit sharing in health research utilising HBMs exists, although some non-binding instruments refer to benefit sharing. In 2000, the Ethics Committee of the Human Genome Organisation (HUGO) in its Statement on Benefit Sharing, endorsed the concept of benefit sharing in research using HBMs by recommending that "all humanity share in, and have access to, the benefits of genetic research".

In 1997, the United Nations Educational, Scientific, and Cultural Organisation (UNESCO) issued a Universal Declaration on the Human Genome and Human Rights, which suggested that "benefits from advances in biology, genetics and medicine, concerning the human genome shall be made available to all, with due regard for the dignity and human rights of each individual".⁶³⁴

In 2005, the UNESCO Universal Declaration on Bioethics and Human Rights advocates in Article 15(1) for sharing the benefits of scientific research within the international community, emphasising specifically the need for benefit sharing with LMICs.⁶³⁵

114

_

⁶²⁹Simm K "Benefit sharing frameworks-justifications for and against benefit sharing in human genetic research" 2007 www.uclan.ac.uk/genbenefit (Date of use: 23 May 2023).

⁶³⁰Lairumbi G, Parker M, Fitzpatrick R and English M "Forms of benefit sharing in global health research undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics and Humanities in medicine* 7(1):1–8.

⁶³¹Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440. ⁶³²Schroeder D "Benefit sharing: it's time for a definition" 2007 *J Med Ethics* 33:205–209.

⁶³³Knoppers BM, Chadwick R, Takebe H *et al.* "HUGO urges genetic benefit sharing" 2002 *Community Genet* 3(2): 88–92.

⁶³⁴UN Declaration on the Human Genome and Human Rights http://portal.unesco.org/en/ev.php-URL_ID=13177&URL_DO=DO_TOPIC&URL_SECTION=201.html (Date of use: 23 May 2023).

⁶³⁵UNESCO Universal Declaration on Bioethics and Human Rights https://unesdoc.unesco.org/ark:/48223/pf0000146180 (Date of use: 23 May 2023).

Despite these globally accepted frameworks, benefit sharing remains an unresolved and divisive topic. Different national positions and varying degrees of concern on legislative, ethical and social frameworks between HICs and LMICs all contribute to the problematic nature of implementing national laws consistently and per international directives. Description of the problematic nature of implementing national laws consistently and per international directives.

This chapter seeks to compare and contrast the ethico-legal frameworks for benefit sharing in health research in South Africa with the frameworks of Uganda, the United Kingdom (UK) and Australia. The UK and Australia are used in this comparison because they are HICs and frontrunners in scientific medical research.⁶³⁸ The UK played a dominant role in the Human Genome Project⁶³⁹ and via the Wellcome Trust, together with the National Institutes of Health (NIH) in the United States and the African Society of Human Genetics, set up the Human Hereditary and Health in Africa (H3Africa) research initiative to promote scientific health research collaborations between HICs and LMICs.⁶⁴⁰ Uganda provides an example of how ethico-legal benefit sharing frameworks differ between LMICS, notably for the purpose of this chapter, between Uganda and South Africa. Furthermore, both Uganda and South Africa belong to the H3Africa consortium. Moreover, the UK, South Africa, Uganda and Australia are all jurisdictions that share a colonial past and whose legal systems were influenced, in different degrees, by the English colonial legal system which not only dictated and controlled the former legal systems in South Africa, Uganda and Australia, but also had an impact on the evolution of laws in these jurisdictions.⁶⁴¹ These jurisdictions also provided useful models for comparison in the first 2018 doctoral study in South Africa on the legal regulation of biobanking by Mahomed. 642

-

⁶³⁶Andanda P, Schroeder D, Chaturvedi S *et al.* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 33–64.

⁶³⁷Mahomed S "Human biobanking in developed and developing countries: an ethico –legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–160.

⁶³⁸Mahomed S "Human biobanking in developed and developing countries: an ethico–legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–160.

⁶³⁹Mahomed S "Human biobanking in developed and developing countries: an ethico–legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–160.

⁶⁴⁰Dauda B and Joffe S "The benefit vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170.

⁶⁴¹Hyam R "Understanding the British empire." 2010 Cambridge University Press.

⁶⁴²Mahomed S "An Ethico-legal framework for the regulation of biobanks in South Africa." 2018 Submitted in fulfilment of the degree of Doctor of Philosophy (PhD) in Bioethics and Health Law, Steve Biko Centre of Bioethics, University of the Witwatersrand, Johannesburg, South Africa. Available at:

4.2 Ethico-legal benefit sharing frameworks for health research using HBMs

4.2.1 United Kingdom

The United Kingdom does not have a written constitution.⁶⁴³ Historically, the United Kingdom may be described as a relatively stable nation whose legal framework has evolved over centuries with relative stability.⁶⁴⁴ The UK is governed by a collection of acts, policies, regulations, common law doctrines, codes of practice, conventions, declarations and recommendations and also adheres to international law.⁶⁴⁵

4.2.1.1 Human Tissue Act of 2004 (HT Act)

The Human Tissue Act of 2004 (HT Act),⁶⁴⁶ proposed in 2004, came into operation in September 2006 in the UK.⁶⁴⁷ The Act was supposedly a response to concerns raised over incidents at Bristol Royal Infirmary and the Royal Liverpool Children's Hospital, where it was established that organs and tissue were being removed and stored without the proper consent of the children who had died. ⁶⁴⁸ The HT Act covers England, Wales and Northern Ireland, except for section 45 and Schedule 4 (which refers to non-consensual DNA analysis) which applies throughout the UK.⁶⁴⁹ The Act sets out to regulate activities concerning the removal, storage, use and disposal of human tissue.⁶⁵⁰ The fundamental principle of the Act relates to consent; different consent requirements apply regarding the removal and use of tissue from the deceased or the living.⁶⁵¹ The HT Act makes no mention of benefit sharing with the donors of human material. Section 32 of the Act which

https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 26 May 2023).

⁶⁴³Constitutions organise, distribute and regulate state power at http://www.ucl.ac.uk/constitution-unit/what-uk-constitution/what-uk-constitution (Date of use: 23 May 2023).

⁶⁴⁴What the UK Constitution? http://www.ucl.ac.uk/constitution-unit/what-uk-constitution/what-uk-constitution (Date of use: 23 May 2023).

⁶⁴⁵What is the UK Constitution? http://www.ucl.ac.uk/constitution-unit/what-uk-constitution/what-uk-constitution (Date of use: 23 May 2023).

⁶⁴⁶UK Human Tissue Act 2004 https://www.legislation.gov.uk/ukpga/2004/30/data.pdf (Date of use: 23 May 2023).

⁶⁴⁷Angell A, Terrant C and Dixon-Woods M "Research involving storage and use of human tissue: how did the Human Tissue Act 2004 affect decisions by Research Ethics Committees?" 2009 J Clin Pathol 62(9):825–9.

⁶⁴⁸ Explanatory Notes of the Human Tissue Act https://www.eui.eu/Projects/InternationalArtHeritageLaw/Documents/NationalLegislation/UnitedKin gdom/humantissueact2004explanatorynotes.pdf (Date of use: 23 may 2023).

⁶⁴⁹UK Human Tissue Act 2004 https://www.legislation.gov.uk/ukpga/2004/30/data.pdf (Date of use: 23 May 2023).

⁶⁵⁰UK Human Tissue Authority (HTA) https://www.hta.gov.uk/guidance-professionals/hta-legislation/human-tissue-act-2004 (Date of use: 23 May 2023).

⁶⁵¹UK Human Tissue Authority https://www.hta.gov.uk/guidance-professionals/hta-legislation/human-tissue-act-2004 (Date of use: 23 May 2023).

prohibits commercial dealings in human material provides in section 32(7) of the Act as follows:⁶⁵²

References in subsections (1) and (2) to reward, in relation to the supply of any controlled material, do not include payment in money or money's worth for defraying or reimbursing—

- (a) any expenses incurred in, or in connection with, transporting, removing, preparing, preserving or storing the material,
- (b) any liability incurred in respect of—
- (i) expenses incurred by a third party in, or in connection with, any of the activities mentioned in paragraph (a), or
- (ii) a payment in relation to which subsection (6) has effect, or
- (c) any expenses or loss of earnings incurred by the person from whose body the material comes so far as reasonably and directly attributable to his supplying the material from his body.⁶⁵³

Section 54(7) of the Act states that material shall not be regarded as from a human body if it is created outside the human body⁶⁵⁴ (such as cell lines).

4.2.1.2 The Human Tissue Authority (HTA)

The Human Tissue Act of 2004 provides for the framework of the Human Tissue Authority (HTA), a regulatory body with the mandate to regulate the removal of human tissue for a range of purposes, including research, medical treatment, education and training. The HTA published seven Codes of Practice to fulfil its regulatory mandate. The Code of Practice of Guiding Principles and the Fundamental Principle of Consent of the HTA emphasises that the dignity of the donor should be respected at all times and where human tissue is imported, importers should try to make sure that it is from a country that has an appropriate ethical and legal framework.

It is noteworthy that any medical findings that may emerge during tissue research that may be medically significant to tissue donors, should be managed by researchers in a manner

⁶⁵² Section 32 (7) of The Human Tissue Act 2004.

⁶⁵³Section 32 (7) of The Human Tissue Act 2004.

⁶⁵⁴Section 54 (7) of The Human Tissue Act 2004.

⁶⁵⁵HTA Code of Practice A https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 23 May 2023).

⁶⁵⁶HTA Code of Practice A https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 23 May 2023).

⁶⁵⁷HTA Code of Practice A https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 23 May 2023).

⁶⁵⁸HTA Code of Practice A https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 23 May 2023).

that respects the consent process. In this regard, section 34 of the HTA Code of Practice and Standards for Research provides that:⁶⁵⁹

Findings of potential medical importance to donors may be made while undertaking human tissue research, including 'incidental findings' beyond the aims of the research. There is no single approach for the feedback of such findings. Researchers are therefore encouraged to consider how they would manage such findings and should be able to demonstrate appropriate arrangements where these are relevant, reflecting these clearly in the information used to support the consent process.⁶⁶⁰

4.2.1.3 The Human Research Authority (HRA)

The Human Research Authority (HRA) was established under the provisions of the Care Act 2014.⁶⁶¹ The HRA protects and promotes the interests of patients and the public in health and social care research.⁶⁶² It is the single body responsible for the oversight of research processes in the UK and achieves its mandate by its published Standard Operating Procedures (SOPs) which each Research Ethics Committee (REC) within the UK Health Departments' Research Ethics Service is required to adopt.⁶⁶³ In the current version of the SOPs,⁶⁶⁴ section 12.3 requires (in reference to research involving human tissue) that the REC review processes should facilitate research that is of benefit to society, within the legal framework established by statute and common law in the UK.⁶⁶⁵

Section 12.31 states that RECs undertaking the ethical review of Research Tissue Banks (RTBs) should note whether there are plans in place to provide donors with feedback on any clinically significant information obtained in research using their samples.⁶⁶⁶ The SOPs also suggest that samples should not be released from RTBs to a project that would need

⁶⁵⁹HTA Code of Practice E https://content.hta.gov.uk/sites/default/files/2020-11/Code%20E.pdf (Date of use: 23 May 2023).

⁶⁶⁰Section 34 of the HTA Code of Practice and Standards E for Research page 10

⁶⁶¹NHS Health Research Authority https://www.hra.nhs.uk/about-us/ (Date of use: 23 May 2023).

⁶⁶²NHS Health Research Authorityhttps://www.hra.nhs.uk/about-us/ (Date of use: 23 May 2023).

⁶⁶³NHS Research Ethics Committee Standard Operating Procedureshttps://www.hra.nhs.uk/about-us/committees-and-services/res-and-recs/research-ethics-committee-standard-operating-procedures/ (Date of use: 23 May 2023).

⁶⁶⁴Version 7.5.1 of the HRA SOPs for RECs which came into effect from 2 August 2021.

⁶⁶⁵ Section 12.3 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 23 May 2023).

⁶⁶⁶Section 12.31 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 23 May 2023).

further data or tissue from donors, unless it is for feedback on clinically significant information.⁶⁶⁷

Interestingly, where applications for a collaborative research project are presented to UK RECs and the samples are from donors from another country, the UK RECs are not expected to undertake a detailed review of the consent arrangements or any other research activities undertaken by collaborators in the source country.⁶⁶⁸

4.2.1.4 Medical Research Council (MRC)

The Medical Research Council (MRC) funds medical research to prevent illness, produce therapies and improve human health. The MRC has established its own ethical and governance framework in a series of guidelines to which its research units, institutes and their funded research projects are required to adhere. In the MRC ethics series, *Good Research Practice: Principles and Guidelines*, section 2C refers to collaborative research, specifying that formal written agreements should be used to clarify and agree on key aspects, which include but are not limited to, the responsibilities, ownership, custodianship, transfer and arrangement of research data and samples (including return or disposal) for future use, as well as arrangements for handling intellectual property.

The MRC's Operational and Ethical Guidelines relating to Human Tissue and Biological Samples for Use in Research⁶⁷² specifically state that samples donated for research are to be treated as donations although conditions may sometimes apply.⁶⁷³ The MRC Guidelines promote a gift relationship between research donors and researchers and advise that the formal responsibility for custodianships of collection human biological

119

⁶⁶⁷ Section 12.32(c) of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 23 May 2023).

⁶⁶⁸ Section 12.47 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 23 May 2023).

⁶⁶⁹The Medical Research Council (MRC) https://www.ukri.org/councils/mrc/ (Date of use: 23 May 2023).

⁶⁷⁰MRC principles and guidelines for good research practice https://www.ukri.org/publications/principles-and-guidelines-for-good-research-practice/ (Date of use: 23 May 2023).

⁶⁷¹Section 2C of MRC Ethics Series of Good research practice: Principles and Guidelines https://www.ukri.org/publications/principles-and-guidelines-for-good-research-practice/ (Date of use: 23 May 2023).

⁶⁷²MRC Human tissue and biological samples for use in research https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 23 May 2023).

⁶⁷³Section 2 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 23 May 2023).

materials should rest with institutions/corporates rather than individual researchers.⁶⁷⁴ Section 7 of these guidelines refers to financial issues in research using human biological materials and reiterates that the MRC stands by the principle that the human body and its parts cannot be commodified.⁶⁷⁵ The MRC allows payments to be made to research participants to compensate for expenses and time but prohibits the kind of payments that could promote undue inducement.⁶⁷⁶ This provision resembles section 60(4) of South Africa's NHA that prohibits payment for tissue donations, except for the compensation of "reasonable" costs, which point to costs relating to travel and time spent.

The MRC recognises the need to align with industry in supporting research and asserts that human biological materials do not have inherent intellectual property (IP) but that IP can arise via research that uses HBMs, which may be sold or licensed.⁶⁷⁷ The MRC encourages clarity of arrangements when granting commercial access to human biological materials that were initially donated for research projects funded by the public or charity sectors.

While acknowledgeing that donors, where possible, must be informed when their sample or products derived from the sample may be commercialised, the MRC clarifies that donors must be informed that they are not entitled to any ensuing profits or have any IP rights generated from sample use in the academic sector.⁶⁷⁸

It is clear that no specific benefit sharing frameworks exist with the donor or communities in the MRC guidelines relating to health research using human biological materials.

The next section will turn to the relevant Australian ethico-legal benefit sharing frameworks relevant to research using HBMs.

120

_

⁶⁷⁴Section 2 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 24 May 2023).

⁶⁷⁵Section 7 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 24 May 2023).

⁶⁷⁶Section 7 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 24 May 2023).

⁶⁷⁷Section 7 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 24 May 2023).

⁶⁷⁸Section 7 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 24 May 2023).

4.2.2 Australia

Australia is similar to the United Kingdom regarding its position as a high-income country (HIC) that occupies a leading position in health research.⁶⁷⁹ Australia has a written constitution which is the supreme law of the land under which the government functions. The National Health and Medical Research Council Act 1992 (NHMRC Act) established the National Health and Medical Research Council (NHMRC), 680 whose mandate, aside from funding health and medical research in Australia, is, among others, to advise the Australian Government and facilitate networking in the research community by bringing academics and industry together, whilst also building commercial literacy among researchers and helping them to protect intellectual property.⁶⁸¹

In 2007 the NHMRC, together with the Australian Research Council (ARC) and Universities of Australia, published the National Statement on Ethical Conduct in Human Research (the National Statement) applicable to all research involving human beings.⁶⁸² The National Statement was revised again in 2018.

4.2.2.1 National Statement on Ethical Conduct in Human Research (the National Statement)

The concept of benefit sharing is prevalent throughout the National Statement, which focuses on the ethical aspects of the design, review and conduct of human research. It is explicitly stated that the National Statement is not a legally binding document.⁶⁸³ Nevertheless, access to research funds from the body would require compliance to the principles of the National Statement even though it is not legally binding.

⁶⁷⁹Chalmers D "Biobanking and privacy laws in Australia" 2015 Journal of Law Medicine and Ethics 43(4):703-713.

⁶⁸⁰Australia National Statement on Ethical Conduct in Human Research (The Statement) https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 24 May 2023).

⁶⁸¹Australia National Health and Medical Research Council Research Policy https://www.nhmrc.gov.au/research-policy (Date of use: 24 May 2023).

The National Statement https://www.nhmrc.gov.au/about-us/publications/national-statement-ethicalconduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 24 May 2023).

⁶⁸³https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 20 October 2021).

Section 1 of the National Statement⁶⁸⁴ refers to the values and principles of ethical conduct, one of which is justice. The National Statement affirms that distributive justice is encouraged where there is a fair distribution of the benefits and burdens of research. Section 1.4 (d) of the National Statement explains that in research that is just, there is a fair distribution of the benefits of participation in the research. Section 1.4(f) reiterates that there should be fair access to the benefits of research.⁶⁸⁵ Section 1.11 of the National Statement refers to the value of respect and urges researchers to respect the privacy and confidentiality of research participants and also to be culturally sensitive to such research participants and honour any specific agreements made with the participants or their communities.⁶⁸⁶

Section 2 of the National Statement is dedicated to the themes of consent, risk and benefit.⁶⁸⁷ In Chapter 2.1.3⁶⁸⁸ the guidelines stipulate that some of the steps that need to be taken to acceptably mitigate risks, should include identifying the potential benefits and to whom these benefits are likely to accrue. The guidelines further stipulate that when weighing the risk-to-benefit ratio:

Those reviewing the research should take into account any willingness by participant populations to assume greater risks because of the potential benefits to them, their families, or groups to which they belong.⁶⁸⁹

The guidelines also instruct that for someone to consent to participate in research, the purpose, risks and potential benefits of the research must be discussed with the participant.⁶⁹⁰ Chapter 2.2.6 provides that before participants give consent to participate in research, information regarding any payments to participants or any expected benefits to the wider community must be communicated to them.⁶⁹¹

Referring to research using human biomaterials (read together with all the other guidelines of the National Statement), chapter 3.2.12(j) advises that donors should be notified of any

⁶⁸⁴Section 1 of the National Statement on Ethical Conduct in Human Research of the NHMRC at 9–11.

⁶⁸⁵Section 1.4 of the National Statement on Ethical Conduct in Human Research of the NHMRC at 10.

⁶⁸⁶Section 1.11 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 11.

⁶⁸⁷Section 2 of the National Statement on Ethical Conduct in Human Research of the NHMRC at 12–22.

⁶⁸⁸Chapter 2.1.3 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 14.

⁶⁸⁹Chapter 2.1.3 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 14.

⁶⁹⁰Chapter 2.2.2 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 16.

⁶⁹¹Chapter 2.2.6 (i)(j)(l) of the National Statement on Ethical Conduct in Human Research of the NHRMC at 17.

potential of commercialisation of any outcomes using their HBMs, how this would be managed and to whom the benefits, if any, would be distributed.⁶⁹²

4.2.2.1.1 Research involving marginalised communities

Chapter 4.7⁶⁹³ of the National Statement pronounces on human research involving the Aboriginal and Torres Island peoples.⁶⁹⁴ The Guidelines stress that the benefits from research involving these communities should include the advancement and capacity building of these communities.⁶⁹⁵ Research should advance their interests and all possible benefits should be discussed and agreed to by the Aboriginal or Torres Strait Islander research stakeholders.⁶⁹⁶ All benefits that materialise from research with participants from these communities should be distributed fairly and in a manner that is agreeable to the participants.⁶⁹⁷

4.2.2.1.2 Research involving international collaborations

With regard research collaborations between Australia and other countries, chapter 4.8.11⁶⁹⁸ provides that the distribution of benefits and burdens should be fair to the participants as well as their communities and that the research should not be exploitative. The research should also consider the expectations of the participants, their communities and participants' post-research welfare.⁶⁹⁹

4.2.2.2 Australian Code for the Responsible Conduct of Research (the Code)

The Australian Code for the Responsible Conduct of Research (the Code) strives to encourage research to be conducted ethically, responsibly and with integrity.⁷⁰⁰ The Code outlines the expectations for research undertaken in Australia or conducted under the

⁶⁹²Chapter 3.2.12(j) of the National Statement on Ethical Conduct in Human Research of the NHRMC at 45. ⁶⁹³Chapter 4.7 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 77-79.

⁶⁹⁴According to Dudgeon, P, Wright M, Paradies Y *et al.* "The social, cultural and historical context of Aboriginal and Torres Strait Islander Australians" 2010 *Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practice*" 25–42.

These are two cultural groups that make up Indigenous Australia, now in the minority in Australia. Their colonisation marginalised the groups and they have had to struggle to claim equality and cultural recognition.

⁶⁹⁵Chapter 4.7.7 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 78.

⁶⁹⁶Chapter 4.7.8 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 78.

⁶⁹⁷Chapter 4.7.9 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 79.

⁶⁹⁸Chapter 4.8.11 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 81.

⁶⁹⁹Chapter 4.8.12 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 81.

⁷⁰⁰Australia Code for the Responsible Conduct of Research https://www.nhmrc.gov.au/about-us/publications/australian-code-responsible-conduct-research-2018#block-views-block-file-attachments-content-block-1 (Date of use: 24 May 2023).

auspices of Australian institutions.⁷⁰¹ The Code consists of several Guides that support its agenda. One of the Guides pertains to collaborative research.⁷⁰² Section 3.1 of the Guide on Collaborative Research⁷⁰³ states that institutions that become involved in a collaborative research project should ensure that an agreement is reached with all project partners that should include any plans to commercialise research outputs and any entitlements to commercial returns.

It is clear from the discussion above that Australia has made significant strides in the development of an ethico-legal framework for benefit sharing in health research using HBMs.

Having analysed the benefit sharing frameworks in the UK and Australia, two HICs that often initiate health research collaborations with LMICs, the next section will turn to a discussion of the frameworks in Uganda and South Africa.

4.2.3 Uganda

Uganda is an LMIC and like most African countries, including South Africa, its ethical and regulatory frameworks show some influence by those of HICs.⁷⁰⁴ The impact of the ethicolegal frameworks of HICs is not always desirable, because the HIC frameworks are not always a good fit for the different contexts that apply in LMICs. These HIC frameworks may also fail to consider the traditional and cultural significance attached to human biological material found in LMICs, which in turn has an impact on LMIC participants involved in health research.

⁻

⁷⁰¹Australia Code for the Responsible Conduct of Research https://www.nhmrc.gov.au/about-us/publications/australian-code-responsible-conduct-research-2018#block-views-block-file-attachments-content-block-1 (Date of use: 24 May 2023).

⁷⁰²Australia Code for the Responsible Conduct of Research https://www.nhmrc.gov.au/about-us/publications/australian-code-responsible-conduct-research-2018#block-views-block-file-attachments-content-block-1 (Date of use: 24 May 2023).

⁷⁰³Section 3.1 of the Guide on Collaborative Research at 2–3.

⁷⁰⁴Sathar MA and Dhai A "Laws, regulations and guidelines of developed countries, developing countries in Africa, and BRICS regions pertaining to the use of human biological material (HBM) in research" 2012 *SAJBL* 5(1):51-54.

4.2.3.1 Constitution of Uganda

The Constitution of Uganda was adopted and enacted in 1995 and is the supreme law of the land.⁷⁰⁵ Section XX of the Constitution, in fulfilling its social and economic mandate, states that "the State shall take all practical measures to ensure the provision of basic medical services to the population".⁷⁰⁶

Although is no mention of health research in the Constitution, the Constitution acknowledges that Uganda's people have previously been exposed to exploitation and that their interests and well-being should be safeguarded.⁷⁰⁷

A closer look at the Public Health Act,⁷⁰⁸ enacted in 1935 to consolidate the law regarding the preservation of public health, reveals that it does not comment on all matters regarding research and makes no mention of benefit sharing.

4.2.3.2 National Guidelines for Research Involving Humans as Research Participants, (National Guidelines)

The Uganda National Council for Science and Technology (UNCST), founded in 1990, aims to advise on the formulation of national policy regarding science and technology, whilst also growing and developing indigenous science and technology.

The National Guidelines for Research involving Humans as Research Participants⁷⁰⁹ (National Guidelines) were published in 2014 by the UNCST and are the first guidelines to regulate health research in Uganda. The National Guidelines state that research participants should benefit from research and not be exploited. The oversight of research

125

⁷⁰⁵Constitution of Uganda https://www.ilo.org/dyn/natlex/docs/ELECTRONIC/44038/90491/F206329993/UGA44038.pdf (Date of use: 24 May 2023).

⁷⁰⁶Section XX of The Constitution of the Republic of Uganda.

⁷⁰⁷Constitution of Uganda https://www.ilo.org/dyn/natlex/docs/ELECTRONIC/44038/90491/F206329993/UGA44038.pdf (Date of use: 24 May 2023).

⁷⁰⁸The Public Health Act of Uganda http://library.health.go.ug/sites/default/files/resources/Public%20Health%20Act.pdf (Date of use: 24 May 2023).

⁷⁰⁹Uganda National Guidelines for research Involving Humans as Research Participants https://uncst.go.ug/main/wp-content/uploads/download-manager-files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 24 May 2024).

involving humans as research participants is done first at the organisational level via Research Ethics Committees (RECs) and then at the national level by UNCST, together with the Uganda National Health Research Organisation (UNHRO), for health research.⁷¹⁰

4.2.3.2.1 Research involving international collaborations

As far as international collaborative research is concerned, section 4.5.4(b)⁷¹¹ of the National Guidelines provides that the local Research Ethics Committee (REC) overseeing the project is responsible for the project, since the local REC is in a better position to understand the cultural sensitivities of the population.⁷¹²

Concerning benefits, section 5.3.1⁷¹³ of the National Guidelines dictates that an informed consent form must be provided to each research participant, which should include a description of the reasonably expected resultant benefits for the participant or others, 714 a statement that if the participant withdraws from the research they will still be entitled to the said benefits,⁷¹⁵ and finally a statement that the participants will receive feedback on the research progress and any incidental research findings.⁷¹⁶

Section 6.7 suggests that incentives offered to research participants should not be considered a research benefit but rather a recruitment incentive and should not unduly induce research participants.⁷¹⁷

4.2.3.2.2 Research HBMs and ownership of human biological samples

Section 10⁷¹⁸ of the National Guidelines specifically applies to human materials and declares that a consent form that is separate from the enrolment consent form must be used for samples that are collected and stored for secondary use. The secondary consent

⁷¹⁰Section 3.1 of the Uganda "National Guidelines" at 4.

⁷¹¹Section 4.5.4 (b) of the Uganda "National Guidelines" at 12.

⁷¹²Section 4.5.4 (b) of the Uganda "National Guidelines" at 12.

⁷¹³Section 5.3.1 (c) of the Uganda "National Guidelines "at 17.

⁷¹⁴Section 5.3.1 (c) of the Uganda "National Guidelines" at 17.
715Section 5.3.1 (h) of the Uganda "National Guidelines" at 17.
716Section 5.3.1 (l) of the Uganda "National Guidelines" at 17.

⁷¹⁷Section 6.7 of the Uganda "National Guidelines" at 22.

⁷¹⁸Section 10 of the Uganda "National Guidelines" at 28.

form should include information on the potential risks and benefits of storing samples for future research.⁷¹⁹

Regarding the ownership of samples, the National Guidelines recognise that samples belong to the sample donors who could withdraw the samples if samples are linked.⁷²⁰ A custodianship exists between the sample donors and the recognised institution which holds the donated sample.⁷²¹ In the event that the sample is transferred between institutions, locally or internationally, the custodian must negotiate an agreement with the receiving institution in the form of a Material Transfer Agreement (MTA).⁷²² The MTA should have clauses stating (1) who owns any new products resulting from use of the transferred sample (if this is absent, the provider organisation assumes ownership in Uganda); (2) directions on what to do if the product of research is commercialisable (including sharing royalties); (3) a separate MTA if commercialisation occurs; (4) what technologies would be transferred to the provider organisation or country (Uganda) and (5) other benefits, such as capacity building and/or infrastructure development that the provider organisation should expect.⁷²³

Section 13.0(a)⁷²⁴ states that when the indigenous knowledge of a community is used, the community should receive fair benefits from the utilisation of such knowledge.

4.2.3.3 National Research Biobanking Guidelines

The National Research Biobanking Guidelines were published in January 2021.⁷²⁵ The guidelines aim to establish a framework for the certification and operation of biobanks in several fields (including healthcare) and to provide easy access to high-quality biospecimens and their associated data from the biobanks.⁷²⁶

https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 24 May 2023).

⁷¹⁹Section 10.2 of the Uganda "National Guidelines" at 28.

⁷²⁰Section 10.3 of the Uganda "National Guidelines" at 28. ⁷²¹Section 10.3 of the Uganda "National Guidelines" at 28.

⁷²²Section 10.4 of the Uganda "National Guidelines" at 28.

⁷²³ Section 10.4(j)(k)(l) of the Uganda "National Guidelines" at 30.

⁷²⁴Section 10.4(j)(k)(i) of the Uganda "National Guidelines" at 30.

⁷²⁵Uganda National Research Biobanking Guidelines https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 24 may 2023).

⁷²⁶Uganda National Research Biobanking Guidelines

Section 1.1⁷²⁷ of the Guidelines provide that the rights and welfare of research participants and the common good are more important than the research interests of the custodian organisation and the end users of the biobank.⁷²⁸

Section 8⁷²⁹ refers to ownership and custodianship as well as benefit sharing and intellectual property. Section 8.1⁷³⁰ states that sample donors own their biospecimens, with the primary source institution (a recognised and registered organisation in Uganda) as the custodian of the biospecimens and having the authority, via a Material Transfer Agreement (MTA) with the sample donor, to decide how to use, transfer, store and decide on the future use of the samples while considering the rights and welfare of the sample donor.⁷³¹

4.2.3.3.1 Intellectual property

Addressing any intellectual property that may result from research using biospecimens in Section 8.2,⁷³² the guidelines require the biobank and primary source institution to define an intellectual property policy which should be enforced through an MTA and/or a Data Transfer Agreement (DTA).⁷³³ The primary source institution is to ensure that:⁷³⁴

- a. There are available policies and procedures on benefit sharing in line with applicable national policies, regulations and laws:
- b. Benefits from IP are shared in different ways and should be prenegotiated these include the; financial benefits, information, licensing, or transferring of technology or materials;
- c. The derivatives from the donors' biological material shall be taken as new
- d. products and should be considered Intellectual Property.

Thus, benefit sharing arrangements form an integral part of the Ugandan health research framework, where human materials are used for health research purposes and even which may even extend to intellectual property.

⁷²⁷Section 1.1 of the Uganda National Research Biobanking Guidelines at 1.

⁷²⁸Section 1.1 of the Uganda National Research Biobanking Guidelines at 1.

⁷²⁹Section 8 of the Uganda National Research Biobanking Guidelines at 20.

⁷³⁰Section 8.1 of the Uganda National Research Biobanking Guidelines at 20.

⁷³¹Section 8.1 of the Uganda National Research Biobanking Guidelines at 20.

⁷³²Section 8.2 of the Uganda National Research Biobanking Guidelines at 20.

⁷³³Section 8.2 of the Uganda National Research Biobanking Guidelines at 20.

⁷³⁴Section 8.2 of the Uganda National Research Biobanking Guidelines at 20.

The next section analyses the ethico-legal framework regarding benefit sharing in health research when human biological materials are used in South Africa. Since Chapter 2 of this thesis dealt extensively with the South African ethico-legal framework pertaining to benefit sharing, the section below will comprise a summary to of the key points in order to provide context to the comparisons between the selected jurisdictions in this chapter.

4.2.4 South Africa

The Constitution of the Republic of South Africa, 1996 is the supreme law of the country.⁷³⁵ The National Health Act 61 of 2003 governs all health research in the country.⁷³⁶ No legally binding law speaks directly to benefit sharing in health research using human biological materials.⁷³⁷ Health research involving HBMs are addressed in the Department of Health's Ethics in Health Research Guidelines of 2015, discussed next.

4.2.4.1 Department of Health Ethics in Health Research Guidelines, 2015

The Department of Health, National Ethics Guidelines⁷³⁸ provide a minimal benchmark for conducting ethical and responsible research in the country. Benefit sharing is not specifically addressed in the Guidelines, however the Guidelines provide that "the population from which the participants are drawn will benefit from the research results if not immediately, then in the future". The Guidelines advise that fair reimbursement for study participation is just and when recruitment is difficult, inducements may be offered to research participants. The Guidelines further stipulate that when obtaining informed consent, research participants should be informed of the potential benefits of their participation, both during and after the research.

⁷³⁷Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

⁷³⁵The Constitution of the Republic of South Africa https://www.justice.gov.za/legislation/constitution/saconstitution-web-eng.pdf (Date of use: 24 May 2023).

⁷³⁶61 0f 2003, Chapter 9.

⁷³⁸DoH Ethics in Health Research Guideline https://www.health.gov.za/up-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

⁷³⁹DoH Ethics in Health Research Guidelines https://www.health.gov.za/up-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

⁷⁴⁰DoH Ethics in Health Research Guidelines http://www.health.gov.za/up-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf_(Date of use: 23 May 2023).

⁷⁴¹DoH Ethics in Health Research Guidelineshttp://www.health.gov.za/up-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf_(Date of use: 23 May 2023).

4.2.4.2 Health Professions Council of South Africa (HPCSA) Guidelines

In 2016, the Health Professions Council of South Africa (HPCSA) published revised guidelines for health practitioners and health researchers in their mandate as health professionals.⁷⁴² Booklet 13 deals with ethical guidelines for health researchers. The Guidelines caution that, considering South Africa's violent and discriminatory past, which led to the marginalisation of specific racial groups, the misuse of power in research cannot be ignored.⁷⁴³ The Guidelines also state that burdens and benefits should be balanced within different population groups. Section 6.6.3 of the Guidelines proposes that at the end of a study, research participants should be entitled to benefit from the study by accessing the best proven prophylactic, diagnostic and therapeutic methods identified by such study.⁷⁴⁴

4.2.4.3 National Material Transfer Agreement of Human Biological Materials

In 2018, the government of South Africa gazetted a template for the National Material Transfer Agreement (MTA) of Human Biological Materials.⁷⁴⁵ Section 7 of the MTA expressly states that "the sharing of benefits should be discussed and negotiated between the Provider and Recipient before Materials are transferred to the Recipient".⁷⁴⁶ The MTA defines a benefit as:⁷⁴⁷

Amongst others, the sharing of information; use of research results; royalties; acknowledgement of the Provider as the source of the Materials; publication rights; transfer of technology or Materials; and capacity building.

Benefit sharing is also described in the MTA as: "[...] the process or act of sharing in the benefits that derive from the Project in a manner that is fair and equitable". 748

The SA MTA is a legally binding template that must be completed and adjusted for the individual institution's context by the providing and recipient institutes (with the HREC

⁷⁴²HPCSA mandate https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_1_Guidelines_for_Good_Pr actice_vDec_2021.pdf (Date of use: 26 May 2023).

⁷⁴³HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s for Health Researchers.pdf (Date of use: 26 May 2023).

⁷⁴⁴HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s_for_Health_Researchers.pdf (Date of use: 26 May 2023).

⁷⁴⁵Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷⁴⁶Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷⁴⁷Section 2.2 of the SA MTA, Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷⁴⁸Section 2.3 of the SA MTA, Proclamation No 719 Government Gazette 41781 of 20 July 2018.

constituting a party to the agreement) prior to the transfer of HBMs outside South Africa's borders. The implementation of the national MTA provides a template that comprises of key elements that are legally recognised. Benefit sharing is one of these elements. However, as parties to the SA MTA may negotiate the terms of benefit sharing arrangements and as such, benefit sharing is not an absolute requirement. The SA MTA also does not provide guidance on what terms should be included in a benefit sharing agreement.

4.3 Comparative analysis

Although ethical guidelines are generally not considered legally binding, non-adherence to their principles could result in healthcare professionals being found guilty of professional misconduct. In addition, RECs may not approve research projects that dismiss ethics principles, as outlined in the Guidelines above.

In most HICs, the notion of altruistic research that attempts to avoid undue inducement⁷⁴⁹ is still very much prevalent, as is exemplified by the ethical guidelines relating to health research in the United Kingdom.⁷⁵⁰ In the UK, all human biological material used in health research is regarded as an altruistic donation to benefit all towards fulfilling distributive justice.⁷⁵¹ This also fulfils the general rule that participants relinquish ownership rights to biospecimens once they have donated samples.⁷⁵² Scholars have argued that the notion of the distributive justice of benefits is only suited to HICs where everybody would benefit equally from the results of health research. In LMICs, where there is a history of injustices in health research (often brought about by a lack of integrity), cultural insensitivity and

_

⁷⁴⁹Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 9–32.

⁷⁵⁰Section 12.3 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 23 May 2023).

⁷⁵¹Section 2 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 23 May 2023).

⁷⁵²Mahomed S "Human biobanking in developed and developing countries: an ethico-legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–60.

unfair collaborations,⁷⁵³ the fair-exchange model for benefit sharing fails and could, indeed, lead to exploitation.⁷⁵⁴

Interestingly, the Australian National Statement on Ethical Conduct in Human Research⁷⁵⁵ has incorporated the concept of benefit sharing as one of the pillars of ethical research. The statement elaborates by stating that not only should benefit sharing arrangements be in place, but the need to identify to whom said benefits should accrue also exists.⁷⁵⁶ Although the concept of the ownership of human biological materials is not specifically mentioned, the National Statement is unambiguous that donors of HBMs should be notified in the event of the commercialisation of outcomes using their samples and how this would be managed.⁷⁵⁷ This is unlike the UK, where guidelines stipulate that HBMs donors are not entitled to any profits that might arise from the commercialisation of outcomes using their samples.

In Uganda, the National Guidelines state that all participants maintain ownership of their samples.⁷⁵⁸ This statement contradicts case law, such as *Moore v. Regents of the University of California*,⁷⁵⁹ *Greenberg v. Miami Children's Hospital Research Institute, Inc*⁷⁶⁰ and *Washington University v. Catalona*,⁷⁶¹ which show the law's discomfort with classifying HBMs in terms of property and ownership.⁷⁶² In terms of common law in

⁷⁵³Moodley K and Singh S "'It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1): 1–9.

⁷⁵⁴Andanda P, Schroeder D, Chaturvedi S *et al* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

⁷⁵⁵Australian National Statement on Ethical Conduct in Human Researchhttps://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 26 May 2023).

⁷⁵⁶Section 2 of the Australian National Statement on Ethical Conduct in Human Research at 12–15.

⁷⁵⁷Section 3 of the Australian National Statement on Ethical Conduct in Human Research at 23–60.

⁷⁵⁸Section 10.3 of the Ugandan "National Guidelines".

⁷⁵⁹Cited in Chapter 3 of this thesis, from Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481: 167–191. The California Supreme Court rejected Mr Moore's property rights claim to his cells, concluding that this would hamper scientists from conducting medical research.

⁷⁶⁰Cited In Chapter 3 of this thesis, from Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191 where the Florida Court determined that any property right to blood and tissue samples "evaporates once the sample is voluntarily given to a third party".

⁷⁶¹Cited in Chapter 3 of this thesis, from Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191, whereby the Court dictated that the patients had altruistically donated the HBMs to research and no longer retained any ownership rights to the samples or the authority to transfer said samples to third parties.

⁷⁶²Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

Uganda, ownership is a real right, defined on the basis of entitlements.⁷⁶³ Thus, the National Guidelines state that even when participants withdraw from a study, they are still entitled to the benefits promised to them in the informed consent form.⁷⁶⁴ The Uganda National Guidelines provide that the primary institute or organisation will hold the donors samples in trust, which will always act in the best interest of the sample donors.⁷⁶⁵ The custodian will then draft MTAs when samples are transferred outside Uganda to other institutions/organisations that require clauses pertaining to varying benefit arrangements, including the development of a product of commercial value with the use of the donated HBMs.⁷⁶⁶

The National Research Biobanking Guidelines of 2021⁷⁶⁷ reiterate that the donor of HBM samples remains the owner of a sample, who has to adhere to a custodianship relationship with a primary source institution in Uganda.⁷⁶⁸ The primary source institution has the authority to decide the use, transfer, storage and future use of the samples, considering the rights and welfare of the sample donor. As such, a biobank will have full custody of the sample per the MTA between parties.⁷⁶⁹

In South Africa, similar to Uganda's position, the National Material Transfer Agreement of Human Biological Materials provides in section 3.3 that the provider of the HBMs remains the custodian of the materials and that the donor remains the owner of the material until such materials are destroyed.⁷⁷⁰ Benefit sharing is specifically mentioned in section 7 of the MTA, which states that before human biological materials are transferred, the sharing of benefits should be discussed and negotiated.⁷⁷¹

2023).

_

⁷⁶³Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁷⁶⁴Section 5.3.1 (h) of the Ugandan "National Guidelines".

⁷⁶⁵Uganda National Guidelines for Research involving Humans as Research Participants https://uncst.go.ug/main/wp-content/uploads/download-manager-files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 24 May

⁷⁶⁶Uganda National Guidelines for Research involving Humans as Research participants https://uncst.go.ug/main/wp-content/uploads/download-manager-files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 24 May 2023).

⁷⁶⁷Uganda National Biobanking Guidelines https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 24 May 2023).

⁷⁶⁸Uganda National Biobanking Guidelineshttps://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 24 May 2023).

⁷⁶⁹Uganda National Biobanking Guidelines https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf ((Date of use: 24 May 2023).

⁷⁷⁰Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷⁷¹Proclamation No 719 Government Gazette 41781 of 20 July 2018.

To conclude, whilst the UK has no framework governing benefit sharing for health research using HBMs, Australia and Uganda have explicitly included benefit sharing between donors and recipients in their ethical guidelines to promote ethical health research. The South African framework acknowledges the need for benefit sharing in health research using HBMs, yet it does not provide firm guidance or direction on how this should occur.

It is also worth noting that Australia has specific provisions regarding benefit sharing when dealing with specific communities deemed vulnerable or marginalised communities. The Guidelines stipulate that in research using HBMs from marginalised communities, benefit sharing agreements should be discussed, agreed upon with the community, and also be delivered. These agreements must be for the benefit of the communities and their advancement. The UK guidelines do not address research that uses HBMs from specific populations. Uganda's National Guidelines provide that when the indigenous knowledge of a community is used, the community should receive fair benefits from the utilisation of their knowledge. The same National Guidelines also emphasise the need for health research to return benefits to the community and the country and to avoid exploitative research.

Turning to South Africa, there is no mention of community benefits or how to address benefit sharing with marginalised communities in ethico-legal frameworks in the context of benefit sharing. South Africa has a difficult past, which resulted in the creation of vulnerable and marginalised communities.⁷⁷⁶ This must be borne in mind and addressed when benefit sharing agreements are negotiated between HICs and vulnerable South African communities.

As far as collaborative research in the United Kingdom is concerned, the Guidelines of the Human Research Authority do not require RECs to review consent arrangements or any other research activities undertaken by collaborators in the source country.⁷⁷⁷ On the other

134

_

⁷⁷²Chapter 4.7.7 of the National Statement on Ethical Conduct in Human Research of the NHRMC. ⁷⁷³Chapter 4.7.7 of the National Statement on Ethical Conduct in Human Research of the NHRMC.

⁷⁷⁴Section 13.0(a) of the Uganda "National Guidelines".

⁷⁷⁵Uganda National Guidelines for Research involving Humans as Research Participants https://uncst.go.ug/main/wp-content/uploads/download-manager-

files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 24 May 2023).

 ⁷⁷⁶Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440.
 ⁷⁷⁷Section 12.47 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07lVkXt.pdf (Date of use: 24 May 2023).

hand, the MRC Principles and Guidelines require that for MRC funded research, signed agreements between research organisations and representatives of the collaborating organisation must be completed before the research commences. These agreements may include ownership, custodianship, transfer and the future use of samples.⁷⁷⁸ In Australia, the Guidelines in the National Statement indicate that in collaborative research with other countries, the distribution of benefits and burdens should be fair to participants and communities whilst being sensitive to the needs of communities and protecting them against potential exploitation.⁷⁷⁹ Uganda specifically requires that an MTA and/or a DTA is signed between the primary institute in Uganda and the receiving institute. This DTA and/or MTA has/have specific clauses pertaining to benefit arrangements that the primary institute or Uganda could expect.⁷⁸⁰ South Africa requires a provider and recipient to negotiate and agree on benefit sharing before human biological material may be exchanged.⁷⁸¹

With regard to intellectual property that may arise from the use of HBM, the UK maintains that sample donors have no intellectual property rights generated from research using their HBMs.⁷⁸² In its National Statement, Australia proposes that researchers should agree on the ownership of any property created, although the agreement need not be contractual.⁷⁸³ Uganda requires the primary organisation which is the custodian of the HBM samples to define IP policy via an MTA/DTA which outlines how the benefits arising from IP are to be shared.⁷⁸⁴ In South Africa, any IP generated from publicly funded research and development using HBMs is owned by the recipient of the public funds; however, coownership is possible with a private entity if benefit sharing arrangements are made with intellectual property creators.⁷⁸⁵

The different approaches to benefit sharing in the United Kingdom and Australia as HICs are interesting. The UK maintains that health research should be altruistic and Australia

⁷⁷⁸MRC ethics series: Good research practice: Principles and Guidelines https://www.ukri.org/wpcontent/uploads/2021/08/MRC-0208212-Good-research-practice 2014.pdf (Date of use: 24 May

⁷⁷⁹Chapter 4.8.12 of the National Statement on Ethical Conduct in Human Research of the NHRMC.

⁷⁸⁰Section 8.2 of the Uganda National Research Biobanking Guidelines at 20.

⁷⁸¹Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁷⁸²MRC ethics series: Good research practice: Principles and Guidelines ehttps://www.ukri.org/wpcontent/uploads/2021/08/MRC-0208212-Good-research-practice 2014.pdf (Date of use: 24 May 2023).

⁷⁸³Chapter 3.1.44 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 35.

⁷⁸⁴Section 8.2 of the Uganda National Research Biobanking Guidelines at 20

⁷⁸⁵Intellectual property rights from publicly financed research and development https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 24 May 2023).

determines that research cannot be ethical if all parties who participate in research do not benefit equally. As LMICs, both Uganda and South Africa were subjected to research injustices and unfair collaborations in the past. This historical context continues to inform efforts to ensure fair benefit sharing in health research using HBMs.

Table 1: Positions on ownership of HBMs and benefit sharing in Australia, the UK, Uganda and South Africa

Country	Ownership of HBMs	Benefit sharing legislation/guidelines
United Kingdom	All samples are considered altruistic donations	Not addressed
Australia	Not addressed	National Statement Ethical guidelines state that: i. Benefit sharing is one of the pillars of ethical research. ii. Benefit sharing arrangements must be in place in research and to whom these benefits are to be accrued. iii. Donors of HBMs are to be notified in the event of the commercialisation of outcomes using their samples, how this will be managed and to whom the benefits, if any, will be distributed. iv. When using samples from marginalised/vulnerable communities, benefit sharing arrangements should be agreed upon with the communities and should be for their advancement. v. In international collaborations, benefits and burdens should be fairly distributed and be fair to their communities
Uganda	Donors retain ownership of their HBMs. A custodianship exists between the donor and primary in Uganda that holds the sample	National Guidelines and Biobanking Guidelines state that: i. Informed consent forms should include anticipated benefits to the participants and communities. ii. Regarding the future use of stored HBMs, a secondary consent form is to be used which should describe the risks and benefits of storing samples for future research. iii. An MTA is required when samples are transferred between institutions. The MTA should have clauses on various benefit sharing arrangements iv. In the event of the commercialisation of outcomes using samples held in trust, a second MTA is to be negotiated stating what benefits Uganda or the primary institution in Uganda would be realised. v. In the event of any intellectual property that arises from research using HBMs, the biobank and primary source institution should define an intellectual property policy and this should be enforced using an MTA and/or data transfer Agreement (DTA).
South Africa	The Providing institute remains the custodian of the materials and the Donor remains the owner of the materials until such material is destroyed.	The National MTA states that before the transfer of HBMs, the sharing of benefits should be discussed and negotiated.

4.4 Conclusion

This chapter has identified one of the main arguments against benefit sharing in research as the potential for undue inducement that could lead to the exploitation of research participants.⁷⁸⁶ Despite this concern, many scholars have argued that not distributing fair and equal benefits to all who participate in health research, constitutes a form of exploitation.⁷⁸⁷

None of the jurisdictions explored in this chapter have a legally binding regulation that requires benefit sharing in health research where HBMs are used. Although benefit sharing is provided for in ethical guidelines in the UK, Australia and Uganda, these are only persuasive sources and not legal rules.⁷⁸⁸ It has been argued that "global bioethics have to contend with a regulatory crisis in terms of the existing public law silences and health inequities especially in low and middle income countries".⁷⁸⁹ A study by De Vries *et al*⁷⁹⁰ concludes that in a likely response to fears of exploitation regarding genomic and biobank research in LMICs, existing regulations are either absent, outdated, conservative or difficult to navigate. The lack of legislation could also be attributed to the fact that the law relies on tradition and legal precedent,⁷⁹¹ and must be seen to both follow tradition and precedent when formulating public policy or legislation in the age of rapidly evolving health research.

Ethical guidelines in the UK direct that all donations are considered gifts and that no benefit sharing is necessary. Conversely, the various guidelines in Australia, Uganda and South Africa all propose that benefit sharing is necessary and should be achieved by agreements or as part of an MTA/DTA. It was argued in this chapter that the notion of altruistic research is viable in HICs where the population has access to good healthcare, but to suggest that

⁷⁸⁶Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

⁷⁸⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

⁷⁸⁸Mahomed S "Human biobanking in developed and developing countries: an ethico-legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–60.

⁷⁸⁹Andanda P, Schroeder D, Chaturvedi S *et al* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 33–64.

⁷⁹⁰De Vies J, Munung SN, Matimba A et al "Regulation of genomic and biobanking research in Africa: a content analysis of ethics guidelines, policies and procedures from 22 African countries" 2017 BMC Medical Ethics 18(1):1–9.

⁷⁹¹Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

this should be the case in LMICs or among marginalised communities, is paternalistic and short-sighted, as most research participants in these communities live in poverty and have no or limited access to health care services.

Health research using HBMs is a dynamic and fast-evolving field, with unique ethical challenges arising in the African research context. As a result, there is a need for guidance in the legal frameworks around ethical research.

The next chapter considers an important aspect of this study, namely to interrogate what may considered as a benefit for the purpose of benefit sharing in health research, and how best all such benefits could be structured in a legal document that could assist with effectively managing benefit sharing arrangements in South Africa.

CHAPTER 5

KEY ISSUES TO BE ADDRESSED IN BENEFIT SHARING AND A PROPOSED TEMPLATE FOR BENEFIT SHARING IN SOUTH AFRICA

5.1 Introduction

Earlier chapters in this thesis have determined that the concept of sharing benefits in health research is not new and that it has become more glaringly evident in times of global health crises. ⁷⁹² It has been suggested that to regulate collaborative research, the fair benefits approach should adopt a procedural strategy. ⁷⁹³ It is indeed shameful that to date, allegations of the use of African DNA without consent for commercial applications continues to be reported in HICs without the transfer of any benefits to LMICs. ⁷⁹⁴ To wit, during the 2019 Ebola outbreak in Western Africa, blood samples that had been collected from patients, held by American and British authorities and used for commercial development, were withheld from researchers from West African countries. ⁷⁹⁵ The current global SARS-CoV2 has further magnified the issue of research participants in LMICs rarely directly benefitting from the research in which they participate, as seen by the vaccine scarcity and vaccine hoarding by HICs, despite the fact that some of these LMICs, including South Africa, were involved in COVID-19 vaccine research and trials. ⁷⁹⁶

It took a global pandemic such as COVID-19 to emphasise that especially in LMICs, benefit sharing should be considered as a matter of justice aimed at avoiding exploitation and protecting the vulnerable.⁷⁹⁷ According to Cook Lucas *et al*,⁷⁹⁸ every new global health crisis highlights the need for a solution to the exclusion of human biological materials (HBMs) from the CBD and the access and sharing of these HBMs. Nevertheless, it is

⁷⁹²Moodley K, Blockman M, Hawkridge *et al* "Hard choices: Ethical challenges in phase 1 of COVID-19 vaccine roll-out in South Africa" 2021 SAMJ 111(6):554-558.

⁷⁹³London AJ and Zollman KJS "Research at the auction block: problems for the fair benefits approach to international research" 2010 *Hastings Center Report* 40(4):34–45.

⁷⁹⁴Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 *Developing World Bioethics* 21:125–130.

⁷⁹⁵Mckenna M "Colonialists are coming for blood-Literally" 2019 *Ideas* https://www.wired.com/story/ebola-epidemic-blood-samples/ (Date of use: 26 May 2023) .

⁷⁹⁶Moodley K, Blockman M, Hawkridge AJ *et al* "Hard choices: ethical challenges in phase 1 of COVID-19 vaccine roll-out in South Africa" 2021 *SAMJ* 111(6):554–558.

⁷⁹⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

⁷⁹⁸Cook Lucas JC, Schroeder D, Arnason G *et al* "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

evident that many countries, including South Africa, lack adequate legislative structures and governance frameworks to facilitate fair benefit sharing for all who participate in health research.⁷⁹⁹ In the absence of binding law, ethics guidelines might provide guidance akin to customary international law.800

The never-ending questions around the type and range of benefits, to whom the benefits should accrue and when and how they should be provided, remain obstacles to implementing benefit sharing in health research.⁸⁰¹ The concept of benefit sharing is also problematic because various disciplines use it without a specific definition.802

Sudoi et al 803 suggest that the use of the terms "advantage" or "profits" in the definition of benefit sharing infers that those benefits could be either monetary or non-monetary. If we accept Schroeder's⁸⁰⁴ definition that benefit sharing is a technical term to be used as a tool to achieve justice, then it is necessary to redress the frequent injustices being committed in LMICs by HICs through exploitative research⁸⁰⁵ and enforce benefit sharing plans and their implementation as a prominent feature in research proposals and grant applications.806

5.2 Legal frameworks for benefit sharing

Chapter 2 of this thesis explored the legal framework for benefit sharing in South Africa and also canvassed benefit sharing models in other jurisdictions. The objective of this chapter, which is to develop a framework for benefit sharing in health research in South Africa, will require that cross-referencing to other benefit sharing frameworks discussed in chapter 2 of this thesis may be necessary.

⁷⁹⁹Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117.

⁸⁰⁰Andanda P, Schroeder D, Chaturvedi S, Mengesha E and Hodges T "Legal frameworks for benefit sharing: from biodiversity to human genomic" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 33-64.

⁸⁰¹Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 BMC Medical Ethics 22(1):1-16.

⁸⁰²Schroeder D "Benefit sharing: it's time for a definition" 2007 *J Med Ethics* 33:205–209.

⁸⁰³ Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 BMC Medical Ethics 22:102.

 ⁸⁰⁴Schroeder D "Benefit sharing: it's time for a definition" 2007 *J Med Ethics* 33:205–209.
 805Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 AMA Journal of Ethics 22(1):28-35.

⁸⁰⁶Bedeker A, Nichols M, Allie T et al. "A framework for the promotion of ethical benefit sharing in health research" 2022 BMJ Global Health 7(2): e008096.

5.2.1 Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation

The Nagoya Protocol, 807 a subsidiary agreement to the CBD that has been ratified by at least 163 countries, refers to access and the benefit sharing of non-human genetic resources. The Protocol aims to promote benefit sharing combined with material transfer agreements, but as discussed earlier in this thesis, excludes human biological resources and digital sequence information. The Nagoya Protocol proposes a substantive list of both monetary and non-monetary benefits that could be shared. Although some of the proposed benefits are not relevant to or appropriate in the context of human health research, some scholars argue that in the absence of clear international guidelines and in a world of ever advancing biotechnology where research is increasingly multifaceted and could incorporate genetic resources from plants, pathogens, animals and humans, the scope of access and benefit sharing of the CBD should expand to include human genetic resources.

5.2.2 HUGO Ethics Committee Statement on Benefit Sharing

On the 9th of April 2000, the HUGO Ethics committee issued a statement on benefit sharing between concerned parties (which may include governments, academic institutions and participating communities) regarding human biological material.⁸¹¹ One of their suggestions was that "profit-making entities dedicate a percentage (e.g., 1%–3%) of their annual net profit to healthcare infrastructure and/or to humanitarian efforts".⁸¹²

Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity 2011. http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 26 May 2023).

⁸⁰⁸Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

⁸⁰⁹Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity 2011. http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (Date of use: 26 May 2023)

⁸¹⁰ Chaturvedi S, Crager S and Ladikas M "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 153–178.

⁸¹¹HUGO Ethics Committee Statement on Benefit Sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,who%20participated%20in%20such%20research (Date of use: 24 May 2023).

⁸¹²HUGO Ethics Committee on Benefit Sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,who%20participated%20in%20such%20research. (Date of use: 20 May 2023).

5.2.3 CIOMS International Ethical Guidelines for Health-Related Research Involving Humans

The Council for International Organisations of Medical Sciences (CIOMS) updated their Guidelines in 2016, which endorse benefit sharing and advise that benefit sharing agreements⁸¹³ should be negotiated. The Guidelines do not offer any direction or guidance on how the difference in negotiating powers between stakeholders in HICs and those in LMICs should be navigated.⁸¹⁴

5.2.4 San Code of Research Ethics

Specific to South Africa, the San Code of Research Ethics⁸¹⁵ requires that expected benefits are discussed with research participants and the community. Although the benefits could be non-monetary, they could include co-research opportunities, the transfer or sharing of skills, development of research capacity, and roles for translators and research assistants.

The commercialisation of products arising from research using human biological materials has ensured that the private sector is the main investor in health research. Hese frameworks, although promoting benefit sharing, fail to address the issue of public and privately funded research designs. Some research is sponsored by governments and/or non-profit organisations, which, nonetheless, respond to public health needs as opposed to profit-driven research. The inequities in global health between LMICs and HICs are large. To address this, HICs are called upon to make data, samples and interventions

⁸¹⁴Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

⁸¹³The International Ethical Guidelines for Health-related Research Involving Humans prepared by CIOMS and WHO Geneva 2016 https://cioms.ch/wp-content/uploads/2017/01/WEB-CIOMS-EthicalGuidelines.pdf (Date of use: 23 May 2023).

⁸¹⁵Chennells R and Schroeder D "The San Code of Research Ethics: Its Origins and History" T*RUST Equitable Research* 2018 http://www.globalcodeofconduct.org/wp-content/uploads/2019/02/SanCodeHistory.pdf (Date of use: 23 May 2023).

⁸¹⁶Simm K "Benefit sharing frameworks-justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 25 May 2023).

⁸¹⁷Simm K "Benefit sharing frameworks –justifications for and against benefit sharing in human genetic research." 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 25 May 2023).

available to correct past injustices and also to enable the negotiation of fair contracts for vulnerable populations.⁸¹⁸

5.3 Challenges to benefit sharing

The need for benefit sharing in health research is glaringly evident, ⁸¹⁹ yet controversy persists. For example, when the Indonesian government withheld its avian flu samples which had been distributed via the WHO's Global Influenza Surveillance Network, the government argued that despite providing the H5N1 samples or the development of specific vaccines, the resultant vaccines would be unaffordable for its citizens. ⁸²⁰ If benefit sharing is understood to be a mechanism to counter exploitation, ⁸²¹ then a lack of trust between human biological resource providers and resource users is an obstacle preventing the implementation of fair benefit sharing between all stakeholders involved in research. ⁸²²

Samples derived from human biological materials can be costly to store since they often require a continuous cold chain, expensive freezers and generators. As this would be costly in most LMICs, the lack of appropriate infrastructure and equipment seemingly makes it easier to ship samples to HICs. Some LMICs may have the capacity to store such samples but the lack of appropriate molecular diagnostics to work with those samples is one of many obstacles hindering the implementation of equal benefit sharing between HICs and LMICs.

⁸¹⁸Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

⁸¹⁹Simm K "Benefit sharing frameworks—justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit available at: www.uclan.ac.uk/genbenefit (Date of use: 25 May 2023).

⁸²⁰ Sedyaningsih ER, Isfandari S, Soendoro T et al "Towards mutual trust, transparency and equity in virus sharing mechanism: the avian influenza case of Indonesia" 2008 Ann Acad Med Singapore 37:482– 488.

⁸²¹ Schroeder D "Benefit sharing: it's time for a definition" 2007 J Med Ethics 33:205–209.

⁸²² Sedyaningsih ER, Isfandari S, Soendoro T *et al* "Towards mutual trust, transparency and equity in virus sharing mechanism: the avian influenza case of Indonesia" 2008 *Ann Acad Med Singapore* 37:482–488

⁸²³Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

⁸²⁴Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

Some health research involving biobanking requires vast amounts of data and samples as well as the proper storing and sharing of this information.⁸²⁵ Accessing and sharing this information is key to biobanking, but the large numbers of stakeholders involved and the power dynamics between them, as well as the question of who benefits, when and how, often present challenges.⁸²⁶ Issues pertaining to the re-use of samples like consent and ownership are also obstacles to attempting to implement benefit sharing in practice.⁸²⁷

The issue of post study access also presents obstacles to the practical implementation of benefit sharing. 828 Health research using HBMs does not always result in big and/or immediate returns or viable therapeutics. 829 Many years may pass before tangible products such as therapeutics or vaccines are developed and these developments cost a considerable sum of money. A mechanism to pay for and distribute these products must be found; however, questions around how this may be achieved and who should be responsible for this could lead to unfair benefit sharing of products developed from the research. 830

Some scholars⁸³¹ suggest that research ethics committees (RECs) in LMICs are often ineffective at providing independent oversight in the implementation of fair benefit sharing frameworks. A study conducted in 1999 to examine the effectiveness of RECs in LMICs concluded that the "major constraints identified are shortage of resources and inadequate training of the REC committee members as well as pressure from researchers and sponsors".⁸³² However, with reference to South Africa, Chapter 9 of the NHA states that all institutions at which health research is conducted must establish or have access to a health REC which is registered and audited by the National Health Research Ethics

_

⁸²⁵Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 *BMC Medical ethics* 22(1):1–16.

⁸²⁶Berg K "The ethics of benefit sharing" 2001 Clin Genet 59(4):240–243.

⁸²⁷Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 *BMC Medical ethics* 22:102.

⁸²⁸ Schroeder D and Gefenas E "Realizing benefit sharing—the case of post-study obligations" 2012 *Bioethics* 26(6):305–314.

⁸²⁹Schroeder D and Gefenas E "Realizing benefit sharing–the case of post-study obligations" 2012 *Bioethics* 26(6):305–314.

⁸³⁰ Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

⁸³¹Millum J "Sharing the benefits of research fairly: two approaches" 2012 *J Med Ethics* 38:219–223.

⁸³²Nyika A, Kilama W, Chilengi R *et al.*" Composition, training needs and independence of ethics review committees across Africa: are the gate-keepers rising to the emerging challenges?" 2009 *J Med Ethics* 35:189–193.

Council.⁸³³ In 2019, Mahomed and Labuschaigne⁸³⁴ reviewed the changing role of RECs in South Africa during past decade. They concluded that the complexity of novel and specialised research, coupled with complex and intricate research protocols funded by private sponsors across national borders, are among others some of the challenges that caused capacity and resource constraints on the part of RECs, resulting in breaches regarding the protection of participants.

Sudoi *et al*⁶³⁵ identify the tensions that exist between benefit sharing and other ethical issues such as undue inducement, altruistic research and the commodification of a person, as further conceptual and practical challenges to the implementation of benefit sharing.

It is important to acknowledge that operational benefit sharing involves multiple stakeholders who must all benefit at some stage; not acknowledging this fact could impede the implementation of benefit sharing. 836 Recognising that different stakeholders are involved in research means that effective communication structures between the scientific community and relevant national and international policymakers need to be developed. 837

The conceptual and practical problems of benefit sharing remain primarily because the present ethical and legal policies to promote benefit sharing are non-binding legislation that offer inconsistent and incomplete frameworks for benefit sharing.⁸³⁸

With specific reference to South Africa, the recently gazetted Protection of Personal Information Act (POPIA)⁸³⁹ effected the legal enforcement of research participants' right to privacy regarding their data with no regard for how this could affect the use of personal information in health research in SA.⁸⁴⁰ Health research, particularly collaborative research to develop diagnostics, treatments and respond to pandemic outbreaks is expatiated when large amounts of samples and personal data can be transferred between research

⁸³³ National Health Act No.61 of 2013

⁸³⁴Mahomed S and Labuschaigne M "The role of research ethics committees in South Africa when human biological materials are transferred between institutions" 2019 *S Afr J Bioethics Law* 12(2):79-83.

⁸³⁵Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 *BMC Medical ethics* 22(1):1–16.

⁸³⁶Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

⁸³⁷Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

⁸³⁸Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117. ⁸³⁹South Africa. Protection of Personal Information Act No.4 of 2013.

⁸⁴⁰Mahomed S, Loots G and Stauton C "The role of data transfer agreements in ethically managing data sharing for research in South Africa" 2022 S Afr J Bioethics Law 15(1):26–30.

institutions locally or internationally.⁸⁴¹ Section 72(1)(e) of the POPIA allows the transfer of personal data outside South African borders if it is for the benefit of the research participant in the absence of consent from a said participant, given that if the participant could give consent, they would likely give it.⁸⁴² Mohamed *et al*⁸⁴³ point out that this exemption would require that every participant is accounted for in the decision-making process and this may be impractical and even impossible where large datasets are transferred outside of South Africa.

Finally, as highlighted by Cheng and Pang,⁸⁴⁴ the lack of HICs stakeholders' understanding of the local culture, religious beliefs and the concept of ethics in LMICs may pose further impediments to the realisation of benefit sharing.

5.4 Proposed benefit sharing models

The absence of a binding and streamlined international legal regime to direct a framework for benefit sharing in human health research becomes problematic as national governments evolve their own access and benefit sharing frameworks that might not support global research initiatives.⁸⁴⁵ The issues of public versus private or semi-private research must be considered when designing a framework for benefit sharing in health research, as should property rights, public engagement, compensation of research participants, consent issues, data sharing, secondary use of samples and data, returning results, royalties, financial gains, other types of gain and beneficiaries.⁸⁴⁶

_

⁸⁴¹Mahomed S, Loots G and Stauton C "The role of data transfer agreements in ethically managing data sharing for research in South Africa" 2022 S Afr J Bioethics Law 15(1):26–30.

⁸⁴²South Africa. Protection of Personal Information Act No.4 of 2013. Section 72(1)(e).

⁸⁴³Mahomed S, Loots G and Stauton C "The role of data transfer agreements in ethically managing data sharing for research in South Africa" 2022 S Afr J Bioethics Law 15(1):26–30.

⁸⁴⁴ Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117.

⁸⁴⁵ Chaturvedi S, Crager S, Ladikas M et al "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 153–178.

⁸⁴⁶Cambon-Thomsen A, Rial-Sebbag E and Knoppers BM "Trends in ethical and legal frameworks for the use of human biobanks" 2007 *Eur Respir J* 30:373–382.

5.4.1 Table 2: Types of benefits that accrue to various stakeholders in a research project⁸⁴⁷

	Health benefits	Commercial/Economic benefits	Scientific benefits
Individual level	Designer drugs and other individual aspects of personalised medicine	Profits to investors	Non-instrumental knowledge: Development of science and gaining of new information as a value itself, regardless of whether it is useful to humans
Communal level	Relief to disease-affected populations, etc.	Non-medical benefits to communities Capacity building	
National, state level	Efficient health care services, policy planning, etc.	Development of biotech and related sectors, new jobs, etc.; capacity building	
Global level	Eradication of disease		

Some commentators regard Table 2 above as not exhaustive of the types of benefits that could be negotiated in health research and that it is not possible to pass judgement on the deliverability of the suggested benefits. 848

Several models of benefit sharing frameworks dominate ethical debates. The next section will turn to some of the key models.

5.4.2 Reasonable availability model

In the 1990s, the reasonable availability model for benefit sharing was proposed to prevent the exploitation of HIV research participant trials in LMICs.⁸⁴⁹ The post study obligation of the research would be to provide post-trial vaccines and other therapeutics. The weakness of this model was that beneficiaries had to wait for benefits that might not emerge. A further criticism was the issue of who should choose what suitable benefits were to be provided

⁸⁴⁷Simm K "Benefit sharing frameworks –justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 25 May 2023).

⁸⁴⁸Simm K "Benefit sharing frameworks–justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit available at: www.uclan.ac.uk/genbenefit (Date of use: 25 may 2022).

⁸⁴⁹ Participants in the 2001 "Conference on ethical aspects of research in developing countries moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits'" 2004 Hastings Center Report 34(3):17–27.

to the LMIC host.⁸⁵⁰ Most people in LMICs have poor access to good healthcare and cannot afford to wait long periods for interventions, unlike their counterparts in HICs.

5.4.3 Fair benefits model

To address the weaknesses in the reasonable availability model, the fair benefits model was proposed. This model challenges the notion that benefits can only arise from the results of a study and proposes that the host community could negotiate for benefits beyond those tied to the results of the study.⁸⁵¹ This model has been criticised for trying to turn research into an auction whereby wealthy sponsors from HICs would choose to conduct research in communities in LMICs with low negotiating power.⁸⁵²

Both the reasonable availability model and the fair benefits model concur that the underlying purpose of research is to create generalised knowledge and that benefits should accrue to all who participate in research.⁸⁵³ However, both models fail to address the issue of how to determine a fair benefit and who is responsible for the oversight of protecting participants in LMICs.⁸⁵⁴

5.4.4 Private benefit sharing model

The private benefit sharing model mentioned in Chapter 2 of this thesis⁸⁵⁵ applies to private biobanks or tissue banks, whereby private biotechnology companies act as brokers of tissue and health data for many researchers.⁸⁵⁶ This system is problematic when the regulation of biobanks is inadequate, as in SA,⁸⁵⁷ because HBMs and medical data are stored in perpetuity, without any indication of who may benefit from the use and re-use of these samples in the absence of binding legal frameworks. The problem with this model is

⁸⁵⁰ Participants in the 2001 "Conference on ethical aspects of research in developing countries moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits'" 2004 Hastings Center Report 34(3):17–27.

⁸⁵¹Participants in the 2001 "Conference on ethical aspects of research in developing countries moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits'" 2004 *Hastings Center Report* 34(3):17–27.

⁸⁵²London AJ and Zollmann K "Research at the auction block: problems for the fair benefits approach to international research" 2010 Hastings *Center Report* 40(4):34–45.

⁸⁵³Macpherson CC "Research ethics guidelines and moral obligations to developing countries: capacity-building and benefits" 2019 *Bioethics* 33:389–395.

⁸⁵⁴Macpherson CC "Research ethics guidelines and moral obligations to developing countries: capacity-building and benefits" 2019 *Bioethics* 33:389–395.

⁸⁵⁵ Chapter 2, Section 2.5.1 of this thesis.

⁸⁵⁶Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12): 1180–1184.

⁸⁵⁷Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impendiments to accessing healthcare" 2019 *SAJBL* 12(1):27–31.

typically exemplified by allegations of the misuse of African DNA in the UK;858 specifically, the allegation that a South African research site shared data and samples of DNA collected from indigenous groups in African countries, including South Africa and Botswana, in a legitimate collaboration research agreement with a university in the United States. These samples were then transferred to the Wellcome Sanger Institute for genome analysis as part of the legitimate collaboration.859 Thereafter, the institute entered into negotiations with Thermo Fisher Scientific in an attempt to make gene chips with the African data for commercial purposes without the consent of nor an MTA from their African research partners.860 Both Sanger and Thermo Fisher subsequently denied these allegations although one university in South Africa has since demanded the return of its DNA samples.861

5.4.5 Altruistic model of benefit sharing

The altruistic model of benefit sharing, described in Chapter 2 of this thesis⁸⁶² is driven by the notion that all health research should be altruistic and should be for the benefit of all.⁸⁶³ This model is designed in line with the healthcare systems of HICs that benefit all, unlike the disproportionate health systems of LMICs. To propose such a model in LMICs would be tantamount to exploitation.⁸⁶⁴ It has also been shown that even in HICs, this model is not always accepted by the population, as was the case in the State of Iceland,⁸⁶⁵ discussed in Chapter 2 of this thesis. In Iceland, a statute on a Health Sector Database (HSD Act) was passed by the Icelandic government, allowing deCODE genetics to construct and operate a centralised database linking medical records with genealogical

⁸⁵⁸Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 *Developing World Bioethics* 21:125–130.

N "Stellenbosch University demands return of DNA samples but UK university hits back" 2019 *Timeslive** https://www.timeslive.co.za/news/south-africa/2019-10-16-stellenbosch-university-demands-return-of-dna-samples-but-uk-lab-hits-back/ (Date of use: 23 May 2023).

⁸⁶⁰Stokstad E "Genetics lab accused of misusing African DNA 2019 *Science* 366(6465):555-556 https://www.science.org/doi/full/10.1126/science.366.6465.555 (Date of use: 23 May 2023).

⁸⁶¹Njilo N "Stellenbosch University demands return of DNA samples but UK university hits back" 2019 *Timeslive* https://www.timeslive.co.za/news/south-africa/2019-10-16-stellenbosch-university-demands-return-of-dna-samples-but-uk-lab-hits-back/ (Date of use: 23 May 2023).

⁸⁶²Chapter 2 Section 2.5.2 of this thesis.

⁸⁶³Simm K "Benefit sharing frameworks–justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 25 May 2023).

⁸⁶⁴Schroeder D, Gefanas E, Chennells R *et al* "Realizing benefit sharing: is there a role for ethics review?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 179–202.

⁸⁶⁵Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

and genetic information. The statute was rejected on the premise that the Icelandic government had exploited its population by claiming all medical records and by using the notion of presumed consent.⁸⁶⁶

5.4.6 Charitable trust model

The charitable trust model proposes a fiduciary relationship in which trustees hold title to property but are obligated to keep or use said property for the benefit of the beneficiary. This benefit sharing model, discussed in Chapter 2,868 has proved successful in cases whereby various rare disease research groups have constructed tissue banks to enable researchers to control the research design, implementation of research results and benefit sharing. The United Kingdom Human Tissue Authority (HTA) also benefits from this model, as it acts as the trustee of all human tissue or human biological materials collected in the UK.

I believe that a benefit sharing model, based on the model of the charitable trust model, could be adopted in South Africa. This framework recognises the various stakeholders that are part of a research project at different levels of society while simultaneously acknowledging that it is possible to have different types of fair benefits at each stakeholder level, even in the absence of a final, tangible benefit.

In recent years, new models of benefit sharing arrangements have been proposed by various stakeholders participating in health research in LMICs. The Human Hereditary and Health in Africa (H3Africa) Initiative is a North–South collaborative initiative that empowers African researchers to be competitive in genomic sciences, establishes and nurtures effective collaborations among African researchers on the African continent and generates unique data that could be used to improve both African and global health.⁸⁷¹ The H3Africa project is a joint initiative of the United States National Institutes of Health, the Wellcome

⁸⁶⁶Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

⁸⁶⁷Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

⁸⁶⁸ Chapter 2 Section 2.5.3 of this thesis.

⁸⁶⁹Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

⁸⁷⁰Human Tissue Authority: The regulator for human tissue and organs https://www.hta.gov.uk/policies/human-tissue-act-2004 (Date of use: 23 May 2023).

⁸⁷¹Human Heredity and Health in Africa https://h3africa.org/ (Date of use: 24 May 2023).

Trust and the African Society of Human Genetics⁸⁷² that endorses capacity building as the primary obligation of its research agenda.⁸⁷³ The H3Africa's envisioned benefit sharing model moves away from the premise espoused by other benefit sharing models, namely that the primary goal of research is to generate generalisable knowledge, towards the notion that the primary goal of research should be capacity building.⁸⁷⁴

5.4.7 H3Africa's capacity-building model

The H3Africa project maintains that supporting infrastructure and training researchers is the main goal of the project to support genomics research for the benefit of African populations and societies.⁸⁷⁵ The model proposes that H3Africa achieves its primary research objectives of building African capacity and infrastructure via five mechanisms: providing exclusive grants to researchers affiliated with African institutions; training and educating young African researchers; maintaining the leadership of H3Africa among African researchers; prioritising data analysis and publication to African researchers and developing African bioinformatics infrastructures and networks.⁸⁷⁶

While there is strength in the proposed model, it fundamentally rejects the premise that the primary purpose of research in LMICs is to create generalised knowledge; instead, it embraces the notion that the primary and immediate goal of all research is capacity building.⁸⁷⁷ Its weakness resides in the uncertainty of sustaining research beyond the NIH-Wellcome Trust funding period.⁸⁷⁸ It is worth noting that Dauda and Joffe question whether this H3Africa benefit model fully represents all the stakeholders involved in the H3Africa project with regard to benefit sharing.⁸⁷⁹

⁸⁷²Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–170.

⁸⁷³Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–170.

⁸⁷⁴Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–7170.

⁸⁷⁵H3Africa Working Group "Harnessing genomic technologies toward improving health in Africa: opportunities and challenges" 2011 H3Africa White Paper. Washington, DC: National Institutes of Health.

⁸⁷⁶Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *developing world bioethics* 18:165–170.

⁸⁷⁷ Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *developing world bioethics* 18:165–170.

⁸⁷⁸Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 developing world bioethics 18:165–170

⁸⁷⁹ Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *developing world bioethics* 18:165–170.

Interestingly, H3Africa projects are mostly collaborations between the most advanced academic institutions in Africa that already boast great African researchers and some of the best infrastructure in Africa. Only the primary investigator of a project needs to be associated with an African institution; further, the priority of data analysis and publication to African researchers is exclusive for a brief period. The aim of H3Africa is noble, but if young African researchers are taken to HICs that have the best infrastructure in the world for no more than 24 months and then return to their LMICs, their development and research are often stunted and it matters not if they have priority with data analysis and publication. H3Africa achieves capacity building but not at the level where this could be their primary goal for research.

Table 3 below details the benefit sharing models that are most likely to be in use in LMICs, as pointed out by Dauda and Joffe.⁸⁸⁰

5.4.8 Table 3: Benefit sharing models limited to LMICs881

Benefit model	Primary goal of research	Conception of benefits
Reasonable availability	Generalisable knowledge	Products from the research are appropriate benefits to the community
Fair benefit	Generalisable knowledge	Community can negotiate for fair benefits, including but not limited to products derived from the research, human and infrastructure capacity, etc.
H3Africa	Capacity building	Capacity building for research and health improvement is the intrinsic purpose of programme

Table 2 illustrates how the H3Africa framework differs from other models that are used in research in LMICs. The H3Africa framework moves away from the traditional notion that the primary goal of all research is to obtain generalisable knowledge to the primary goal of research for the sake of capacity building.

153

⁸⁸⁰Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 developing world bioethics 18:165–170

⁸⁸¹Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–170.

5.4.9 Ethical benefit sharing model that includes multiple stakeholder benefits

Bedeker *et al*⁸⁸² suggest an alternate framework to promote ethical benefit sharing in health research that attempts to cater to the multiple stakeholders that are usually involved in research projects. Acknowledging that it is often difficult to operationalise benefit sharing within research programmes, they propose a two-dimensional framework that would enable research stakeholders to identify opportunities in research programmes.⁸⁸³ In the first dimension of the model, a socioecological model is used to identify stakeholders involved in the research project at the micro-, meso- and macro-level. The second dimension identifies nine different types of benefit sharing that may be achieved during a research programme.⁸⁸⁴

This proposed framework by Bedeker *et al*⁸⁸⁵ recognises the fact that a research project includes a multi-faceted team and that benefits to various stakeholders can accrue in varying forms at different times during the research project, as shown in Table 4 below. This suggested framework is progressive as it identifies to whom, when and what kind of benefits could be accrued during a research project. The framework suggested by Bedeker *et al* encourages using English as a common language that all participants involved in a research project can understand, since scientific language may not appeal to all the stakeholders.⁸⁸⁶ The current study concurs with this suggestion, as some scientific terminology can be hard to grasp, leading to a lack of comprehension by participants.

⁻

⁸⁸²Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

⁸⁸³Bedeker A, Nichols M, Allie T et al 2022 BMJ Global Health 7:e008096.

⁸⁸⁴Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

⁸⁸⁵Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

⁸⁸⁶Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

5.4.10 Table 4: Summary of the elements of the two-dimensional benefit sharing framework⁸⁸⁷

Dimension 1: Stakeholders Macro-level stakeholders

These stakeholders generally make decisions and provide services at a national or higher level and include global and regional organisations, national organisations, governments, policymakers, regulatory bodies and organisations, legislators and public health officials. For example, the WHO, the African Union or the South African government.

Meso-level stakeholders

These stakeholders may impact provincial, state, municipal or institutional levels. They may include some larger community groups. Examples include academic institutions, ethics review boards, particular population groups, provincial governments, provincial health services, funders, educators, biotech or private health service companies, institutions and facilities.

Micro-level stakeholders

These are individuals, families or small community groups who operate at a personal or interpersonal level. For example, members of the general population, research participants and their families, researchers and students, healthcare providers, patients, community leaders and community advisory boards.

Dimension 2: Benefit sharing categories

Financial: Direct monetary gain by stakeholders.

Health and wellbeing:

Improved individual and/or population health and

wellbeing of stakeholders.

Infrastructure: Built or logistical infrastructure

that benefits stakeholders.

Equipment: Specialised equipment used by

stakeholders to conduct their work.

Skills capacity: Learnt specialised skills to

undertake tasks and conduct work.

Knowledge: Specialist knowledge that improves how

stakeholders solve problems, undertake tasks and conduct work.

Services capacity: Capacity for stakeholders to provide certain services to the public and general

population.

Career development: Opportunities for stakeholders to establish employment security and/or progress in their careers.

Attribution and recognition: Appropriate

Attribution and recognition: Appropriate acknowledgement and advertising of inputs and contributions from stakeholders.

communication from characteristics.

Some scholars⁸⁸⁸ have advocated for the Convention on Biological Diversity (CBD) to be the main treaty at the global level, to direct the advancement of benefit sharing frameworks and that it should be expanded to include human biological materials like genetic resources. However, to date, despite the 168 signatories to the CBD⁸⁸⁹ and the ratification of the Nagoya Protocol by 136 Parties,⁸⁹⁰ the issue of access and benefit sharing is still a major stumbling block at stalled negotiations on trade-related aspects of intellectually property rights (TRIPs) at the World Trade Organisation (WTO).⁸⁹¹

⁸⁸⁷Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

⁸⁸⁸Chaturvedi S, Crager S, Ladikas M *et al* "Promoting an Inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 153–178.

⁸⁸⁹Convention on Biological Diversity https://www.cbd.int/information/parties.shtml#tab=0 (Date of use: 24 May 2023).

⁸⁹⁰Convention on Biological Diversity https://www.cbd.int/information/parties.shtml#tab=2 (Date of use: 24 May 2023).

⁸⁹¹Chaturvedi S, Crager S, Ladikas M *et al* "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" In Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 153–178.

5.5 Proposed model for benefit sharing framework for health research in South Africa

Unfortunately, although there has been an acknowledgement of the need to include benefit sharing in research programmes, the legal translation and the practical implementation of benefit sharing have been slow globally. With particular reference to South Africa, the challenge is to find a benefit sharing model that tempers (not diminishes) commercial interests, redresses economic imbalance and gives research participants fairer and more active roles in influencing the sharing of benefits. It is also important that ethical principles of the model should accommodate changing scientific principles which involve the sharing and secondary use of human biological resources and data, locally and globally, as well as the associated changes in consent.

The most pertinent questions that must be addressed when evaluating benefit sharing in health research are the ownership of human biological materials, 895 how and to whom benefits should accrue and what comprises a fair benefit.896

This study proposes for the donors of HBMs and/or data to transfer the custodianship of said donations to a trust, as the primary institution that obtains the donated materials, set up to protect and safeguard said materials. The transfer should be executed via a process of informed consent with the encouragement of continuous communication between the two parties to enable effortless negotiations on any future use of the donated materials. The trust is then authorised to sign the current MTA⁸⁹⁷ with an end user (the recipient). The MTA should be seen to benefit the beneficiary of the trust via the addition of a benefit sharing agreement as an annexure to the currently gazetted MTA.

⁸⁹²Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7:e008096.

⁸⁹³Slabbert, MN "The legal regulation of access and benefit sharing with regard to human genetic resources in South Africa" 2011 J. Contmpt Roman–Dutch Law 74: 605.

⁸⁹⁴De Vries J, Munung SN, Matimba A et al "Regulation of genomic and biobanking research in Africa: a content analysis of ethics guidelines, policies and procedures from 22 African countries" 2017 BMC Medical Ethics 18(1): 1–9.

⁸⁹⁵Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

⁸⁹⁶Simm K "Benefit sharing frameworks-justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 24 May 2023).

⁸⁹⁷South Africa Material Transfer Agreement for Human Biological Materials Government Notice 719 Government Gazette 41781 of 20 July 2018.

In order to understand what the general purpose of a trust is, it is necessary to briefly explain the nature and objectives of a trust in South Africa. A trust is a legal entity which is created to hold assests fot the benefit of certain persons or entities. ⁸⁹⁸ In South Africa, trusts are regulated by the Trust Protocol Control Act 57 of 1988 ⁸⁹⁹ and the General Laws (Anti-Money Laundering and Combating Terrorism Financing) Amendment Act of 2022, ⁹⁰⁰ which amended the aforementioned Act. Assets can be transferred into a Trust by sale, donation or upon death in terms of will. ⁹⁰¹ South African law recognises three types of trust, ⁹⁰² namely an "ownership trust" whereby the founder or settlor transfers ownership of assets or property to a trustee(s) (in fiduciary capacity) to be held for the benefit of defined or determinable beneficiaries of the trust; ⁹⁰³ a "bewind trust", where a founder or settlor transfers ownership of assets or property to beneficiaries of the trust, but control over the assets or property is give to the trustee(s); ⁹⁰⁴ and a "curatorship trust", which is similar to a "bewind trust", except that the assets are administered on behalf of a beneficiary who does not have the capacity to manage his/her affairs. ⁹⁰⁵

A valid trust requires, among others, that it is set up in writing in the trust instrument; that there is a clear identification of trust property; the object of trust must be lawful and identified; there must be a binding obligation on the trustees (who are authorised and capable) to administer the trust; finally, the beneficiaries must be clearly identified.⁹⁰⁶

It is important to note that some scholars⁹⁰⁷ have suggested the adoption of a Data Transfer Agreement (DTA), in addition to the MTA, to ethically manage data-sharing

⁸⁹⁸https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-

trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities (Date of use: 10 October 2023).

⁸⁹⁹https://www.justice.gov.za/legislation/acts/1988-57.pdf (Date of use: 10 October 2023).

⁹⁰⁰https://www.gov.za/documents/general-laws-anti-money-laundering-and-combating-terrorism-financing-amendment-act-22-2022 (Date of use: 10 October 2023).

⁹⁰¹https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-

trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities (Date of use : 10 October 2023).

⁹⁰² https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-

trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities (Date of use: 10 October 2023).

⁹⁰³https://www.sars.gov.za/businesses-and-employers/trusts/types-of-trust/ (Date of use: 10 October 2023).

⁹⁰⁴https://www.sars.gov.za/businesses-and-employers/trusts/types-of-trust/ (date of use: 10 October 2023).

⁹⁰⁵https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-

trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities (Date of use: 10 October 2023).

⁹⁰⁶https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-

trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities (Date of use: 10 October 2023).

⁹⁰⁷Mahomed S, Loots G and Stauton C "The role of Data Transfer Agreements (DTA) in ethically managing data sharing for research in South Africa" 2022 S Afr Bioethics Law 15(1):26–30.

research in South Africa. Efforts are underway in South Africa to discuss the development of a national DTA for health research.⁹⁰⁸ As this remains an ongoing consultative process,⁹⁰⁹ it remains to be seen if the DTA should be separate from the MTA or if there is a possibility of integrating the proposed DTA with the current MTA.⁹¹⁰ At the time of writing of this thesis, the National Department of Health has tasked the National Health Research Ethics Council to oversee the revision of the current MTA.⁹¹¹

Annexure A below describes the framework for the benefit sharing agreement that is proposed in this thesis.

5.5.1 Annexure A: Proposed benefit sharing agreement framework

Benefit sharing agreement

The proposed benefit sharing agreement must be signed by the Trust and the Recipient/End User.

Definitions

"Beneficiary" refers to the sample and/or data donor who retains ownership rights to these materials unless the donor chooses to be de-linked from the materials or the materials are destroyed.

"Trust" refers to the primary institution receiving the donated HBMs and/or data.

The Trust may, but is not limited to, an academic institution or such affiliated institutions that are not entirely private entities to allow transparency in the decisions of the board of trustees.

"Board of trustees" refers to a group of individuals who are appointed or elected to manage and have authority over the use of the HBMs and/or data held in trust.

The Trust comprises Trustees who may, but are not limited to:

- Scientists who are health professionals and/or academic research scientists familiar with the topic of health research.
- Sociologists who could act as a liaison between HBM donors and other trustees with specific relevance to the issues of consent and maintaining open lines of communication.
- Prominent community members selected by the community to represent their interests. If the respective
 research is to use HBMs and/or data from a specific minority or marginalised community or group, then the
 people who are chosen by that community or group to represent them should serve on the board of
 trustees.

⁹⁰⁸A webinar titled "Ethically managing data transfers for research in SA" organised by the SA Medical Research Council, in collaboration with the Department of Science & Innovation of South Africa was held on Jun 23, 2022 to deliberate the way forward for a National DTA. https://www.assaf.org.za/wpcontent/uploads/2022/06/22412-Data-Transfer-Webinar-Programme-final.pdf (Date of use: 24 May 2023).

⁹⁰⁹Mahomed S, Loots G and Stauton C suggest that the enforcement of the Protection of Personal Information Act No 4 of 2013 in July 2021 while safeguarding the right to privacy of research participants, hinders other rights which ought to be considered and safeguarded like the right to non-discrimination, the right to dignity and the right to enjoy the benefits of scientific progress, which could be accommodated via a DTA.

⁹¹⁰Mahomed S, Loots G and Stauton C "The role of data transfer agreements (DTA) in ethically managing data sharing for research in South Africa" 2022 *S Afr Bioethics Law* 15(1):26–30.

⁹¹¹Discussed with the NHREC during a quarterly meeting in 2022 by the Chief Director of Medical Forensic Services, Ms Pakiso Netshidzivhani, communicated to the REC community in South Africa.

- An elected donor to serve on the board of trustees to promote transparency of negotiation processes.
- The functions of the Trust are, but are not limited to:
- i. Forming a Research Ethics Committee (REC) to vet proposed research projects from stakeholders wishing to utilise the HBMs held by the Trust.
- ii. The election of a representative who would be the responsible person to negotiate its terms.
- ii. The explanation and education of the HBMs donors regarding the use of their donated materials, the research objectives, expected research outcomes and potential benefits if any, that would accrue to the donors and/or their communities. Using language or visual aids that are appropriate for flawless communication with the donor.
- iii. In consultation with all stakeholders of a research project, discuss the fairness of the level or amount of benefit proportional to the specific risk taken.
- iv. In consultation with the donors of the samples, determine whether to link or de-link samples and document the decision. This will enable donors to withdraw their samples at any time if they are linked.
- v. Advising the HBM donors of their rights and the right to withdraw their consent for any research project using their linked donated samples.
- vi. Negotiating and agreeing to MTAs that are favourable to the beneficiary of the Trust with entities that would use the donated human biological materials.
- vii. Having a continuous relationship with donors as opposed to one-off signatory agreements, which would enable the Trust to easily obtain secondary consent if new projects arise.
- viii. Communicating with the donor/beneficiary about any possible commercialisation of products arising from the use of their donated samples in specific research projects and explanation of how this would benefit the donor, if at all.
- ix. In the event of commercialisation, the Trust can enter into another MTA with the specific stakeholder on behalf of the beneficiary, which would map the way forward as to what specific benefits would accrue, to whom and when.

The range of benefits listed below may accrue among different stakeholders during a proposed research project and it is the duty of the Trust to negotiate in good faith with the recipient of the HBMs for the benefit of all. In this context:

'Individual' level refers to individuals, small specific groups of people or family members.

'Community' level refers to larger collections of persons who have certain interests or attitudes in common like academic institutions, specific population groups and/or provincial governments or provincial health departments.

'National level' refers to bodies that make decisions that apply to the entire country and affect all South African peoples.

'Marginalised communities' refers to a group of people who have historically been disadvantaged by the State.

The End User/Recipient of the HBMs should specifically state which benefits will accrue and to whom:

Expected benefits		No
Individual level		
Monetary compensation (direct)		

•	Monetary compensation (in stipend, grocery or voucher, specify)	
•	Access to healthcare, knowledge	
•	Access to personalised medicine	
•	Inclusion in research papers	
•	Capacity development	
•	Royalties	
•	Other benefits (specify)	
Commi	unity level	
•	Monetary (funding or grants for specific community projects)	
•	Infrastructure development	
•	Capacity development	
•	Advanced equipment and training in its usage	
•	Strengthening of local RECs via knowledge transmission	
•	International collaborations and recognition of local researchers in	
	publications	
•	Technology transfer	
•	Personalised Medicine for a specific group	
•	Milestone Payments	
•	Royalties	
•	Other benefits (specify)	
NI di		
Nationa		
•	Access to resources	
•	Infrastructure development	
•	Creation of Centers of Excellence for Research	
•	Information	
•	In-country Manufacture of specific drugs and/or vaccines	
•	Technology transfer	
•	Specialist equipment with local training of service engineers	
•	Specialist in Academia and Collaboration research	
•	Policy Guidance	
•	Other benefits (specify)	
	alised Communities	
•	Monetary (funds or grants for the community)	
•	Infrastructure development	
•	Personalised medicine	
•	Royalties	
•	Other benefits (specify)	

In the case of monetary compensation, the expected amount must be listed.

5.6 Conclusion

Health research collaborations between HICs and LMICs are often characterised by inequitable and/or neo-colonial practices and power imbalances that plague ongoing research.⁹¹² While it stands true that the overall goal of health research is to attain global health and wellness for all, 913 health research using HBMs cannot occur in an exploitative environment that takes unfair advantage of people's vulnerabilities.914 Benefit sharing should be a tool for guarding against exploitation and not the basis of a strategy to address urgent global health needs or resolve inherent issues of global distributive justice. 915

With regard to health research using HBMs, Africa is in a unique position because the African genome is genetically more diverse than those of other races, meaning that African human genes may contain undiscovered disease-causing variants. 916 The nature of genomic research and biobanking research challenges traditional methods of research because it requires greater openness, the sharing of resources, collaborations between scientists from the Global North and Global South and the re-use of substantial amounts of samples and data. 917 The requirements of this specific health research challenge the traditional ideas of informed consent and data privacy.918 There is also the persistent question of the ownership of HBMs and in South Africa, this issue remains unsettled and non-specific.⁹¹⁹ The enactment of the Protection of Personal Information Act (POPIA) ⁹²⁰

⁹¹²Bedeker A, Nichols M, Allie T et al "A framework for the promotion of ethical benefit sharing in health research" 2022 BMJ Global Health 7:e008096.

⁹¹³Hollis A, Pogge T and Schroeder D "Beyond benefit sharing: towards realising the human rights to health" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 203-216.

⁹¹⁴Millum J "Sharing the benefits of research fairly: two approaches" 2012 *J Med Ethics* 38:219–223.

⁹¹⁵Hollis A, Pogge T and Schroeder D "Beyond benefit sharing: towards realising the human rights to health" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 203-216.

⁹¹⁶African populations have higher numbers of average variant single nucleotide polymorphisms (SNP) sites at 3.3 million per individual, compared to Europeans with 2.9 million and Japanese/Chinese with 2.8 million per individual. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1462113/ (Date of use: 26 May

⁹¹⁷De Vries J, Munung SN, Matimba A et al "Regulation of genomic and biobanking research in Africa: a content analysis of ethics guidelines, policies and procedures from 22 African countries" BMC Medical Ethics 18.1 (2017): 1-9.

⁹¹⁸Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 Developing World Bioethics 21(3):125-

⁹¹⁹Mahomed S "Human biobanking in developed and developing countries: an ethico-legal comparative analysis of the frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 Cambridge Quarterly of Healthcare Ethics 30(1):146-60.

⁹²⁰ POPIA was enacted on 1 July 2020. The Act enables the Constitutional right to privacy. All data collected under medical practice and research, unless anonymised, is protected by this Act.

has raised the question of whether genetic samples can be completely anonymised by researchers.⁹²¹

In this chapter, I have argued that the establishment of a charitable trust for HBMs in the context of benefit sharing in health research could address some of these pertinent questions. In terms of this model, academic medical centres and/or research institutions would cease to be brokers of the HBMs and instead become custodians of the samples. The trust will promote compliance with data privacy and informed consent requirements without compromising its value as an information-rich HBM supplier. The trust could serve as a scientific reserve and social cohesion agent among the various stakeholders involved in specific research projects. Moreover, a trust will be in the ideal position to create and facilitate continuous communication channels with the donor community, researchers, policymakers and teaching hospitals for fostering trust in health research and its benefits for all stakeholders involved.

-

⁹²¹Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 *Developing World Bioethics* 21(3):125– 13

⁹²²Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

CHAPTER 6

CONCLUSION AND RECOMMENDATIONS

6.1 Introduction

The significant increase in health research in low- and middle-income countries (LMICs)⁹²³ over the past few decades, sponsored (in part) by institutes in HICs and coupled with the historical exploitation of research participants in the African setting,⁹²⁴ demands that the issue of benefit sharing between HICs and LMICs that collaborate in research to be taken seriously. This explosion in health research in LMICs, combined with the global commercialisation of health research products sometimes, via privately funded businesses,⁹²⁵ calls for the issue of benefit sharing in research using human biological resources to be addressed.

There have been several cases of exploitation in research, as demonstrated in Indonesia. During the height of a global avian flu pandemic between 2005 and 2006, Indonesia provided the World Health Organisation (WHO) with avian flu samples the were subsequently used without the requisite permissions. The WHO shared the Indonesian samples with various US laboratories and Hong Kong University for vaccine production purposes. Subsequently, a rift emerged between the WHO and Indonesia when Indonesia discovered that non-Indonesian researchers and their firms in HICs were filing patent applications using the Indonesian virus samples without permission from the

_

⁹²³Lairumbi G, Parker M, Fitzpatrick R and English M "Forms of benefit sharing in global health undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities in Medicine* 7(1):1–8.

⁹²⁴Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136– 138.

⁹²⁵Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 AMA Journal of Ethics 22 (1):28-35.

⁹²⁶Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

⁹²⁷Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

⁹²⁸Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

Indonesian government.⁹²⁹ The use of HBM samples from LMICs by private firms to develop therapeutics such as vaccines, usually in HICs, which are then sold back to LMICs—often at inflated prices or not made available at all—is a reality which requires scrutiny.

The issue of benefit sharing in health research remains controversial yet pertinent. There is an argument that all health research ought to be altruistic in the quest for generalised knowledge, 930 but it cannot be ignored that health research conducted in (LMICs) like South Africa is frequently undertaken in communities that have previously been exploited by HICs through colonialism or discriminatory and segregation systems like Apartheid. 931

The arguments against benefit sharing in such communities become invalid when regarding the idea of benefit sharing in LMICs as justice in exchange, to address the wrongs of unfair research collaborations. This research has long been and is being perpetuated in the form of 'helicopter' or 'parachute' research and the ongoing controversies around human biological samples from these regions, including South Africa, illegally being used for research in HICs. 932 The move towards equitable benefit sharing is gaining momentum in the international arena, as it appears in many national and International ethics guidelines since 2005, 933 such as the Convention of Biological Diversity (CBD), the HUGO Committee Statement on benefit sharing, the Declaration of Helsinki (WMA 2008) and the UNESCO Universal Declaration on Bioethics and Human Rights (UNESCO 2005). 934 Despite these guidelines, the unjust use of HBMs originating from LMICs by HICs is still ongoing 935 because national laws decline to address what constitutes an appropriate benefit within the ever-evolving field of human health research and biotechnology.

⁹²⁹Cook Lucas J, Schroeder D, Arnarson G, Andada P, Kimani J, Fournier V and Krishnamurthy M "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 95–127.
930D and Leffe C "The horseft benefit sharing vision of HOAfrica" 2000. Benefit as Month Biography 100.

⁹³⁰Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170. ⁹³¹Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440.

⁹³² Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 South African Medical Journal 106(2):136–138.

⁹³³These Guidelines are covered extensively in Chapter 2 of this thesis.

⁹³⁴Schroeder D and Lucas CJ "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) *Benefit sharing from biodiversity to human genetics* (Springer Science 2013) 217–230.

⁹³⁵Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 *South African Medical Journal* 106(2):136–138.

This final chapter addresses the questions raised as study objectives and provides recommendations, where appropriate, based on the conclusions made in the relevant chapters. The study also proposes a draft benefit sharing agreement which could accompany MTAs whenever human biological materials are transferred for the purpose of research. This draft agreement appears as Annexure A to chapter 5 above.

6.2 Recommendations

6.2.1. Benefit and benefit sharing in the context of health research: underlying ethico-legal norms

The concept of benefit sharing remains controversial for many reasons. One such reason is the lack of clarity around what defines a benefit and what a benefit should comprise. Regarding non-human genetic resources, the Nagoya Protocol suggests that a benefit may include both non-monetary and monetary benefits.⁹³⁶ This thesis has determined that in human health research, the intricacies of what a benefit truly entails are seldom expressed.⁹³⁷ The notion of fair benefit in the context of human biological material is often put forward, but in truth, no concise definition of what this phrase essentially entails, exists. This lack of a comprehensive definition means it is often used to suit the narrative of a specific situation.938

I propose that to answer the notion of fair benefit there needs to be an agreement of what the benefit entails between a resource provider, a providing institution and a recipient institution. To achieve this, the providing institution of any HBMs would become a Trustee of the materials for the HBM donor/beneficiary and negotiate in good faith for the fair benefits with the HBM recipient.

Altruistic research is not possible if it has been conducted via unfair collaborations where the burdens and benefits of research are not equally shared and in LMICs that historically

⁹³⁶Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation, Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (date of use: 24 May 2023).

⁹³⁷See Chapter 2 of this thesis.

⁹³⁸Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 BMC Medical Ethics 14(1):1-8.

have been subject to exploitation and collaborators who lack integrity and display cultural insensitivities. 939

The failure to conclusively define the term 'benefit sharing', with whom and when said benefits should accrue in a research project has led to some collaborators taking advantage of the situation by ignoring the concept of benefit sharing or not sharing benefits fairly. Paper I propose that for the sake of uniformity, the definition of benefit sharing used in the Convention of Biological Diversity (CBD) framework for access and use of genetic resources should be adopted for use in the context of international human health research, as it incorporates the notion that benefit sharing with resource providers is necessary. This can be extended to apply to research participants and their respective communities, as the definition adheres to the philosophical principle that benefit sharing is meant to achieve justice in exchange for providing resources.

There are no binding legal norms that inform the concept of benefit sharing in the context of human health research using HBMs. This probably stems from the law's uneasiness with the commodification of the self and the obvious unequal bargaining powers that are evident between collaborators from HICs and those from LMICs. This often emerges in instances where research is funded by institutions from the Global North. However, as discussed in this study, there are a multitude of guidelines that address benefit sharing worldwide. 942

Specific to South Africa, the Department of Health (DoH) National Ethics Guidelines stipulate that it is expected that research participants and their communities would benefit from participating in research at some point, ⁹⁴³ yet the Guidelines do not elaborate on when and how this should occur. Although the HPSCA Guidelines also call for benefit sharing with research participants and the balancing of burdens and benefits, they allude to the

⁹³⁹Moodley K and Singh S "It's all about trust": reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

⁹⁴⁰Xiong L and Wang Y "Harvard University's genetic research in China is illegal" 2002 *Outlook Weekly* 15:48–50.

⁹⁴¹According to the CBD framework, benefit sharing is defined as "the action of giving a portion of advantages or profits derived from the use of genetic resources or traditional knowledge to resource providers in order to achieve justice in exchange".

⁹⁴²Chapter 2.3 of this thesis speaks to these Guidelines, the CBD, the Declaration of Helsinki and the UNESCO Declaration on Bioethics and Human Rights and the HUGO Committee Statement on benefit sharing.

⁹⁴³Department of Health National Ethics Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

need for benefit sharing without further direction.⁹⁴⁴ The National MTA template requires benefit sharing to be negotiated via an agreement between research collaborators prior to the commencement of a research project.⁹⁴⁵ While there is recognition of the concept of benefit sharing and what a benefit entails, the list needs to be expanded and explain to whom these benefits should accrue when translated into practice.

The idea that the notion of benefit sharing defeats altruism in human health research conducted in LMICs cannot stand. If the injustices of the past against historically marginalised peoples are to be corrected, then benefit sharing must occur. The notion that benefit sharing should not occur because of fears about the undue inducement of vulnerable research participants in LMICs is intrinsically condescending because it negates the checks and balances that exist in the ethical oversight provided by Research Ethics Committees (RECs). Moreover, this belief perpetuates the exploitation of research participants and their communities because it would heighten the risk of such individuals participating in research without benefiting from it. For everyone in the world to attain the highest possible state of health for all—which is a human right—then it is only fitting to promote fair collaborations between HICs and LMICs, whereby the burdens and benefits of the research are equally shared.

6.2.2 Historical concept of benefit sharing

Globally, the term benefit sharing has been used as a technical term in human and non-human genetic research to explain the compensation and/or reward for the provider of the resource by the user of the resources. ⁹⁴⁶ In terms of justice, a fair benefit would fall under the principles of justice. ⁹⁴⁷

The notion of benefit sharing emerged from the principle of *distributive justice* that has its foundations in the concept of *res communis*.⁹⁴⁸ This concept proposes that all resources obtained from common heritage belong to all humankind and must be shared equally to help balance the inequity between HICs and LMICs and clarifies that a benefit need not

⁹⁴⁴HPCSA General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s_for_Health_Researchers.pdf (Date of use: 25 May 2023).

⁹⁴⁵Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁹⁴⁶Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioethics* 6: 135–143.

⁹⁴⁷Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

⁹⁴⁸Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing world bioeth* 18:165–170.

always be tangible. ⁹⁴⁹ This study concurs with this viewpoint since technology transfer and capacity building are benefits that are very useful in LMICs.

In the context of the access to and sharing of non-human genetic resources, the CBD—though acknowledging that the conservation of biodiversity is a universal concern—holds that the genetic resources of individual countries are sovereign, invoking benefit sharing as a principle of international law in the Nagoya Protocol, a supplementary agreement to the CBD, to promote fair and equitable sharing arising from the use of a nation's resources. This exchange, which operates for the mutual benefit of the resource provider and the resource user, is *justice in exchange*, which gave rise to the notion of fair benefit sharing.

The concept of benefit sharing emerged towards the end of the 20th century and purports to serve to enact justice. PEC Recognising that scientific advancement without benefit sharing is unjust has developed as the foundational principle informing the philosophical principle of benefit sharing. However, even though the idea of benefit sharing exists, it is not always realised and when this does not transpire, it could be argued that injustice and the possible exploitation of a party have occurred.

As regards human biological resources, no precedent for the concept of benefit sharing in human health research exists, presumably because this defeats the fundamental notion that altruism should drive all health research, compounded by the murky topic of the ownership of human biological resources.⁹⁵⁴ The argument against benefit sharing also points out that it may lead to the undue inducement of participants in a study and that the

_

 ⁹⁴⁹Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing world bioeth* 18:165–170.
 ⁹⁵⁰Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011 http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (date of use: 23 May 2023).

⁹⁵¹Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 9–32.

⁹⁵²The notion of benefit sharing has been accepted in many national and international guidelines because of its inclusion in the CBD, the Declaration of Helsinki (WMA 2008) and the UNESCO Universal Declaration on Bioethics and Human Rights (UNESCO 2005).

⁹⁵³Schroeder D and Cook Lucas J "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 217–230.

⁹⁵⁴Mahomed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8(Suppl 1):60–64.

more vulnerable a population, the more likely this undue inducement will result in exploitation of the population. 955

The law dictates that the human body and its parts are *res extra commercium* and cannot be commodified. This has led to many challenges for the concept of benefit sharing in human health research. However, many organisations have emphasised the need for benefit sharing in health research with UNESCO, the HUGO Committee, the WMA and CIOMS all pushing the agenda for benefit sharing, albeit without elaborating on how and with whom benefit sharing should occur.⁹⁵⁶

In relation to benefit sharing for non-human biological resources in South Africa specifically, the country is a signatory of the Nagoya Protocol and is bound by international law via this mandate, which unambiguously regulates the access to and benefit sharing of non-human genetic resources. The Biodiversity Act 10 of 2004⁹⁵⁷ elaborates on how a benefit sharing agreement should be designed and what it should contain when using South African non-human biological resources.⁹⁵⁸ The San Code of Ethics⁹⁵⁹ deals directly with how benefit sharing agreements should be drafted when dealing with non-human resources that originate from San communities.⁹⁶⁰

There is no legally binding statute within the South African law concerning benefit sharing for health research using human biological materials, apart from the national Material Transfer Agreement (MTA)⁹⁶¹ template which defines what a benefit may constitute and calls for a benefit sharing agreement between parties before any transfer of HBMs takes place.⁹⁶² Numerous national guidelines speak to the need for benefit sharing in health

⁹⁵⁵Schroeder D and Cook Lucas J "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 217–230.

⁹⁵⁶Section 2.2.3 of this thesis extensively details these statements.

⁹⁵⁷Republic of South Africa. Biodiversity Act 10. Government Gazette 2004.

⁹⁵⁸Chapter 2.2.4 of this thesis elaborates on the Biodiversity Act 10 of 2004.

⁹⁵⁹The San people of South Africa are natives of the Kalahari Desert and have suffered years of marginalisation. Following a disastrous attempt by Phytofarm and Unilever to patent their traditional *Hoodia gordonni* plant without the San communities' consent, San leaders formed the Working Group for Indigenous Minorities in South Africa, which, together with NGOs and the CSIR developed the San Code of Ethics to negotiate benefit sharing arrangements to benefit the San people.

⁹⁶⁰The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

⁹⁶¹Proclamation No 719 Government Gazette 41781 of 20 July 2018.

⁹⁶²Section 2.3 of the National MTA.

research using HBMs, namely the Department of Health's Ethics in Health Research Guidelines, 963 the HPCSA Guidelines and the National MTA template.

6.2.3 Review and evaluation of existing ethical and legal frameworks relevant to South Africa

6.2.3.1 Ownership of HBMs in South Africa

The question of the ownership of HBMs in South Africa is possibly one of the more troubling issues for the implementation of benefit sharing since the notion requires the identification of who should share in the benefits that arise from human health research. Without addressing the issue of the ownership of HBMs, it is almost impossible to address benefit sharing in health research in South Africa. The concept of human dignity outlined in the Bill of Rights of the Constitution of South Africa demands that every human being has intrinsic value and thus, should not be commodified because this violates their dignity and intrinsic worth.⁹⁶⁵ This principle is enforced by the National Health Act (NHA),⁹⁶⁶ which prohibits the sale of and trade of human tissue. The law seems silent on the notion of a person treating their HBMs as property yet becomes vocal on the issue of intellectual property emanating from research using HBMs, supposedly to promote research. This stems from the concept of separated bodily materials being deemed res nullius until brought under the control of the first individual who obtains said HBMs, is capable of gaining proprietary rights and in some cases, a right of ownership if the recipient individual processes the HBM in some way. The commercialisation of products emanating from health research gives rise to complex legal, ethical and philosophical questions because some schools of thought maintain that donors of HBM have ownership rights to their samples in perpetuity, while other scholars maintain that donors cannot be given proprietary rights to their HBM samples as this would enable the commodification of the samples and, in turn, hinder research.967

⁹⁶³Department of Health National Ethics Guidelines http://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 23 May 2023).

⁹⁶⁴HPCSA General Ethical Guidelines for health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guideline s for Health Researchers.pdf (Date of use: 26 May 2023).

⁹⁶⁵The Constitution of South Africa 1996.

^{96661 0}f 2003, Chapter 9.

⁹⁶⁷Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

The question of the ownership of HBMs needs to be addressed or clarified; moreover, to promote health research further, it must happen in a context where cultural values are respected to foster trust with communities that donate HBMs for research. This researcher proposes that for fair benefit sharing to occur, all those involved in research ought to benefit and that the notion of ownership should not hinder the sharing of benefits among all stakeholders. This could be addressed by a custodianship relationship between the HBM donor and the first recipient of the donated HBMs, whereby the recipient becomes a charitable trust that acts in the best interest of its beneficiary/HBM donor.

6.2.3.2 National Health Act 61 of 2003 and relevant Regulations

In South Africa, all health research is governed by the NHA.⁹⁶⁸ The NHA and its various Regulations are not streamlined when it comes to certain definitions, for example, of tissue and human biological materials that are relevant to human health research. Consequently, this leads to a failure to legally classify specific HBMs. If a unified legal definition remains non-existent, the creation of commercial products using human tissue may flourish.

There is no mention of benefit sharing in the Regulations relating to the use of HBMs. 969 These Regulations stipulate that informed consent is required for the removal of HBM from living and deceased persons for research while not addressing the issue of consent relating to any future use of the extracted HBMs. This abstention to articulate the matter means that the ownership of the extracted HBMs is left open-ended; it complicates benefit sharing and furthermore, does not clarify to whom benefits should accrue.

The Regulations Relating to the Artificial Fertilisation of Persons⁹⁷⁰ expressly provide for the ownership of gametes and the product of the gametes, namely that whoever has possession of the gametes used in the process of the artificial fertilisation is considered the owner of the HBMs. This is an anomaly in the legal framework as this is the closest South African law comes to acknowledging proprietary rights in HBMs, ostensibly to make the process of artificial fertilisation less cumbersome.

The concept of custodianship is introduced in the Regulations Relating to Stem Cell banks, 971 whereby informed consent is a prerequisite for the donation of stem cells to a

⁹⁶⁸⁶¹ Of 2003, Chapter 9.

⁹⁶⁹Proclamation No 11 Government Gazette 35081 of 27 February 2012.

⁹⁷⁰Notice 1165 in Government Gazette 40312 of 30 September 2015.

⁹⁷¹Notice 183 in Government Gazette 35099 of 2 March 2012.

biobank. In turn, donors are anonymised but said stem cells cannot be used for any purpose without the authority of the National Health Department via the Director General (DG). The DG then appoints health officers to act on its behalf, essentially granting them custodianship of the stem cells.

In the Regulations Regarding the General Control of Human Bodies, Tissue, Blood, Blood Products and Gametes, ⁹⁷² the prerequisite for informed consent to obtain the donation of specific HBM from a living donor is reiterated, i.e., the donor retains no rights to the donated HBM. Donations from the bodies of deceased persons require the recipient of the HBM to use it exclusively for the purposes it has been donated for, suggesting that the deceased donor retains some control over the HBMs.

The NHA and its Regulations do not address consent for the secondary use of samples nor does it speak to benefit sharing since this would require identifying to whom benefits should accrue when using HBMs in health research. Until the issues of ownership and informed consent to the secondary use of samples are unambiguously addressed in the legal framework, the fair and equitable implementation of benefit sharing will not be possible.

6.2.3.3 National Material Transfer Agreement for Human Biological Materials

The MTA's⁹⁷³ recognition of donors' ownership of HBMs until such HBMs are destroyed, as well as an acknowledgement of the provider⁹⁷⁴ of the materials as the custodian of the materials are legally pre-eminent in South Africa.⁹⁷⁵ This clause in the MTA⁹⁷⁶ is significant because it introduces the notion of custodianship between the donor of the HBM and the first recipient of the donation, who becomes the provider of the HBM, making it easier to identify to whom and with whom benefits can be shared when using donated HBM. It is also the first legal document to define the terms *benefit* and *benefit sharing*.⁹⁷⁷ The MTA further points out the need for a benefit sharing arrangement that must be agreed upon by

⁹⁷²Notice No. R. 180 in Government Gazette 35099 of 2 March 2012.

⁹⁷³Proclamation No 11 Government Gazette 41781 of 20 July 2018.

⁹⁷⁴Provider being the providing institution of the HBM.

⁹⁷⁵Proclamation No 11 Government Gazette 41781 of 20 July 2018.

⁹⁷⁶ Section 3.3 of the MTA 2018.

⁹⁷⁷In the MTA, a benefit is described as "amongst others, the sharing of information; use[;] of research results; royalties; acknowledgement of the Provider as the source of the Materials; publication rights; transfer of technology or the process materials; and capacity building", while benefit sharing is described as "the process or act of sharing in the benefits that derive from the Project in a manner that is fair and equitable".

the provider and recipient⁹⁷⁸ before the HBMs are transferred. However, as solely a template, the SA MTA might be changed by parties to suit their specific contexts, provided that the principles within it are respected. It is unclear how this would be monitored in the case of benefit sharing. Thus, the need for a South African benefit sharing agreement which sets out actual clauses that must be considered during the negotiation process is crucial. Annexure B of the national MTA provides for a section of the benefit sharing agreement that may be negotiated between the provider and the recipient of the HBM.⁹⁷⁹ There is no provision for a benefit sharing arrangement between the donor and the provider.

The MTA has been criticised by scholars who suggest it has failed to align with the broader legal environment in clarity and practicality. Nevertheless, other academics point out that it is a step in the right direction. 981

6.2.3.4 Protection of Personal Information Act 4 of 2013 (POPIA Act)

The POPIA⁹⁸² was enacted to guarantee all South Africans the constitutional right to privacy. HBM samples do not fall under the category of personal information but the data derived from the samples fall under the category of personal information, meaning the data and information stored in tissue banks and/or biobanks in South Africa are protected under the POPIA. The Act prohibits a responsible party from processing personal information concerning the ethnic origin, health or biometric information of data subjects without informed consent or necessitated by service to the public.⁹⁸³

The Act does not define benefit sharing or mandate its application in the context of health research, but any such research will undoubtedly lead to the need for data-sharing agreements that would provide an opportunity for benefit sharing agreements to be negotiated when using data derived from HBMs held in South African tissue banks and/or biobanks.

11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 26 May 2023)

⁹⁷⁸Recipient being the recipient institution.

⁹⁷⁹Annexure B of the MTA Proclamation No 11 Government Gazette 41781 of 20 July 2018.

⁹⁸⁰Thaldar DW, Botes M and Nienaber A "South Africa's new standard material transfer agreement: proposals for improvement and pointers for implementation" 2020 *BMC Medical Ethics* 21(1):1–13.

⁹⁸¹Labuschaigne M, Dhai A *et al* "Protecting participants in health research: the South African Material Transfer Agreement" 2019 S *Afr Med J* 109(5):353–356.

⁹⁸²POPIA Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 26 May 2023).

⁹⁸³Section 26 and Section 27 of the POPIA. https://www.gov.za/sites/default/files/gcis document/201409/3706726-

To this end, the ASSAf is in the process of developing a code of conduct for the research sector (including the health research sector), which will provide more guidance on how personal data should be processed and shared. Notably, under POPIA, one of the grounds for transferring personal information internationally includes where it is for the benefit of the participant and consent to the transfer is not reasonably practicable to obtain, recognising that if it were reasonably practicable, the data subject would be likely to provide it. This premise requires that the transfer would be for the benefit of each data subject. Although it may be criticised for being almost impossible to achieve when large datasets are transferred for research purposes, the POPIA indeed acknowledges the benefit of data subjects when their personal information is shared through collaborative efforts, thus strengthening the argument for entering into benefit sharing arrangements.

6.2.3.5 Intellectual Property Rights from Publicly Funded Research and Development Act

While the ownership of HBMs is not clear in South African law, the law takes a different position in the case of intellectual property emanating from research using HBMs. The IPR Act applies to publicly funded Research and Development (R&D). BBMs. It stipulates that intellectual property emanating from publicly funded research projects must be identified, protected, used and commercialised for all South Africans. The owner of the intellectual property remains the recipient of the public funds. Benefit sharing arrangements are encouraged between the recipients of the public funds and the intellectual property creators. The IPR Act states that the recipients of public funds can co-own intellectual property with private entities, provided stringent conditions have been met, one of which is the establishment of a benefit sharing agreement. The HBM participants/donors are protected by laws and ethical guidelines that are, however, absent from the negotiation processes that could allow for benefits to accrue directly to them and/or their communities. It would be useful if a benefit sharing agreement is negotiated with the donors of the HBM and the providing institution/local researcher before a REC approve a research project,

⁹⁸⁴Rachel A, Fola A, Dominique A *et al* "POPIA code of conduct for research" 2021 *S Afr J Sci* 117(5-6):1–12.

⁹⁸⁵Section 72 (1)(e) of the POPIA Act https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 26 May 2023).

⁹⁸⁶IPR Act for Publicly Funded R & D https://www.gov.za/sites/default/files/gcis_document/201409/33433675.pdf (Date of use: 26 May 2023).

⁹⁸⁷Section 5 of the IPR Act 51 of 2008.

⁹⁸⁸Section 15(2) of the IPR Act 51 of 2008.

whereby the REC would follow up and ensure that the negotiated benefits are indeed accrued in the event of the development of viable intellectual property.

6.2.3.6 Patent Act 57 of 1978

The Patent Act 57 of 1978 provides for registering and granting letters of patent for inventions to protect intellectual property rights. 989 In the age of biotechnology and advances in technology, many biotechnology and pharmaceutical companies pursue patents for their inventions to protect their proprietary interests and gain profits. The Companies and Intellectual Property Commission (CICP) registers all patents in South Africa. 990 According to Pouris and Pouris, 991 in South Africa, it is often the case that patents in the biotechnology and pharmaceutical industries have been granted to protect the research, development and innovation by South African universities and researchers' locally but due to cost constraints, patents are usually not filed internationally. The registration approach to lodging a patent in South Africa makes it one of the least costly countries to do so since foreign inventors are able to protect their inventions very inexpensively, disadvantaging local researchers. 992

Similar to the IPR Act, proprietary rights are often granted to the intellectual property developer and so, the Patent Act makes no provision for benefit sharing with the donors of HBMs⁹⁹³ for research and development (R&D) in human health research using the results from donated HBMs in patented innovations in biotechnology and engineered products. The patenting of these products can be highly contentious in healthcare research if their creation is vital to saving lives or curing diseases.⁹⁹⁴

In most LMICs, donors of HBMs rarely have access to the patented end products; moreover, when public funds are used to fund the R&D of these patented innovations, it is only expected that some form of benefit sharing arrangement should be in place between the HBM donors and researcher and/or university receiving the public funds.

⁹⁸⁹Act No. 57 Government Gazette 6012 0f 17 May 1978.

⁹⁹⁰Companies and Intellectual property Commission http://www.cipc.co.za (Date of use: 26 May 2023).

⁹⁹¹Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 *South African Journal of Science* 107(11/12):24–33.

⁹⁹²Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 *South African Journal of Science* 107(11/12):24–33.

⁹⁹³Patent Act 57 of 1978.

⁹⁹⁴Mahomed S "An ethico-legal framework for the regulation of biobanks in South Africa" (PhD in Bioethics and Health Law University of the Witwatersrand 2018) Available at: http://wiredspace.wits.ac.za/handle/10539/25331.

6.2.4 Department of Health Guidelines for Ethics in Health Research, 2015

The DoH Guidelines⁹⁹⁵ propose that all health research should show distributive justice and all who participate in health research should benefit from the research.⁹⁹⁶ The Guidelines state that donors of HBM and their communities should, at some point, benefit from the research. The Guidelines do not expand on what benefits, with whom and how benefit sharing should occur.

6.2.5 Existing regional and international benefit sharing frameworks

In the absence of legislation on a benefit sharing model that could be applied in South Africa, this study critically analysed foreign benefit sharing models to determine how best to design a benefit sharing model for use in South Africa. Globally, onlyno legally binding framework Despite these internationally accepted directives, benefit sharing is still a divisive topic between HICs as there are different levels of concern on legislative, ethical and social frameworks. This cannot easily be addressed at a national level as it requires governments to communicate and effect policies that would standardise the receptiveness of guidelines.

To analyse different benefit sharing frameworks, the study compared the relevant frameworks utilised in the United Kingdom and Australia since they are examples of HICs that are frontrunners in health research. Uganda, an LMIC similar to South Africa, was also selected for legal comparison since Uganda has a prominent health research presence globally. These countries also share a colonial history and some similarities in their legal structures and systems.

6.2.5.1 United Kingdom

The UK is a HIC that generally provides high-quality healthcare to all its citizens. The concept of benefit sharing is non-existent in health research involving HBMs because all donations for health research are considered altruistic.⁹⁹⁷ Informed consent is a

⁹⁹⁵DoH National guidelines for Ethics in Health Research https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 26 May 2023).

⁹⁹⁶Section 2.1 of the DoH 2015 Ethics in Health Research.

⁹⁹⁷Section 2 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 26 May 2023).

fundamental principle of the Human Tissue Act (HT Act) of 2004⁹⁹⁸ which regulates the storage, use and disposal of human tissue. The HT Act expressly prohibits the commodification of human material and states that any HBM altered outside the body should not be deemed as originating from a human body (such as cell lines).⁹⁹⁹

The Human Tissue Authority (HTA)¹⁰⁰⁰ recommends that where potentially important medical information is found in the process of research, feedback to donors is encouraged and this should be reflected by researchers to donors during the consent process.¹⁰⁰¹

The Human Research Authority (HRA)¹⁰⁰² directs research ethics committees (RECs) on research involving human tissue and facilitates research that is beneficial to society within the legal framework of the UK. The HRA must indicate whether there are plans to inform donors of any clinically significant findings. In the event of any collaborative research, UK RECs are not obliged to review consent agreements from outside of the UK nor any other agreements made by the collaborating partners.¹⁰⁰³

The Medical Research Council's Ethical Guidelines¹⁰⁰⁴ on research practices regarding collaborative research stipulate the need for a formal written agreement between parties regarding ownership, custodianship, responsibilities, future use of samples and handling of intellectual property. The MRC's Operational and Ethical Guidelines relating to Human Tissue and Biological samples for Use in Research provide that all donations are altruistic (with some exceptions) and encourage that the custodianship of the donated material should rest with the institutions/corporations rather than individual researchers.¹⁰⁰⁵ In its guidelines, the MRC specifically prohibits the commodification of the human body and insists that research participants can only be paid for their time and expenses to prevent undue inducement. The MRC Guidelines acknowledge that intellectual property could

⁻

⁹⁹⁸Human Tissue Act 2004 https://www.legislation.gov.uk/ukpga/2004/30/data.pdf (Date of use: 26 May 2023).

⁹⁹⁹Section 54(7) of the Human Tissue Act 2004.

¹⁰⁰⁰Human Tissue Authority Code of Practice A https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 26 May 2023).

¹⁰⁰¹ Section 34 of the HTA Code of Practice and Standards for Research page 10. https://content.hta.gov.uk/sites/default/files/2020-11/Code%20E.pdf (Date of use 26 may 2023).

¹⁰⁰²NHS Health Research Authority https://www.hra.nhs.uk/about-us/ (Date of use: 26 May 2023).

¹⁰⁰³Section 12.47 of HRA SOPs found at https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Procedures_Version_7.5.1_August_2021_Final_Accessible_07IVkXt.pdf (Date of use: 26 May 2023).

¹⁰⁰⁴Medical Research Council https://www.ukri.org/councils/mrc/ (Date of use: 26 May 2023).

¹⁰⁰⁵ Section 2 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 26 May 2023).

arise from research using HBMs. In addition, the guidelines elucidate that although donors must be informed of the IP development and possible commercialisation where possible, HBM donors must be told that they are not entitled to any profits or IP rights that are generated from their donated samples and being used in academic settings. ¹⁰⁰⁶

No legally binding framework for benefit sharing with donors of HBM in the UK exists although the concept of custodianship is prevalent in the various guidelines. As a HIC altruistic research is promoted in the UK as it likely that most of their citizens will share in the benefits of all research. South Africa cannot afford this approach to health research; being an LMIC, the attainment of therapeutics and the equal distribution of benefits that may arise from human health research to all its citizens is not guaranteed. The explosion in health research that precipitated an increase in North–South collaborations forces LMICs to enforce benefit sharing agreements not only to remedy past exploitative collaborations but also to be balanced with the public interest and respect for local traditions. 1007

6.2.5.2 Australia

Australia—like the UK—is a high-income country whose citizens have access to high-quality healthcare. Australia is also at the forefront of health research; the National Statement on Ethical Conduct in Human Research (the National Statement) applies to all research involving human beings. The National Statement is not legally binding but access to research funds from the National Health and Medical Research Council (NHMRC) requires compliance with the Principles of the National Statement. The National Statement is not legally binding but access to research funds from the National Health and Medical Research Council (NHMRC) requires compliance with the Principles of the National Statement.

The question of the ownership of HBMs is not addressed at all in the National Statement but the concept of benefit sharing is deemed to be one of the pillars of ethical research throughout the document. The Guidelines state that benefit sharing arrangements must be in place during a research project and to whom and with whom these benefits accrue must

¹⁰⁰⁶Section 7 of MRC Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 26 May 2023).

¹⁰⁰⁷Slabbert M "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

¹⁰⁰⁸National Statement on Ethical Conduct in Human Research https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 26 May 2023).

National Statement on Ethical Conduct in Human Research https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 26 May 2023).

be clear. The Guidelines also state that in the event that commercialisation occurs, the HBMs donors should be notified and discussions must ensue on how this is to be managed and how the benefits, if any, will be distributed.¹⁰¹⁰

Regarding research involving marginalised communities, specifically the Aboriginal and Torres Island peoples, the National Statement requires that benefit sharing agreements should be agreed upon with these specific communities and should be for the advancement and capacity building of the peoples in the community.¹⁰¹¹

Regarding collaborative research with other countries, the National Statement advocates for the fair distribution of benefits and burdens between the collaborators in a collaboration that is not exploitative.¹⁰¹²

The National Statement declares that benefit sharing is a pillar of ethical research and that all health research must be seen to comply in order to obtain funding from the Australian government. Furthermore, donors are entitled to benefit from any commercialisation that may result from their donated HBM. The National Statement directs that benefit sharing should occur via agreements or as part of a Material/Data Transfer Agreement. The South African MTA for Human Biological Materials also calls for the negotiation of benefits before the commencement of a research project; however, no benefit sharing agreement is required between the donor and the institution that receives the donated HBM.

6.2.5.3 Uganda

Uganda is an LMIC like South Africa as well as one of the few countries at the forefront of research collaborations with HICs in Africa.

The National Guidelines for Research Involving Humans as research participants (National Guidelines)¹⁰¹⁵ and the National Research Biobanking Guidelines¹⁰¹⁶ regulate health research in Uganda. In Uganda, donors retain ownership of their donated HBMs if the latter

¹⁰¹⁰ Section 1 of the National Statement https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1_(Date of use: 26 May 2023).

¹⁰¹¹Chapter 4.7 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 77–79. ¹⁰¹²Chapter 4 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 77–79.

¹⁰¹³Chapter 4 of the National Statement on Ethical Conduct in Human Research of the NHRMC at 77–79.

¹⁰¹⁴Proclamation No 11 Government Gazette 41781 of 20 July 2018.

National Guidelines of Research Involving Human as Research Participants https://iuea.ac.ug/sitepad-data/uploads/2021/03/Human-Subjects-Protection-Guidelines-July-2014.pdf (Date of use: 26 May 2023).

¹⁰¹⁶National Research Biobanking Guidelines https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 26 may 2023).

are linked to the donor and a custodianship exists between the HBM donor and the primary receiving institute in Uganda. 1017 The National Guidelines and Biobank Guidelines state that research participants should benefit from research and must not be exploited. In both guidelines, a material transfer agreement (MTA) must be negotiated between the custodian of the HBM and the receiving institution if donated HBMs are transferred between institutions, either locally or internationally. 1018 The MTA should have clauses pertaining to benefit sharing arrangements, stating what benefits like capacity building and/or infrastructure development are likely to accrue. In the event of commercialisation using the samples held in trust, a second MTA needs to be negotiated, stating what benefits for Uganda or the primary institution in Uganda would be realised. 1019

Regarding intellectual property that could potentially emanate from the use of HBMs stored in trust, the biobank and primary source institution should define an intellectual property policy that should be enforced by an MTA and/or a data transfer agreement (DTA).¹⁰²⁰

Uganda has entrenched benefit sharing agreements in their law. Being an LMIC, Uganda has a similar history to South Africa's, which gave rise to an unequal society in which good healthcare was not available to all. There is a need for a legal framework to protect donors and local institutions and/or communities concerning advancements in scientific research which could benefit all.

6.2.6 Key elements in developing a benefit sharing agreement for health research in South Africa

Many countries, including South Africa, lack legislative structures and governance frameworks to facilitate fair benefit sharing to all stakeholders involved in health research.¹⁰²¹ Every new global health crisis highlights the need for the fair benefit sharing of HBMs samples between LMICs and HICs, to avoid exploitation and protect the vulnerable.¹⁰²² In the absence of legally binding law to guide benefit sharing frameworks, ethical guidelines can be used, although the reviewed ethical guidelines suggest that some

¹⁰¹⁷Section 10.3 of the Uganda "National Guidelines" at 28.

¹⁰¹⁸Section 10.4 of the Uganda "National Guidelines" at 28.

¹⁰¹⁹Section 10.4(j)(k)(l) of the Uganda "National Guidelines" at 30.

¹⁰²⁰Section 8.2 of the Uganda National Research Biobanking Guidelines at 20.

¹⁰²¹Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117.

¹⁰²²Mckenna M "Colonialists are coming for blood-Literally" 2019 *Ideas* found at https://www.wired.com/story/ebola-epidemic-blood-samples/ (Date of use: 26 May 2023).

guidelines are absent, outdated, conservative or difficult to navigate.¹⁰²³ To arrive at a suitable template for benefit sharing in health research in SA, this study refers to pre-legal frameworks previously discussed in this thesis.¹⁰²⁴

6.2.6.1 Legal frameworks for benefit sharing

The Nagoya Protocol¹⁰²⁵ refers to access to and the benefit sharing of non-human genetic resources but excludes HBMs and digital sequence information.¹⁰²⁶ The Nagoya Protocol proposes a comprehensive list of benefits, both monetary and non-monetary that could be included in benefit sharing arrangements.¹⁰²⁷ Some scholars have advocated that the scope of access and benefit sharing of the CBD via supplementary Nagoya Protocol should be expanded to include HBMs.¹⁰²⁸ The HUGO Ethics Committee issued a statement in 2002 on benefit sharing regarding HBMs¹⁰²⁹The Council for International Organisations and Medical Sciences (CIOMS) updated their Guidelines for Health-related Research involving Humans in 2015, advocating for negotiated benefit sharing between LMICs and HICs in health research.¹⁰³⁰

These international legal frameworks and guidelines to which South Africa subscribe are upheld as principles of international customary law but do not take into account the obvious difference in negotiating powers between stakeholders in HICs and those in LMICs. ¹⁰³¹

¹⁰²³De Vries J, Munung SN, Matimba A et al. "Regulation of genomic and biobanking research in Africa: a content analysis of ethics guidelines, policies and procedures from 22 African countries" 2017 BMC Medical Ethics 18(1):1–9.

¹⁰²⁴ Discussed in Chapter 2 of this thesis.

Nagoya Protocol on Access to Genetic Resources and Fair and Equitable Sharing of Benefits Arising from their Utilisation. Secretariat of the Convention of Biological Diversity, 2011. http://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf (date of use: 26 may 2023).

¹⁰²⁶Bedeker A, Nichols M, Allie T *et al.* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰²⁷Chapter 2.1.1 of this thesis covers the list.

¹⁰²⁸Chaturvedi S, Crager S, Ladikas M "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) *Benefit Sharing From Biodiversity to Human Genetics* (Springer Science 2013) 153–178.

¹⁰²⁹HUGO Statement on Benefit Sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committee%20rec ommends,who%20participated%20in%20such%20research (Date of use: 23 May 2023).

¹⁰³⁰CIOMS and WHO International Ethical Guidelines for health research Involving Humans https://cioms.ch/wp-content/uploads/2017/01/WEB-CIOMS-EthicalGuidelines.pdf (Date of use: 26 May 2023).

¹⁰³¹Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

Specific to South Africa, the San Code of Research Ethics¹⁰³² and the National MTA of Human Biological Materials¹⁰³³ call for benefit sharing. The MTA leaves the benefit sharing agreement to be negotiated between the provider and recipient of the HBM. The National MTA¹⁰³⁴ states what possible benefits could accrue, but the list needs to be expanded and to be more specific on to whom the benefits should accrue to accommodate the various stakeholders in a research project.

These frameworks discussed above indeed promote benefit sharing yet fail to address the issue of publicly and privately funded research design in a world where the private sector has become the biggest investor in health research due to the commercialisation of research products. Therefore, it is necessary to have specific benefit sharing agreements at hand to enable vulnerable populations to negotiate fair contracts. 1036

6.2.6.2 Challenges to benefit sharing

Despite global efforts to promote benefit sharing,¹⁰³⁷ benefit sharing plans and implementation rarely feature in research programmes, grant applications or the requirements of research ethics committees.¹⁰³⁸ The need for benefit sharing in health research is a concept that is agreed on by many research stakeholders, yet controversy persists.¹⁰³⁹

The lack of trust and integrity between providers and resource users and disrespect of other cultures are seemingly obvious problems in the implementation of fair benefit sharing arrangements. The lack of infrastructure, equipment and electricity in LMICs means that some HBMs which need to be stored with a continuous cold chain are then shipped to HICs because it is costly to store them in the LMIC. This has led to some researchers

¹⁰³²This pertains specifically to the San communities that require any monetary or non-monetary benefits to be discussed with the research participants and/or the community.

¹⁰³³Proclamation No 719 Government Gazette 41781 of 20 July 2018.

¹⁰³⁴Proclamation No 719 Government Gazette 41781 of 20 July 2018.

¹⁰³⁵Simm K "Benefit sharing frameworks—justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 26 May 2023).

¹⁰³⁶Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

¹⁰³⁷Chapter 2 of this thesis.

¹⁰³⁸Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰³⁹Sedyaningsih ER, Isfandari S, Soendoro T *et al* "Towards mutual trust, transparency and equity in virus sharing mechanism: the avian influenza case of Indonesia" 2008 *Ann Acad Med Singapore* 37:482–

¹⁰⁴⁰Moodley K and Singh S "It's all about trust: reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

in HICs withholding samples, 1041 which contributes to situations where fair benefit sharing is hindered. 1042 The large number of stakeholders involved and the power dynamics between them, as well as the question of who benefits, when and how, present challenges to the implementation of benefit sharing agreements. 1043 In South Africa, the enactment of the POPIA¹⁰⁴⁴ gives research participants legal rights to their data, which could cause difficulties for health research and benefit sharing agreements. This is because transferring personal data outside of South African borders must be for the benefit of the research participant, whereby in the absence of consent from the participant, given that if they could give consent, they would. 1045 Moreover, consulting a large number of participants when large numbers of data need to be transferred out of SA is impractical and almost impossible to achieve. Other challenges to fair benefit agreements and their implementation include, but are not limited to, post study access. 1046 It has been suggested that some research ethics committees in LMICs are inadequately trained and often lack the resources to be effective and subsequently, are pressured by researchers and sponsors, thereby contributing to the lack of the implementation of fair benefit sharing agreements. 1047 Multiple stakeholders are involved in research projects and must all benefit at some stage of the project; hence, there is a need for effective communication between the scientific community and the relevant national and international policymakers. 1048

6.2.6.3 Proposed benefit sharing models

The failure to implement benefit sharing arrangements is due to non-binding legislation that provides inconsistent and incomplete frameworks for benefit sharing.¹⁰⁴⁹ Accordingly, the result of this scenario is often that national governments develop their own access and

¹⁰⁴¹Mckenna M "Colonialists are coming for blood-Literally" 2019 *Ideas* found at https://www.wired.com/story/ebola-epidemic-blood-samples/ (Date of use: 26 May 2023).

¹⁰⁴²Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

¹⁰⁴³Berg K "The ethics of benefit sharing" 2001 *Clin Genet* 59(4):240–243.

¹⁰⁴⁴South Africa Protection of Personal Information Act No 4. Of 2013.

¹⁰⁴⁵Section 72(1) of the POPIA.

¹⁰⁴⁶Schroeder D and Gefenas E "Realizing benefit sharing-the case of post-study obligations" 2012 *Bioethics* 26(6):305–314.

¹⁰⁴⁷ Nyika A, Kilama W, Chilengi R et al "Composition, training needs and independence of ethics review committees across Africa: are the gate-keepers rising to the emerging challenges?" 2009 J Med Ethics 35:189–193.

¹⁰⁴⁸Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰⁴⁹Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117.

benefit sharing frameworks that might not support global research initiatives.¹⁰⁵⁰ When designing a framework for health research, many issues must be considered, such as the funding of the research (public vs private vs semi-private), property rights, public engagement, compensation for research participants, consent issues, data sharing, the secondary use of samples and data returning results, royalties, financial gains and other types of gain and benefits.¹⁰⁵¹

Several models of benefit sharing frameworks dominate ethical debates.¹⁰⁵² The private benefit sharing model¹⁰⁵³ applies to privately owned biobanks or tissue banks where private biotechnology companies act as brokers of the HBMs and health data for multiple researchers.¹⁰⁵⁴ This system presents challenges in countries like SA where there is little regulation of biobanks or tissue banks.¹⁰⁵⁵ To date, there have been incidences when such private research sites shared data and samples with research sites in HICs without the necessary oversight and consent.¹⁰⁵⁶

The altruistic model of benefit sharing¹⁰⁵⁷ is based on the notion that all health research ought to be altruistic in nature to generate generalised knowledge that could be used for the benefit of all.¹⁰⁵⁸ This model would not work in SA, an LMIC with an unequal healthcare system that disadvantages most of the population. The healthcare systems of HICs benefit all; therefore, to propose such a system in SA would be exploitative.¹⁰⁵⁹

¹

¹⁰⁵⁰Chaturvedi S, Crager S, Ladikas M et al "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 153–178.

¹⁰⁵¹Cambon-Thomsen A, Rial-Sebbag E and Knoppers BM "Trends in ethical and legal frameworks for the use of human biobanks" 2007 *Eur Respir J* 30:373–382.

¹⁰⁵²Chapter 5, Section .3.1 of this thesis.

¹⁰⁵³Chapter 2, Section 2.5.1 of this thesis.

Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 New Engl J Med 349(12): 1180–1184.

¹⁰⁵⁵Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impendiments to accessing health care" 2019 *SAJBL* 12(1):27–31.

¹⁰⁵⁶Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 *Developing World Bioethics* 21:125–130.

¹⁰⁵⁷Chapter 2 Section 2.5.2 of this thesis.

¹⁰⁵⁸Simm K "Benefit sharing frameworks–justifications for and against benefit sharing in human genetic research" 2007 A report for GenBenefit, available at: www.uclan.ac.uk/genbenefit (Date of use: 26 May 2023).

¹⁰⁵⁹Schroeder D, Gefanas E, Chennells R et al "Realizing benefit sharing: is there a role for ethics review?" in Schroeder D and Cook Lucas J (eds) Benefit sharing from biodiversity to human genetics (Springer Science 2013) 179–202.

The reasonable availability benefit sharing model was proposed to prevent the exploitation of research participants in HIV trials in LMICs in the 1990s. 1060 This model's weakness resides in the fact that a viable benefit might never be developed and if it were, who would decide on a suitable benefit for the research participants in the LMICs. 1061

The fair benefits model for benefit sharing has been suggested to address the failures of the reasonable availability model. It stipulates that the host community/participants of the research can negotiate for benefits beyond those resulting from the research project. The model has been criticised for promoting unjust research, whereby rich sponsors from HICs could choose to conduct research in poor communities with low negotiating powers in LMICs. 1063

The reasonable availability and fair benefits models concur that the primary purpose of research is to create generalised knowledge and benefits should accrue to those that participate in research. However, both models fail to address what entails a fair benefit, who decides that and who is responsible for protecting research participants from exploitation in LMICs. 1065

The charitable trust model¹⁰⁶⁶ proposes a fiduciary relationship in which trustees hold title to the property but are obligated to keep or use the property for the benefit of the beneficiary.¹⁰⁶⁷ This model has been successful in cases were rare disease researchers have constructed tissue banks to enable researchers to control research design, the implementation of research results and benefit sharing.¹⁰⁶⁸ The Human Tissue Authority

¹⁰⁶⁰ Participants in the 2001 Conference on Ethical Aspects of Research in Developing Countries Moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits' 2004 Hastings Center Report 34(3):17–27.

Participants in the 2001 Conference on Ethical Aspects of Research in Developing Countries Moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits' 2004 Hastings Center Report 34(3):17–27.

¹⁰⁶²Participants in the 2001 Conference on Ethical Aspects of Research in Developing Countries Moral standards for research in developing countries: from 'reasonable availability' to 'fair benefits' 2004 *Hastings Center Report* 34(3):17–27.

¹⁰⁶³London AJ and Zollmann K "Research at the auction block: problems for the fair benefits approach to international research" 2010 *Hastings Center Report* 40(4):34–45.

¹⁰⁶⁴Macpherson CC "Research ethics guidelines and moral obligations to developing countries: capacity-building and benefits" 2019 *Bioethics* 33:389–395.

¹⁰⁶⁵Macpherson CC "Research ethics guidelines and moral obligations to developing countries: capacity-building and benefits" 2019 *Bioethics* 33:389–395.

¹⁰⁶⁶Chapter 2 Section 2.5.3 of this thesis.

¹⁰⁶⁷Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

¹⁰⁶⁸Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

(HTA) of the UK also uses this model since it acts as the trustee of all human tissue and HBMs collected in the UK.¹⁰⁶⁹

In recent years, more models of benefit sharing have been introduced by various stakeholders active in health research in LMICs.

The H3Africa¹⁰⁷⁰ research consortium has proposed a benefit sharing model in which the primary goal of research is founded on capacity building in LMICs, whereby capacity building and health improvement are the bases of research projects.¹⁰⁷¹ Its weakness resides in the uncertainty of sustaining research and building capacity beyond the NIH-Wellcome Trust funding period.¹⁰⁷²

Bedeker *et al*¹⁰⁷³ propose an ethical benefit sharing model that would allow for benefits to accrue to multiple stakeholders. To promote ethical benefit sharing, the authors propose a two-dimensional framework that would enable research stakeholders to identify opportunities in research programmes and thereby, cater to all of the multiple stakeholders involved in a research project.¹⁰⁷⁴

Some scholars¹⁰⁷⁵ have suggested that the CBD should be the main treaty to influence a benefit sharing model for human biological materials such as genetic resources. However, to date, the issue of access and benefit sharing remains a major stumbling block in negotiations on trade-related aspects of intellectual property rights (TRIPs) at the World Trade Organisation (WTO).¹⁰⁷⁶

¹⁰⁶⁹UK Human Tissue Act https://www.hta.gov.uk/policies/human-tissue-act-2004 (Date of use: 26 May 2023).

¹⁰⁷⁰Human Heredity & Health in Africa https://h3africa.org/ (Date of use: 26 May 2023).

¹⁰⁷¹Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–170.

¹⁰⁷² Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18:165–170

¹⁰⁷³Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰⁷⁴Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰⁷⁵Chaturvedi S, Crager S, Ladikas M *et al* "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 153–178.

¹⁰⁷⁶Chaturvedi S, Crager S, Ladikas M et al "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD?" in Schroeder D and Cook Lucas J (eds) Benefit sharing: from biodiversity to human genetics (Springer Science 2013) 153–178.

6.2.6.4 Proposed model for a benefit sharing agreement for human health research in South Africa

Globally, the incorporation of legally binding benefit sharing frameworks and their translation into practical implementation has been slow.¹⁰⁷⁷ In South Africa specifically, the challenge is to find a benefit sharing model that balances relevant interests, redresses economic imbalance and provides research participants fairer and more active roles in influencing the sharing of benefits.¹⁰⁷⁸ The ethical principles of the model should also accommodate fast-changing scientific principles.¹⁰⁷⁹ Furthermore, the model must have room to accommodate multiple stakeholders in a research project and to communicate effectively in a language that is common to all involved in the research.¹⁰⁸⁰

In Chapter 5 of this thesis,¹⁰⁸¹ the researcher suggested a benefit sharing agreement for health research in South Africa based on the charitable trust model for benefit sharing.¹⁰⁸²

This study proposes that the donor of HBMs and/or data transfers the custodianship of said donations to a trust (the primary institution that obtains the donated materials), which is set up to protect and safeguard said materials. The transfer should be effected via the process of informed consent, with the encouragement of the need for continuous communication between the two parties to enable ease of negotiations on the future use of the donated materials. The trust is then authorised to sign the existing MTA¹⁰⁸³ with an end user (Recipient). The MTA should be seen to benefit the beneficiary of the trust via the addition of a benefit sharing agreement as an annexure to the MTA.

6.4 Conclusion

As expressed earlier in this thesis, while the law must be seen to follow tradition and legal precedent when formulating public policy or legislation, it should also be responsive to changes requiring legal regulation. It is evident that after analysing the ethico-legal

¹⁰⁷⁷Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰⁷⁸ Slabbert, MN "The legal regulation of access and benefit sharing with regard to human genetic resources in South Africa" 2011 J. Contmpt Roman Dutch Law 74: 605.

¹⁰⁷⁹These include the sharing and secondary use of data and human biological materials locally and globally and changes in consent.

¹⁰⁸⁰Bedeker A, Nichols M, Allie T *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2):e008096.

¹⁰⁸¹Chapter 5 of this thesis.

¹⁰⁸²Attached to this Chapter as Annexure B.

¹⁰⁸³South Africa Material Transfer Agreement for Human Biological Materials Government Notice 719. Government Gazette 41781 of 20 July 2018.

framework for benefit sharing in South Africa, the legislation is found lacking. If LMICs are to achieve justice in exchange, it is necessary to right the wrongs committed by HICs in LMICs through exploitative research and also enforce that benefit sharing plans and implementations should feature in research proposals and grant applications and be a requirement in health research involving HBMs in certain instances.

The Recent global COVID-19 health crisis and the alleged misuse of African DNA in the UK¹⁰⁸⁴ have highlighted the need for benefit sharing between HICs and LMICs.Benefit sharing can no longer be looked upon as a notion that will encourage undue inducement of research participants when said participants have no access to life-saving therapeutics, capacity development or infrastructure development, in which case this is akin to exploitation and a failure of justice. The case of the Welcome Sanger Institute should serve as a cautionary tale to researchers and institutions in LMICs that enter into collaborations with researchers in HICs. During this research, it was alleged by some universities in South Africa that the Welcome Sanger Institute commercialised a gene chip using DNA donated by South African and other African donors without following the proper consent processes.

The lack of cultural sensitivity, such as in the example above leads to a lack of trust in communities that value and attach great importance to blood and tissue collected for use as research samples. COVID-19 vaccine hesitancy can be partially blamed on the inability of researchers to cement initial trust in communities.¹⁰⁸⁵

It is also important to recognise that research projects involve multiple stakeholders with different expectations of what benefits ought to accrue to them at various stages of the research project. A trust would solve such issues since it would communicate to various stakeholders simultaneously in negotiating their expectations, using a language that is understood across the different negotiating forums, whether the scientific community or relevant national and international policymakers.

Adopting the charitable trust model, proposed in chapter 5 of this thesis, would make the first recipients of HBMs the trustees of the HBMs instead of brokers having legal fiduciary

188

Moodley K and Kleismidt A "Allegations of misuse of African DNA in the UK: Will data protection legislation in South Africa be sufficient to prevent recurrence" 2021 Developing World Bioethics 21:125-30.
 Bedeker A, Nichols M, Allie T et al "A framework for the promotion of ethical benefit sharing in health research" 2022 BMJ Global Health 7(2):e008096.

duties over the HBMs, whilst permitting the use of the donated HBMs in a way that benefits the donor as a beneficiary of the trust.

BIBLIOGRAPHY

7.1 Books

Anarson and Schroeder Philosophical principles.

Anarson G and Schroeder D "Exploring central philosophical concepts in benefit sharing: vulnerability, exploitation and undue inducement" in Schroeder D and Cook Lucas JC (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 9–32.

Andanda et al Legal frameworks for benefit sharing 34.

Andanda P *et al* "Legal frameworks for benefit sharing: from biodiversity to human genomics" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 33–64.

Basler Common heritage.

Basler K *The concept of the common heritage of mankind in International law* (Nijhoff Publishers 1998).

Chaturvedi *et al Inclusive approach to benefit sharing*Chaturvedi S *et al* "Promoting an inclusive approach to benefit sharing: expanding the scope of the CBD" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 153–178.

Chennelis and Steenkamp International genomics.

Chennelis R and Steenkamp A "International genomics research involving the San people" in Schroeder D, Cook Lucas J *et al* (eds) *Ethics dumping, case studies from north-south research collaborations* (Springer International Publishing 2018) 15–22.

Dictionaries, Oxford. Concise Oxford English Dictionary: CD-ROM edition, Windows/Mac Individual User Version 1.0. (Oxford University Press 2011).

Halewood and Nnadozie Priority to commons.

Halewood M and Nnadozie K "Giving priority to the commons: the international treaty on plant genetic resources for food and agriculture" in Tansey G and Rajotte T (eds) *The future control of food* (London, Earthscan 2008).

Hollis, Pogge and Schroeder Beyond benefit sharing.

Hollis A, Pogge T and Schroeder D "Beyond benefit sharing: towards realising the human rights to health" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 203–216.

Hyam British Empire.

Hyam R *Understanding the British empire* (Cambridge University Press 2010).

Lucas et al Who benefits?

Lucas CJ *et al* "Donating human samples: who benefits? Cases from Iceland, Kenya and Indonesia" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 95–127.

Macklin Double standards.

Macklin R *Double standards in medical research in developing countries* Volume 2 (Cambridge University Press 2004).

Schroeder and Cook Lucas Benefit sharing.

Schroeder D and Cook Lucas J "Benefit sharing: from biodiversity to human genetics—an introduction" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 1–8.

Schroeder and Cook Lucas Best Practice.

Schroeder D and Cook Lucas J "Towards best practice for benefit sharing involving access to human biological resources: conclusions and recommendations" in Schroeder

D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 217–230.

Suzman Regional assessment 4.

Suzman J An introduction to the regional assessment of the status of the San in Southern Africa (Windhoek: Legal Assistance Centre 2001) 4.

Schroeder et al Realising benefit sharing

Schroeder D *et al* "Realising benefit sharing: is there a role for ethics review?" in Schroeder D and Cook Lucas J (eds) *Benefit sharing: from biodiversity to human genetics* (Springer Science 2013) 179–202.

Van der Walt and Pienaar Property.

Van der Walt AJ and Pienaar GJ *Introduction to the Law of Property* 7th ed (Juta 2017) 43–56.

Zhao and Zhang Collaborative genetic research.

Zhao Y and Zhang W "An international collaborative genetic research project conducted in China" in Schroeder D, Cook Lucas J *et al* (eds) *Ethics dumping, case studies from North-South research collaborations* (Springer International Publishing: Cham 2018) 71–80.

7.2 Case Law

Yearworth v North Bristol Trust 2009 (3) WLR 118

Roche v Douglas 2000 WASC (146) 2002 (22) WAR 331

Greenberg v. Miami Children's Hospital Research Institute 264 F. Supp. 2d D. Fla. (2003)

7.3 Dissertations and Theses

Chennels Equitable access.

Chennells RS Equitable access to human biological resources in developing countries: benefit sharing without undue inducement (PhD thesis School of Health: University of Central Lancashire 2014).

Mahomed Ethico-legal framework.

Mahomed S *An ethico-legal framework for the regulation of biobanks in South Africa* (PhD thesis Steve Biko Centre of Bioethics: University of the Witwatersrand PhD thesis 2018) Available at: https://wiredspace.wits.ac.za/server/api/core/bitstreams/7b7894ae-6f5a-4fd6-81cf-65e36ff9563e/content (Date of use: 26 May 2023).

Moodley Legitimacy

Moodley K Legitimacy, trust and governance in biobanking in South Africa (MBA thesis University of Cape Town 2015)

7.4 Journal Articles

Anarson and Van Niekerk 2009 Cambridge Quarterly of Healthcare Ethics.

Anarson G and Van Niekerk A "Undue fear of inducements in research in developing countries" 2009 *Cambridge Quarterly of Healthcare Ethics* 18(2):122–129.

Angell, Terrant and Dixon-Woods 2009 J Clin Pathol.

Angell A, Terrant C and Dixon-Woods M "Research involving storage and use of human tissue: how did the Human Tissue Act 2004 affect decisions by Research Ethics Committees?" 2009 *J Clin Pathol* 62(9):825–9.

Aramoana and Koea 2020 JCO Global Oncology.

Aramoana, J and Koea, J "An integrative review of the barriers to indigenous peoples' participation in biobanking and genomic research" 2020 *JCO Global Oncology* 6:83–91.

Ballantyne 2010 Am J Bioeth.

Ballantyne AJ "How to do research fairly in an unjust world" 2010 Am J Bioeth 10:26–35.

Bedeker et al 2022 BMJ Global Health.

Bedeker A *et al* "A framework for the promotion of ethical benefit sharing in health research" 2022 *BMJ Global Health* 7(2): e008096.

Berg 2001 Clin Genet.

Berg K "The ethics of benefit sharing" 2001 Clin Genet 59:240-243.

Beskow 2016 Annu. Rev. Genom. Hum. Genet.

Beskow LM "Lessons from HeLa cells: the ethics and policy of biospecimens" 2016 *Annu. Rev. Genom. Hum.Genet* 17:395–417.

Beyleveld and Bronsword 1998 Modern Law Review.

Beyleveld D and Bronsword R "Human signity, human rights, and human genetics" 1998 *Modern Law Review* 61(5): 661–680.

Bjorkman and Hansson 2006 J Med Ethics.

Bjorkman B and Hansson SO "Bodily rights and property rights" 2006 J Med Ethics 32:209–214.

Cambon-Thomsen, Rial-Sebbag and Knoppers 2007 Eur Respir J.

Cambon-Thomsen A, Rial-Sebbag E and Knoppers BM "Trends in ethical and legal frameworks for the use of human biobanks" 2007 *Eur Respir J* 30:373–382.

Chalmers 2015 Journal of Law Medicine and Ethics.

Chalmers D "Biobanking and privacy laws in Australia" 2015 *Journal of Law Medicine* and Ethics 43(4):703–713.

Chen and Pang 2015 Bull World Health Organ.

Chen H and Pang T "A call for global governance of biobanks" 2015 *Bull World Health Organ* 93:113–117.

Christopher 1990 Urban Studies.

Christopher AJ "Apartheid and urban segregation levels in South Africa" 1990 *Urban Studies* 3:421–440.

Dauda and Joffe 2018 Developing World Bioethics.

Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioeth* 18:165–170.

De Jonge and Korthals 2006 Developing World Bioethics.

De Jonge B and Korthals M "Vicissitudes of benefit sharing of crop genetic resources: downstream and upstream" 2006 *Dev World Bioeth* 6: 144–157.

Dauda and Dierickx 2013 BMC Medical Ethics.

Dauda B and Dierickx K "Benefit sharing: an exploration on the contextual discourse of a changing concept" 2013 *BMC Medical Ethics* 14(1):1–8.

Dauda and Joffe 2018 Developing World Bioethics.

Dauda B and Joffe S "The benefit sharing vision of H3Africa" 2018 *Developing World Bioethics* 18(2):165–170.

De Vries et al 2017 BMC Medical Ethics.

De Vries J, Munung SN, Matimba A *et al* "Regulation of genomic and biobanking research in Africa: a content analysis of ethics guidelines, policies and procedures from 22 African countries" 2017 *BMC Medical Ethics* 18(1):1–9.

Dhai 2012 SAJBL.

Dhai A "Vulnerability exploited and a population betrayed" 2012 SAJBL 5(2):62–63.

Dhai, Mahomed and Sanne 2015 S Afr J BL.

Dhai A, Mahomed S and Sanne I "Biobank and human health research: balancing progress and protections" 2015 *S Afr J BL* 9:55–59.

Dudgeon et al 2010 Working together.

Dudgeon P, Wright M, Paradies Y *et al* "The social, cultural and historical context of Aboriginal and Torres Strait Islander Australians" 2010 *Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practice*" 25–42

Emmanuel et al 2004 J Infect Dis.

Emmanuel EJ, Wendler D, Killen J *et al* "What makes clinical research in developing countries ethical? The benchmarks of ethical research" 2004 *J Infect Dis* 189:930–937.

Emmanuel, Currie and Herman 2005 The Lancet.

Emmanuel EJ, Currie XE and Herman A "Undue inducement in clinical research in developing countries: is it a worry?" 2005 *The Lancet* 366:336–340.

Evans, Hills and Levine 2020 AMA Journal of Ethics.

Evans NG, Hills K and Levine AC "How should the WHO guide access and benefit sharing during infectious disease outbreaks" 2020 *AMA Journal of Ethics* 22(1):28–35.

Fullerton et al 2010 Science Translational Medicine.

Fullerton SM, Nicholas RA, Guzauskas G *et al* "Meeting the governance challenges of next-generation biorepository research" 2010 *Science Translational Medicine* 2(15): 15cm3–15cm3.

Gerstetter et al 2007 The Journal of World Intellectual Property.

Gerstetter C, Gorlach B, Neumann K *et al* "The international treaty on plant genetic resources for food and agriculture within the current legal regime complex on plant genetic resources" 2007 *The Journal of World Intellectual Property* 10(3/4): 259–283.

Gibson 2008 Jurimetrics.

Gibson SF "The Washington University v Catalona: determining ownership of genetic samples" 2008 *Jurimetrics J* 481:167–191.

Guardasani et al 2015 Nature.

Guardasani D, Carstensen T, Tekola-Ayele FR, *et al* "The African Genome Variation Project shapes medical genetics in Africa" 2015 *Nature* 517:327–32.

HUGO 2000 Clinical Genetics.

HUGO Ethics Committee: "HUGO urges genetic benefit sharing" 2000 *Clinical Genetics* 58:364–366.

Hurst 2008 Bioethics.

Hurst S "Vulnerability in research and healthcare; describing the elephant in the room?" 2008 *Bioethics* 22:191–202.

Jeffers 2001 Advances in Nursing Science.

Jeffers BR "Human biological materials in research: ethical issues and the role of stewardship in minimizing research risks" 2001 *Adv. Nurs. Sci* 24(2):32–46

Knoppers et al 2002 Community Genet.

Knoppers BM, Chadwick R, Takebe H *et al* "HUGO urges genetic benefit sharing" 2002 *Community Genet* 3(2): 88–92.

Labuschaigne et al 2019 S Afr Med J.

Labuschaigne M, Dhai A, Mahomed S *et al* "Protecting participants in health research: the South African Material Transfer Agreement" 2019 *S Afr Med J* 109(5):353–356.

Labuschaigne and Mahomed 2019 SAJBL.

Labuschaigne M and Mahomed S "Regulatory challenges relating to tissue banks in South Africa: impediments to accessing health care" 2019 S*AJBL* 12(1):27–31.

Lairumbi *et al* 2012 *Philosophy, Ethics, and Humanities' in Medicine*.

Lairumbi G, Parker M, Fitzpatrick R *et al* "Forms of benefit sharing in global health research undertaken in resource poor settings: a qualitative study of stakeholders' views in Kenya" 2012 *Philosophy, Ethics, and Humanities in Medicine* 7(1):1–8.

Lavoie 1989 Virginia Law Review.

Lavoie J "Ownership of human tissue: life after Moore v. Regents of the University of California" 1989 *Virginia Law Review* 75(7):1363–1396.

Ledford 2010 Nature.

Ledford H "Africa yields two full human genomes" 2010 Nature 463:857.

Lee 2019 Bioethics.

Lee E"Our flawed approach to undue inducement in medical research" 2019 *Bioethics* 33:13-18.

Levine et al 2004 American Journal of Bioethics

Levine C, Faden R, Grady C *et al* "The limitations of 'vulnerability' as protection for human research participants" 2004 *AJOB* 4(3):44–49.

Lie 2010 Hastings Center Report.

Lie R "Fair benefit approach revisited" 2010 Hastings Cent rep. 40(4):3.

London and Zollman 2010 Hastings Center Report.

London AJ and Zollman KJS "Research at the auction block: problems for the fair benefits approach to international research" 2010 *Hastings Center Report* 40(4):34–45.

Macklin 2003 Bioethics.

Macklin R "Bioethics, vulnerability and protection" 2003 *Bioethics* 17(5-6):472–486.

Macpherson 2019 Bioethics.

Macpherson CC "Research ethics guidelines and moral obligations to developing countries: capacity-building and benefits" 2019 *Bioethics* 33:389–395.

Mahesh 2015 S Afr J BL.

Mahesh KP "Laws and regulations associated with ownership of human biological material in South Africa" 2015 *S Afr J BL* 15:8(1)11–18.

Mahomed 2021 Cambridge Quarterly of Healthcare Ethics.

Mahomed S "Human Biobanking in Developed and Developing Countries: An Ethico – Legal Comparative Analysis of the Frameworks in the United Kingdom, Australia, Uganda, and South Africa" 2021 *Cambridge Quarterly of Healthcare Ethics* 30:146–160.

Mahomed and Labuschaigne 2019 S Afr J Bioethics Law.

Mahomed S and Labuschaigne M "The role of research ethics committees in South Africa where human biological materials are transferred between institutions" 2019 *S Afr J Bioethics Law* 12(2):79–83.

Mahomed, Loots and Stauton 2022 S Afr Bioethics Law.

Mahomed S, Loots G and Stauton C "The role of data transfer agreements (DTA) in ethically managing data sharing for research in South Africa" 2022 *S Afr Bioethics Law* 15(1):26–30.

Mahomed, Nöthling-Slabbert and Pepper 2013 S Afr J BL.

Mahomed S, Nöthling-Slabbert M and Pepper MS "The legal position on the classification of human tissue in South Africa: can tissue be owned?" 2013 *S Afr J BL* 6(1):16–20.

Mahomed et al 2016 Dev World Bioeth.

Mahomed S, Nahrens K, Slabbert M *et al* "Managing human tissue transfer across national boundaries—an approach from an institution in South Africa" 2016 *Dev World Bioeth* 16:39–35.

Mahomed, Nöthling-Slabbert and Pepper 2017 SAMJ.

Mahomed S, Nöthling-Slabbert M and Pepper MS "Ownership and human tissue-the legal conundrum: a response to Jordaan's critique" 2017 *SAMJ* 107(3): 196–198.

Mohamed and Sanne 2015 S Afr J BL.

Mohamed S and Sanne I "Benefit sharing in health research" 2015 S Afr J BL 8:60-64.

Mahomed and Staunton 2021 BioLaw.

Mahomed S and Staunton C "Ethico-legal analysis of international sample and data sharing for genomic research during COVID-19: a South African perspective" 2021 *BioLaw* 1 Journal available at www.biodiritto.org ISNN2284-4503.

Mayer 2007 Journal of Applied Philosophy.

Mayer R "What's wrong with exploitation?" 2007 *Journal of Applied Philosophy* 24(2):137–150.

Millum 2012 J Med Ethics.

Millum J "Sharing the benefits of research fairly: two approaches" 2012 *J Med Ethics* 38:219–223.

Mngadi et al 2017 J Med Ethics.

Mngadi KT, Singh JA, Mansoor L and Wassenaar DR "Undue inducement:a case study in CAPRISA 008" 2017 *J Med Ethics* 43:824-828.

Moodley et al 2021 SAMJ 111.

Moodley K, Blockman M, Hawkridge AJ *et al.*, "Hard choices: Ethical challenges in phase 1 of COVID-19 vaccine roll-out in South Africa" 2021 *SAMJ* 111(6):554–558.

Moodley and Kleinsmidt 2021 Developing World Bioethics.

Moodley K and Kleinsmidt A "Allegations of misuse of African DNA in the UK: will data protection legislation in South Africa be sufficient to prevent a recurrence?" 2021 Developing World Bioethics 21:125–130.

Moodley and Singh 2016 BMC Medical Ethics.

Moodley K and Singh S "It's all about trust': reflections of researchers on the complexity and controversy surrounding biobanking in South Africa" 2016 *BMC Medical Ethics* 17(1):1–9.

Nienaber 2011 CILSA.

Nienaber A "Consent to and authorisation of the export and use of human biological specimens for future research–perspectives from three African countries" 2011 *CILSA* 63:225–254.

Nöthling-Slabbert 2008 Stellenbosch Law Review.

Nöthling-Slabbert M "Human bodies in law: arbitrary discursive constructions?" 2008 *Stellenbosch Law Review* 19(1): 71–100.

Nyika et al 2009 J Med Ethics.

Nyika A, Kilama W, Chilengi R *et al* "Composition, training needs and independence of ethics review committees across Africa: are the gate-keepers rising to the emerging challenges?" 2009 *J Med Ethics* 35:189–193.

O'Daniel and Haga 2011 Public Health Genomics.

O'Daniel J and Haga SB "Public perspectives on returning genetics and genomics research results" 2011 Public *Health Genomics* 14(6):346–55.

Participants 2004 Hastings Center Report.

Participants in the 2001 "Conference on Ethical Aspects of Research in Developing Countries Moral standards for Research in Developing Countries: from 'reasonable availability' to 'fair benefits'" 2004 *Hastings Center Report* 34(3):17–27.

Petrini 2012 Journal of Blood Medicine.

Petrini C "Ethical and legal considerations regarding the ownership and commercial use of human biological materials and their derivatives" 2012 *Journal of Blood Medicine* 3:87–96.

Petrini and Ricciardi 2018 Annali dell'Istituto Superiore di Sanitā.

Petrini C, and Ricciardi W "The Convention on Human Rights and Biomedicine twenty years later: a look at the past and a step towards the futur" 2018 *Annali dell'Istituto Superiore di Sanitā* 1(54)(3):17.

Pouris and Pouris 2011 South African Journal of Science.

Pouris A and Pouris A "Patents and economic development in South Africa: managing intellectual property" 2011 *South African Journal of Science* 107(11/12):24–33.

Rachel et al 2021 S Afr J Sci.

Rachel et al "POPIA code of conduct for research" 2021 S Afr J Sci 117(5-6):1–12.

Resnick 1998 Journal of Medical Ethics.

Resnick DB "The commodification of human reproductive material" 1998 *Journal of Medical Ethics* 24:388–393.

Resnick 2005 Frontis.

Resnick DB "The human genome: common resource but not common heritage" 2005 *Frontis* 197–210.

H3Africa Consortium et al 2014 Science.

H3Africa Consortium, Rotimi C, Abayomi A *et al* "Research capacity. Enabling the genomic revolution in Africa" 2014 *Science* 344 (6190):1346–1348.

Sathar and Dhai 2012 SAJBL.

Sathar MA and Dhai A "Laws, regulations and guidelines of developed countries, developing countries in Africa, and BRICS regions pertaining to the use of human biological material (HBM) in research" 2012 *SAJBL* 5(1):51–54.

Sathar, Dhai and Van der Linde 2014 Dev World Bioethics.

Sathar A, Dhai A and Van der Linde S "Collaborative international research: ethical and regulatory issues pertaining to human biological materials at a South African institutional research ethics committee" 2014 *Dev World Bioethics* Vol 14:150–157.

Schroeder 2006 J Med Ethics.

Schroeder D "Benefit sharing: it's time for a definition" 2006 *J Med Ethics* 205–209.

Schroeder and Lassen-Diaz 2006 Dev World Bioeth.

Schroeder D and Lassen-Diaz C "Sharing the benefits of genetic resources: from biodiversity to human genetics" 2006 *Dev World Bioeth* 5(3):135–143.

Schroeder and Gefanas 2009 Cambridge Quarterly of Healthcare Ethics.

Schroeder D and Gefanas E "Vulnerability: too vague and too broad?" 2009

Cambridge Quarterly of Healthcare Ethics 18:113–121.

Schroeder and Gefanas 2012 Bioethics.

Schroeder D and Gefenas E "Realizing benefit sharing-the case of post-study obligations" 2012 *Bioethics* 26(6):305–314.

Schuklenk and Kleinsmidt 2006 *Developing World Bioethics*.

Schuklenk U and Kleinsmidt A "North-South benefit sharing arrangements in bioprospecting and genetic research: a critical ethical and legal analysis" 2006 *Developing World Bioethics* 1471–8731.

Schwartz 1995 Nous.

Schwartz J "What's wrong with exploitation?" 1995 Nous 29(2):158-188.

Sedyaningsih et al 2008 Ann Acad Med Singap.

Sedyaningsih ER, Isfandari S, Soendoro T *et al* "Towards mutual trust, transparency and equity in virus sharing mechanism: the avian influenza case of Indonesia" 2008 *Ann Acad Med Singap* 37(6):482–488.

Simm 2005 Genomics, Society and Policy.

Simm K "Benefit sharing: an inquiry regarding the meaning and limits of the concept in human genetic research" 2005 *Genomics, Society and Policy* 1(2):29–40.

Simm 2007 Nature Reviews Genetics.

Simm K "Benefit sharing: a look at the history of an ethics concern" 2007 *Nat Rev Genet* 8(7):496–496.

Skene 2002 Nature Reviews Genetics.

Skene L "Ownership of human tissue and the law" 2002 *Nature Reviews Genetics* 3:145–147.

Slabbert 2011 Tydskrif vir Hedendaagse Romeins-Hollandse.

Slabbert MN "The legal regulation of access and benefit sharing of human genetic resources in South Africa" 2011 *74 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 605–632.

Slabbert and Pepper 2011 *Tydskrif vir Hedendaagse Romeins-Hollandse*. Slabbert MN and Pepper MS "'A room of our own?' Legal lacunae regarding genomic sovereignty in South Africa" 2011 *73 Tydskrif vir Hedendaagse Romeins-Hollandse Reg* 432–50.

Staunton and Moodley 2016 *South African Medical Journal*.

Staunton C and Moodley K "Data mining and biological sample exportation from South Africa: a new wave of bioexploitation under the guise of clinical care?" 2016 *South African Medical Journal* 106(2):136–138.

Sudoi, De Vries and Kamuya 2021 BMC Medical Ethics.

Sudoi A, De Vries J and Kamuya D "A scoping review of considerations and practices for benefit sharing in biobanking" 2021 *BMC Medical ethics* 22(1):1–16.

Swain and Marusyk 1990 Hastings Centre Report.

Swain MS and Marusyk RW "An alternative to property rights in human tissue" 1990 *Hastings Centre Report* 20(5):12–16.

Thaldar, Botes and Nienaber 2020 BMC Medical Ethics

Thaldar DW, Botes M and Nienaber A "South Africa's new standard material transfer agreement: proposals for improvement and pointers for implementation" 2020 *BMC Medical Ethics* 21(1):1–13

Thaldar and Shozi 2021 South African Law Journal.

Thaldar DW and Shozi B "The legal status of human biological material used for research" 2021 *South African Law Journal*.

Tishkoff et al 2009 Science.

Tishkoff *et al* "The genetic structure and history of Africans and African Americans" 2009 *Science* 324:1035–44.

Troug, Kesselheim and Joffe 2012 Science.

Troug RD, Kesselheim AS and Joffe S "Paying patients for their tissue: the legacy of Henrietta Lacks" 2012 *Science* 337:37–38.

Winickoff 2003 Jurimetrics.

Winickoff D "Governing population genomics: law, bioethics, and biopolitics in three case studies" 2003 *Jurimetrics* 43(2):187–228.

Winickoff and Winickoff 2003 New Engl J Med.

Winickoff DE and Winickoff RN "The charitable trust as a model for genomic biobanks" 2003 *New Engl J Med* 349(12):1180–1184.

Wolitz, Emanuel and Shah 2009 Lancet.

Wolitz R, Emanuel EJ and Shah S "Rethinking the responsiveness requirements for international research" 2009 *Lancet* 2374: 847–49.

Wood 1995 Social Philosophy and Policy.

Wood A "Exploitation" 1995 Social Philosophy and Policy 12(2):136–158.

Yassin et al 2010 Cancer Epidemiology Biomarkers Prev.

Yassin R, Lockhart N, Del Riego MG *et al* "Custodianship as an ethical framework for biospecimen-based research" 2010 *Cancer Epidemiology Biomarkers Prev* 19(4):1012–1015.

7.5 South African legislation (Acts of Parliament and Regulations)

- 1. Biodiversity Act 10 of 2004.
- 2. Constitution of the Republic of South Africa, 1996.
- 3. General Law (Anti-Money Laundering and Combating Terrorism Financing) Amendement Act of 2022.
- Intellectual Property Rights for Publically Financed Research and Development Act 51 of 2008.
- National Health Act 61 of 2003.
- 6. National Material Transfer Agreement of Human Biological Materials of 2018.
- 7. Patents Act 57 of 1978.
- 8. Patents Amendment Act 20 of 2005.
- 9. Trust Property Control Act 57 of 1988.
- 10. Protection of Personal Information Act 4 of 2013.
- 11. Regulations relating to the Artificial fertilisation of persons. GN R 175 in Government Gazette 35099 of 2 March 2012.
- 12. Regulations relating to the use of human biological material. GN R 177 in Government Gazette 35099 of 2 March 2012.
- 13. Regulations relating to the general control of human bodies, tissue, blood, bloodproducts and gametes. GN R 180 in Government Gazette 35099 of 2 March 2012.
- 14. Regulations relating to Tissue banks. GN R 182 in Government Gazette 35099

- 15. of 2 March 2012.
- 16. Regulations relating to Stem Cell banks. GN R 183 in Government Gazette 35099 of 2 March 2012.
- 17. San Code of Research Ethics

7.6 South African policies and guidelines

- 1. HPCSA "Guidelines for Good Practice in Health Care Professions: Guidelines on Over servicing, Perverse Incentives and Related Matters" September 2016 Booklet 11.
- 2. HPCSA "Guidelines for Good Practice in Health Care Professions: General Ethical Guideline for Health Researchers." September 2016 Booklet 13.
- 3. HPCSA "Guidelines for Good Practice in Health Care Professions: General Ethical Guidelines for Biotechnology Research in South Africa." 2008.
- 4. National Department of Health. 2015. "Ethics in Health Research: Principles, Processes and Structures."

7.7 Foreign legislation

United Kingdom

British Theft Act 1968

Care Act 2014

Human Tissue Act 2004 (United Kingdom).

Uganda

Constitution of the Republic of Uganda, 1995.

Public Health Act 1935 (Uganda).

Australia

National Health and Medical Research Council Act 1992

7.8 Foreign guidelines

Australian Government, NHMRC. National Statement on Ethical Conduct in Human Research 2007 (updated in 2018)

https://www.nhmrc.gov.au/research-policy (Date of use: 26 May 2023).

https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 26 May 2023).

Australian Code for the Responsible Conduct of Research.

https://www.nhmrc.gov.au/about-us/publications/australian-code-responsible-conduct-research-2018#block-views-block-file-attachments-content-block-1 (Date of use: 26 May 2023).

Human Tissue Authority (UK) Code of Practice of Guiding Principles and the Fundamental Principle of Consent https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf (Date of use: 26 May 2023).

Human Tissue Authority (UK) Code of Practice and Standards for Research https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf_(Date of use: 26 May 2023).

Human Research Authority (UK) Standard Operating Procedures https://s3.eu-west-

2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Proce dures_Version_7.5.1_August_2021_Final_Accessible_07lVkXt.pdf (Date of use: 26 May 2023).

Ministry of Health. National Kenyan Guidelines for Research and Development of HIV/AIDS Vaccines. Appendix 5: Biological Material Transfer Agreement. Kenya: Ministry of Health 2005. https://www.globalgiving.org/pfil/1108/projdoc.pdf (Date of use: 26 May 2023).

Medical Research Council (UK) Good Research Practice: Principles and Guidelines https://www.ukri.org/publications/principles-and-guidelines-for-good-research-practice/ (Date of use: 26 May 2023).

Medical Research Council (UK) Operational and Ethical Guidelines relating to Human Tissue and Biological Samples for Use in Research

https://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 26 May 2023).

Uganda's National Guidelines for research Involving Humans as Research Participants https://uncst.go.ug/main/wp-content/uploads/download-manager-files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 26 May 2023).

Uganda's National Research Biobanking Guidelines

https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 26 May 2023).

7.9 International and regional instruments and guidelines

Council for International Organisations of Medical Sciences in collaboration with the World Health Organisation, *International Ethical Guidelines for Health-related Research Involving Humans*, Geneva 2016.

Council of Europe Convention for the protection of human rights and dignity of human beings with regards to the application of biology and medicine: Convention of Human Rights and Biomedicine.

https://rm.coe.int/168007cf98 (Date of use: 10 October 2023).

Nagoya Protocol on Access to Genetic Resources and the Fair and Equitable Sharing of Benefits Arising from their Utilisation. Convention on Biological Diversity 2011.

United Nations Convention on Biological Diversity (1992).

United Nations Convention on Law of the Sea (1982).

United Nations Educational, Scientific and Cultural Organization's International

Declaration on Human Genetic Data (2003).

United Nations Educational, Scientific and Cultural Organisation's (UNESCO's) Universal Declaration on the Human Genome and Human Rights (1997).

United Nations Educational, Scientific and Cultural Organisation's (UNESCO's) Universal Declaration on Bioethics and Human Rights (2005).

World Health Organisation's Guideline for Obtaining Informed Consent for the Procurement and Use of Human Tissues, Cells and Fluids in Research (2003).

World Medical Association Declaration of Helsinki – Ethical Principles for Medical Research Involving Human Subjects adopted by the 18th WMA General Assembly, Helsinki, Finland, June 1964 and amended by the 64th WMA General Assembly, Fortaleza, Brazil, October 2013.

World Medical Association Declaration of Taipei on Ethical considerations regarding Health Databases and Biobanks adopted by the 53rd WMA General Assembly, Washington, DC, USA, October 2002 and revised by the 67th WMA General Assembly, Taipei, Taiwan, October 2016.

7.10 Internet sources

Andada P and Cook Lucas J "Majengo HIV/AIDS Research Case. A Report for GenBenefit, 2007"

https://clok.uclan.ac.uk/16039/1/TRUST-664771-Paradigmatic-Case-Studies-WP1-Final.pdf (Date of use: 26 May 2023).

Australian National Statement on Ethical conduct in Human Researh https://www.nhmrc.gov.au/about-us/publications/national-statement-ethical-conduct-human-research-2007-updated-2018#block-views-block-file-attachments-content-block-1 (Date of use: 27 May 2023).

Bandewar et al "The origins of a research community in the Majengo observational cohort study" https://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-10-630 (Date of use: 27 May 2023).

BDO Wealth advisers "Trusts: All you need to know" https://www.bdo.co.za/getmedia/1ed18ab6-f01e-4a62-83f2-ce1d315d8898/bdo-trust#:~:text=The%20Trust%20Property%20Control%20Act,of%20certain%20persons%20or%20entities. (Date of use: 10 October 2023).

Chennells R and Schroeder D "The San Code of Research Ethics: Its Origins and History" available at: http://www.globalcodeofconduct.org/wp-content/uploads/2019/02/SanCodeHistory.pdf (Date of use: 27 May 2023).

Commercial Use of Human Biological Materials, American Medical Association, Code of Medical Ethics https://www.ama-assn.org/delivering-care/ethics/commercial-use-human-biological-materials (Date of use: 27 May 2023).

Companies and Intellectual Property Commission http://www.cipc.co.za (Date of use: 27 May 2023).

Company profile of Ardais Corp as listed by Bloomberg https://www.bloomberg.com/profile/company/551614Z:US (Date of use: 27 May 2023).

Constitution of Uganda https://www.ilo.org/dyn/natlex/docs/ELECTRONIC/44038/90491/F206329993/UGA4 4038.pdf (Date of use: 27 May 2023).

Global Forum for Health Research The 10/90 report on health research 2000.Geneva: Switzerland

https://apps.who.int/iris/bitstream/handle/10665/66474/Global_Forum_for_Health_Research_eng.pdf?sequence=1&isAllowed=y (Date of use: 10 October 2023).

Health Professional Council of South Africa: Professional Practise Ethics Booklet 13. General Ethical Guidelines for Health Researchers https://www.hpcsa.co.za/Uploads/professional_practice/ethics/Booklet_13_Gen_Ethical_Guidelines_for_Health_Researchers.pdf (Date of use: 27 May 2023).

Hexa Research. Tissue Engineering Market Analysis, Market Size, Application analysis, regional outlook, Competitive strategies and Forecasts, 2016—2024. Felton, CA: Hexa Research, 2015 http://www.hexaresearch.com/research-report/tissue-engineering-market (Date of use: 27 May 2023).

HUGO Ethics Committee Statement on Benefit Sharing https://www.eubios.info/BENSHARE.htm#:~:text=The%20HUGO%20Ethics%20Committ ee%20recommends,the%20issue%20of%20benefit%2Dsharing (Date of use: 27 May 2023).

Human Genetics and Genomics in South Africa: Ethical Legal and Social Implications published by Academy of Science of South Africa http://bit.ly/3sH15hH (Date of use: 27 May 2023).

Human Genome Organisation (HUGO) http://www.hugo-international.org/history (Date of use 27 May 2023).

Human Heredity and Health in Africa3 https://h3africa.org/ (Date of use: 27 May 2023).

Human Tissue Authority: Code of Practice A Guiding Principles and the Fundamental Principle of Consent https://content.hta.gov.uk/sites/default/files/2020-11/Code%20A.pdf (Date of use: 27 May 2023).

International Science Council: membership of Uganda https://council.science/member/uganda-uganda-national-council-for-science-and-technology-uncst/ (Date of use: 27 May 2023).

Laube and Moss "Pseudoxanthoma elasticum" 2005 Arch Dis Child 90:754-756 available at: https://adc.bmj.com/content/archdischild/90/7/754.full.pdf (Date of use: 27 May 2023).

Allied Marker Research "Molecular Diagnostic Market Research" available at https://www.alliedmarketresearch.com/molecular-diagnostics-market (Date of use: 20 September 2023).

Mckenna M "Colonialists are coming for blood-Literally" 2019 *Wired* https://www.wired.com/story/ebola-epidemic-blood-samples/ (Date of use: 27 May 2023).

Njilo N "Stellenbosch University demands return of DNA samples-but UK Lab hits back" 2019 *Timeslive* available at: https://www.timeslive.co.za/news/south-africa/2019-10-16-stellenbosch-university-demands-return-of-dna-samples-but-uk-lab-hits-back/ (Date of use: 27 May 2023).

OECD Frascati Manual available at: https://www.oecd-ilibrary.org/docserver/9789264199040-en.pdf (Date of use: 27 May 2023).

Resnik D "The human genome: common resource but not common heritage" available at: https://www.researchgate.net/publication/255581513_The_human_genome_common_re source_but_not_common_heritage (Date of use: 27 May 2023).

Simm K "Benefit sharing frameworks—justifications for and against benefit sharing in human genetic research" 2007 www.uclan.ac.uk/genbenefit (Date of use: 27 May 2023).

South African Constitution https://www.justice.gov.za/legislation/constitution/pdf.html (Date of use: 27 May 2023).

South Africa. Department of Health Ethics in Health Research Guidelines https://www.health.gov.za/wp-content/uploads/2022/05/NHREC-DoH-2015-Ethics-in-Health-Research-Guidelines-1.pdf (Date of use: 27 May 2023).

South African General Laws (Anti-Money Laundering and Combating TerrorismFinancing) Ammendment Act 22 of 2022 https://www.gov.za/documents/general-laws-anti-money-laundering-and-combating-terrorism-financing-amendment-act-22-2022 (Date of use:10 October 2023).

South Africa Government Protection of Personal Informant Act 2013 https://www.gov.za/sites/default/files/gcis_document/201409/3706726-11act4of2013protectionofpersonalinforcorrect.pdf (Date of use: 27 May 2023).

South Africa Patents Amendment Act 2005 https://wipolex.wipo.int/en/text/179614 (Date of use: 27 May 2023).

South African Trust Property Control Act 57 of 1998 https://www.gov.za/sites/default/files/gcis_document/201505/act-57-1988_0.pdf (Date of use:10 October 2023).

South African Revenue Service https://www.sars.gov.za/businesses-and-employers/trusts/types-of-trust/ (Date of access: 9 October 2023)

Stokstad E "Genetic lab accused of misusing African DNA" 2019 Science 366(6465):555-556 available at: https://www.science.org/doi/full/10.1126/science.366.6465.555 (Date of use: 27 May 2023).

The Convention on Biological Diversity 2002: Bonn Guidelines on Access to Genetic Resources and Fair and Equitable Sharing of the Benefits Arising out of their utilization, https://www.cbd.int/doc/publications/cbd-bonn-gdls-en.pdf (Date of use: 27 May 2023).

The Public Health Act of Uganda http://library.health.go.ug/sites/default/files/resources/Public%20Health%20Act.pdf (Date of use: 27 May 2023).

The San people and the Hoodia plant available at: https://gfbr.global/wp-content/uploads/2015/09/Fifth_Casestudy4.pdf (Date of use: 26 May 2023).

Thambisetty S "Human genome patents and developing countries" available at: http://www.iprcommission.org/papers/pdfs/study_papers/10_human_genome_patents.pd f (Date of use: 27 May 2023).

Uganda National Council for Science and Technology https://www.uncst.go.ug (Date of use: 27 May 2023).

Uganda National Council for Science and Technology: National Research Biobanking Guidelines https://uncst.go.ug/files/National_Biobanking_Gudelines.pdf (Date of use: 27 May 2023).

Uganda National Guidelines for Research involving Humans as Research Participants https://uncst.go.ug/main/wp-content/uploads/download-manager-files/Human%20Subjects%20Protection%20Guidelines%20July%202014.pdf (Date of use: 27 May 2023).

UK Constitution http://www.ucl.ac.uk/constitution-unit/what-uk-constitution/what-uk-constitution (Date of use: 27 May 2023).

United Kingdom Human Tissue Act 2004 Explanatory Notes https://www.eui.eu/Projects/InternationalArtHeritageLaw/Documents/NationalLegislation/UnitedKingdom/humantissueact2004explanatorynotes.pdf (Date of use: 27 May 2023).

United Kingdom Human Tissue Act 2004 https://www.hta.gov.uk/policies/human-tissue-act-2004 (Date of use: 27 May 2023).

United Kingdom NHS Health Research Authority Research Ethics Committee: Standard Operating Procedures https://www.hra.nhs.uk/about-us/committees-and-services/res-and-recs/research-ethics-committee-standard-operating-procedures/ (Date of use: 27 May 2023).

UK Health Department Research Ethics Services: Standard Operating Procedures for Research Ethics Committees https://s3.eu-west-2.amazonaws.com/www.hra.nhs.uk/media/documents/RES_Standard_Operating_Proce dures_Version_7.5.1_August_2021_Final_Accessible_07lVkXt.pdf (Date of use: 27 May 2023).

UK Medical Research Council (MRC) https://www.ukri.org/councils/mrc/Date of use: 27 May 2023).

UK MRC Principles and guidelines for good research practice https://www.ukri.org/publications/principles-and-guidelines-for-good-research-practice/ (Date of use: 27 May 2023).

UK MRC Human tissue and biological samples for use in researchhttps://www.ukri.org/publications/human-tissue-and-biological-samples-for-use-in-research/ (Date of use: 27 May 2023).

United Nations Convention on the Law of the Sea https://www.imo.org/en/ourwork/legal/pages/unitednationsconventiononthelawofthesea.a spx#:~:text=The%20United%20Nations%20Convention%20on,the%20oceans%20and% 20their%20resources. (Date of use: 27 May 2023).

UNESCO Universal Declaration on Bioethics and Human Rights https://unesdoc.unesco.org/ark:/48223/pf0000146180 (Date of use: 27 May 2023).

United Nations Universal Declaration of Human Rights https://www.un.org/en/about-us/universal-declaration-of-human-rights (Date of use: 23 May 2023).

Visiongain 2009 In vitro diagnostics: Market analysis 2009–2024 https://www.visiongain.com (Date of use: 27 May 2023).

Worldbank reports on South Africa

https://thedocs.worldbank.org/en/doc/bae48ff2fefc5a869546775b3f010735-0500062021/related/mpo-zaf.pdf (Date of use: 10 October 2023).

https://databankfiles.worldbank.org/public/ddpext_download/poverty/987B9C90-CB9F-4D93-AE8C-750588BF00QA/current/Global_POVEQ_ZAF.pdf (Date of use :10 October 2023).

World Medical Association of Helsinki. Ethical Principles for Medical Research Involving Human Subjects https://www.wma.net/policies-post/wma-declaration-of-helsinki-ethical-principles-for-medical-research-involving-human-subjects/ (Date of use: 27 May 2023). Yu Ni, Chen F, Ota S et al "Larger genetic differences within africans than between Africans and Eurasians" 2002 Genetics available at https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1462113/ (Date of use: 27 May 2023).

7.11 Newspaper articles

Pomfret and Nelson 2018-12-20

Pomfret J and Nelson D "An isolated region's genetic mother lode" 2000-12-20 https://www.washingtonpost.com/archive/politics/2000/12/20/an-isolated-regions-genetic-mother-lode/4280cf1f-ae9c-42f7-b132-9ddbe26e502f/ (Date of use: 27 May 2023).

Xiong and Wang 2002 Outlook weekly

Xiong L and Wang Y "Harvard University's genetic research in China is illegal"2002 Outlook Weekly 15:48-50

7.12 Miscellaneous sources

A webinar titled "Ethically managing data transfers for research in SA" organised by the SA Medical Research Council, in collaboration with the Department of Science & Innovation of South Africa was held Jun 23, 2022 to deliberate the way forward for a National DTA. https://www.assaf.org.za/wp-content/uploads/2022/06/22412-Data-Transfer-Webinar-Programme-final.pdf (Date of use: 27 may 2023).

H3Africa Working Group. "Harnessing genomic technologies toward improving health in Africa: Opportunities and challenges" 2011 *H3Africa White Paper. Washington, DC:* National Institutes of Health.

7.13 Ethical clearance certificate



UNISA 2020 ETHICS REVIEW COMMITTEE

Date: 2020:08:12

Dear Ms P Chunda-Sinombe ERC Reference No: ST87 Name: P Chunda-Sinombe

Decision: Ethics Approval from 2020:08:12 to 2023:08:12

Researcher: Ms P Chunda-Sinombe

Supervisor(s): Prof M Labuschaigne & Prof S Mahomed

An ethico-legal analysis of benefit-sharing for health research in South Africa

Qualification: PhD in Law

Thank you for the application for research ethics clearance by the Unisa 2020 Ethics Review Committee for the above mentioned research. Ethics approval is granted for 3 years.

The **Negligible risk application** was **reviewed** by the CLAW Ethics Review Committee on 12 August 2020 in compliance with the Unisa Policy on Research Ethics and the Standard Operating Procedure on Research Ethics Risk Assessment.

The proposed research may now commence with the provisions that:

 The researcher will ensure that the research project adheres to the relevant guidelines set out in the Unisa Covid-19 position statement on research ethics attached. Provisional authorisation is granted.



University of South Africa Preller Street, Muckleneuk Ridge, City of Tshwane PO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 3111 Facsimile: +27 12 429 4150 www.unisa.ac.za

- 2. The researcher(s) will ensure that the research project adheres to the values and principles expressed in the UNISA Policy on Research Ethics.
- Any adverse circumstance arising in the undertaking of the research project that is relevant to the ethicality of the study should be communicated in writing to the CLAW Committee.
- 4. The researcher(s) will conduct the study according to the methods and procedures set out in the approved application.
- 5. Any changes that can affect the study-related risks for the research participants, particularly in terms of assurances made with regards to the protection of participants' privacy and the confidentiality of the data, should be reported to the Committee in writing, accompanied by a progress report.
- 6. The researcher will ensure that the research project adheres to any applicable national legislation, professional codes of conduct, institutional guidelines and scientific standards relevant to the specific field of study. Adherence to the following South African legislation is important, if applicable: Protection of Personal Information Act, no 4 of 2013; Children's act no 38 of 2005 and the National Health Act, no 61 of 2003.
- 7. Only de-identified research data may be used for secondary research purposes in future on condition that the research objectives are similar to those of the original research. Secondary use of identifiable human research data require additional ethics clearance.
- No field work activities may continue after the expiry date 2023:08:12. Submission
 of a completed research ethics progress report will constitute an application for
 renewal of Ethics Research Committee approval.

Note:

The reference number ST 87-2020 should be clearly indicated on all forms of communication with the intended research participants, as well as with the Committee.

Yours sincerely,

Prof T Budhram

Chair of CLAW ERC

E-mail: budhrt@unisa.ac.za

Tel: (012) 433-9462

Prof M Basdeo

Executive Dean : CLAW

E-mail: MBasdeo@unisa.ac.za

Tel: (012) 429-8603

orove

URERC 16.04.29 - Decision template (V2) - Approve

University of South Africa Preller Street, Muckleneuk Ridge, City of Tshwane PO Box 392 UNISA 0003 South Africa Telephone: +27 12 429 3111 Facsimile: +27 12 429 4150 www.unisa.ac.za

asdeo

7.14 Digital receipt (Turnitin)



Digital Receipt

This receipt acknowledges that Turnitin received your paper. Below you will find the receipt information regarding your submission.

The first page of your submissions is displayed below.

Submission author: P M CHUNDA-SINOMBE

Assignment title: Complete dissertation/thesis DRAFT

Submission title: Priscilla Chunda-Sinombe complete draft PhD thesis

File name: Priscilla_Chunda-Sinombe_draft_Phd.docx

File size: 395.1K
Page count: 217
Word count: 55,226
Character count: 312,610

Submission date: 14-Mar-2023 10:48AM (UTC+0200)

Submission ID: 2036895683



Copyright 2023 Turnitin. All rights reserved.